# **ENABLING PERSONAL PREFERENCE:**

The Implementation of the Cash and Counseling Demonstration in New Jersey

# Office of the Assistant Secretary for Planning and Evaluation

The Office of the Assistant Secretary for Planning and Evaluation (ASPE) is the principal advisor to the Secretary of the Department of Health and Human Services (HHS) on policy development issues, and is responsible for major activities in the areas of legislative and budget development, strategic planning, policy research and evaluation, and economic analysis.

The office develops or reviews issues from the viewpoint of the Secretary, providing a perspective that is broader in scope than the specific focus of the various operating agencies. ASPE also works closely with the HHS operating divisions. It assists these agencies in developing policies, and planning policy research, evaluation and data collection within broad HHS and administration initiatives. ASPE often serves a coordinating role for crosscutting policy and administrative activities.

ASPE plans and conducts evaluations and research—both in-house and through support of projects by external researchers—of current and proposed programs and topics of particular interest to the Secretary, the Administration and the Congress.

# Office of Disability, Aging and Long-Term Care Policy

The Office of Disability, Aging and Long-Term Care Policy (DALTCP) is responsible for the development, coordination, analysis, research and evaluation of HHS policies and programs which support the independence, health and long-term care of persons with disabilities—children, working age adults, and older persons. The office is also responsible for policy coordination and research to promote the economic and social well-being of the elderly.

In particular, the office addresses policies concerning: nursing home and community-based services, informal caregiving, the integration of acute and long-term care, Medicare post-acute services and home care, managed care for people with disabilities, long-term rehabilitation services, children's disability, and linkages between employment and health policies. These activities are carried out through policy planning, policy and program analysis, regulatory reviews, formulation of legislative proposals, policy research, evaluation and data planning.

This report was prepared under contract #HHS-100-95-0046 between the U.S. Department of Health and Human Services, Office of the Assistant Secretary for Planning and Evaluation, Office of Disability, Aging and Long-Term Care Policy and the University of Maryland. For additional information about this subject, you can visit the DALTCP home page at http://aspe.hhs.gov/daltcp/home.shtml or contact the ASPE Project Officer, Pamela Doty, at HHS/ASPE/DALTCP, Room 424E, H.H. Humphrey Building, 200 Independence Avenue, S.W., Washington, D.C. 20201. Her e-mail address is: Pamela.Doty@hhs.gov.

# **ENABLING PERSONAL PREFERENCE:**The Implementation of the Cash and Counseling Demonstration in New Jersey

Barbara Phillips Barbara Schneider

Mathematica Policy Research, Inc.

March 2003

Prepared for
Office of Disability, Aging, and Long-Term Care Policy
Office of the Assistant Secretary for Planning and Evaluation
U.S. Department of Health and Human Services
Contract #HHS-100-95-0046

# **TABLE OF CONTENTS**

ACK	Page ACKNOWLEDGMENTS			
EXE	CUT	IVE SUMMARY	vi	
l.	INTI A.	RODUCTION  Background	. 1 . 1	
	В.	Solicitation for the Cash and Counseling Demonstration	3	
	C.	Design of the Evaluation  1. Evaluation of Impacts  2. Process Analysis	. 5	
	D.	Guide to this Report	8	
II.	PERA.	RSONAL PREFERENCE: ISSUES AND DECISIONS  New Jersey's Personal Assistance Programs and Goals for the Demonstration  New Jersey's Medicaid Personal Assistance Programs  Goals for the Cash and Counseling Demonstration  Reaction of Stakeholders	<b>11</b> 11 12	
	D.	1. Governmental Stakeholders	13	
	C.	Design Issues and Decisions  1. Eligibility and Appropriateness  2. Outreach and Enrollment  3. Permissible Uses of the Cash Allowance  4. Determining the Level of the Cash Allowance  5. Consulting and Fiscal Services	18 21 23 24	
	D.	Poised for Implementation	31	
III.		GANIZATIONS PROVIDING OUTREACH AND ENROLLMENT, NSULTING, AND FISCAL SERVICES  Outreach and Enrollment Contractor  Criteria  Outreach and Enrollment Contractor  Assuring the Quality of Outreach and Enrollment Services	33 33	

	В.	Consulting Agencies	35
		1. Recruiting Consulting Agencies	35
		2. Recruiting Consultants	
		3. Training Consultants	
		4. Supervision of Consultants	
		5. Assuring the Quality of Consulting	41
	C.	Fiscal Agent	42
		1. Procurement Process	
		2. Host Organization for Fiscal Services	
		3. Staffing for Fiscal Services	
		4. Assuring the Quality of Fiscal Services	46
	D.	Conclusion	46
IV.	OU.	TREACH AND ENROLLMENT	47
	A.	Community Information	47
	В.	Outreach and Enrollment	49
		1. Determining Eligibility and Initial Flyer	49
		2. Databases	
		3. Procedures of the Enrollment Contractor	
		4. Changes in Enrollment Contractor Procedures over Time	
		<ul><li>5. Hiring and Training State Employees as Specialists</li><li>6. Enrollment Procedures Under State Specialists</li></ul>	
		6. Enrollment Procedures Under State Specialists	50
	C.	Attractive and Unattractive Features of Personal Preference	
		1. Attractive Features	
		2. Unattractive Features	59
	D.	Lessons About Outreach and Enrollment	
		1. Should Personal Preference Adopt a Formal Screening Process?	
		2. Providing for Language Diversity	
		3. Importance of Family Participation	
		4. Difficulty in Meeting Enrollment Goals	62
٧.	CAS	SH PLANNING AND MANAGEMENT	
	Α.	Referral to Consulting Agency	
		1. Choice of Consulting Agency	
		2. Selecting a Consultant	67
	B.	Developing the Cash Plan	
		1. Arrangements for the Initial Home Consulting Visit	
		Initial Consulting Home Visit	69 71
		5. Program Review of the Cash Plan	71

	C.	Uses of Cash
	D.	Time to Receipt of Cash 73
	E.	Changes to the Cash Plan751. When Was a Cash Plan Revision Necessary?72. Writing Flexible Cash Plans73. More Efficient Revision of the Cash Plan7
	F.	Keeping Consultants Informed 7
	G.	Major Lessons71. Choice of Consulting Agencies72. Advance Planning for the Home Visit Is Essential73. Techniques to Minimize and Streamline Cash Plan Revision74. Time to Receipt of Cash Allowance75. Resolving Communication Problems7
VI.	REF A.	PRESENTATIVES
	B.	Major Lessons821. Selection of Representative Is Extension of Natural Process82. Need for Representative Depends on Several Factors83. Role of the Representative Varies84. Representatives Are Functioning Well8
VII.	CON A.	Assistance with Employer Responsibilities         86           1. Initial Assistance         8           2. Recruiting Workers         8           3. Hiring Workers         8           4. Time Sheets         8           5. Firing Workers         8           8. Firing Workers         8           8. Firing Workers         8
	B.	Consumer Management of Payroll Functions 89
	C.	Major Lessons961. Consumers Fulfilled Nonfiscal Responsibilities of an Employer92. Recruiting Is a Critical Issue93. Consumers Prefer to Share Fiscal Responsibilities of an Employer9
VIII.	MOI A.	Preventing Abuse of the Cash Allowance

	B.	Monitoring to Prevent Exploitation
		2. Reassessment Visits
		3. Exploitation of Workers?
	C.	Little or No Exploitation or Abuse
		1. Very Little Exploitation of Consumers and Workers
		2. No Abuse of the Cash Allowance
	D.	Major Lessons 98
		Cases of Possible Exploitation of Consumers Can Be Eliminated Before     Cash Payment
		More Attention to Preventing Exploitation of Workers May Be Warranted
		3. Ensuring That Expenditures Are Authorized by a Cash Plan Prevents Abuse
IX.	LES	SSONS FROM NEW JERSEY 100
	Α.	Lessons About Feasibility
		1. Nurse Practice Act Was Not a Barrier
		2. Overcoming Opposition of Traditional Agencies
		<ul> <li>3. Implemented as Planned and Learning from Experience</li></ul>
		5. Achieving Budget Neutrality
	В.	Lessons About Procedures and Structure
		1. Component-Specific Lessons
		2. How Should a Cash Program Be Structured?
	C.	Value of Counseling and Fiscal Services
Х.	TOV	WARD AN ONGOING CASH PROGRAM IN NEW JERSEY 114
	A.	Broader Personal Care Allowance 114
	В.	Continuation of Personal Preference
	Ь.	Continuation of Personal Preference
	C.	Extending Consumer Direction to Other Populations
REF	FERE	ENCES 117
ΔΡΙ	PENI	DIX A. PERSONAL PREFERENCE MATERIALS
$\sim$ i I	F14F	AN ALL ENOUGAET MELENEGOE MALENIALU

# **ACKNOWLEDGMENTS**

Many people have contributed to this report, and several deserve special mention. Dr. Kevin Mahoney, National Project Director, Cash and Counseling, now at Boston College, has guided the demonstration and evaluation from its inception and has been unfailingly helpful. In addition, Dr. Mahoney's comments on an earlier report in this series helped us sharpen the lessons presented here. The thoughtful, thorough comments of Dr. Lori Simon-Rusinowitz, Deputy National Project Director for Cash and Counseling, Center on Aging, University of Maryland, on an earlier report also helped us improve this report. Dr. Pamela Doty, Senior Policy Analyst, Office of the Assistant Secretary for Planning and Evaluation, U.S. Department of Health and Human Services, has been a devoted supporter of the demonstration and evaluation. Her comments on an earlier report challenged us to think more broadly about the lessons we can draw from New Jersey's experience with the Cash and Counseling model.

Dr. Randall Brown, Project Director, Evaluation of the Cash and Counseling Demonstration, Mathematica Policy Research, Inc. (MPR) and Ms. Jennifer Schore, Deputy Project Director for the evaluation, provided careful, thorough comments that helped us improve this report.

Special thanks also are due to members of the Personal Preference staff. They include William Ditto, Project Director, Carolyn Selick, Project Manager, and Renee Davidson, Coordinator of Consultants, who cheerfully and insightfully answered our seemingly endless questions. We also thank the state enrollment specialist whom we interviewed.

We are also indebted to the others we interviewed during our visit to New Jersey, all of whom gave generously of their time. They include senior officials of the New Jersey government, members of the staff of the enrollment contractor, staff of several consulting agencies, staff of the fiscal agent, and representatives of the home care industry in New Jersey.

At MPR, we would like to thank Patricia Ciaccio, who carefully edited the report, and Marjorie Mitchell, who skillfully produced it.

Finally, we are deeply grateful to the people of New Jersey who participated in the demonstration, in either the treatment or the control group. Without their willingness to join in this endeavor, none of this would have been possible. We have learned much from them.

# **EXECUTIVE SUMMARY**

About 1.2 million people receive disability-related supportive services in their homes through state Medicaid plans or home- and community-based waiver services programs. Under state plans, services are restricted largely to human assistance with personal care and homemaking and must be provided by licensed agencies. Additional services (such as adult day care and home modifications) and goods (such as assistive devices) may be offered under waiver programs. However, coverage of these additional services is often limited, and a case manager decides whether they are needed.

In contrast to these traditional service models, states are increasingly offering Medicaid beneficiaries and their families the opportunity to obtain personal care from individual providers--an alternative known as "consumer-directed" care.

Cash and Counseling is an expanded model of consumer-directed supportive services that provides a flexible monthly allowance for consumers to use to hire their choice of workers, including family members, and to purchase other services and goods. Cash and Counseling requires consumers to develop plans that show how they would use the allowance to meet their personal care needs and provides counseling and fiscal assistance to help them plan and manage their responsibilities. Consumers who are unable or unwilling to manage their care themselves may designate a representative, such as a family member, to help them or do it for them. These features make Cash and Counseling adaptable to consumers of all ages and with all types of impairments.

The Robert Wood Johnson Foundation (RWJF) and the Office of the Assistant Secretary for Planning and Evaluation (ASPE) of the U.S. Department of Health and Human Services are sponsoring a demonstration and evaluation of Cash and Counseling. A National Program Office operates from the University of Maryland, Center on Aging. The Centers for Medicare & Medicaid Services (CMS) is helping with the demonstration, primarily with technical assistance and obtaining waivers of federal Medicaid regulations. Mathematica Policy Research, Inc. (MPR) is evaluating the demonstration, which has been implemented in three states: Arkansas, Florida, and New Jersey. The evaluation uses a randomized design--half the enrollees are assigned to a treatment group to receive the cash allowance, half to a control group to continue in traditional supportive services.

This report describes the design and implementation of Personal Preference, New Jersey's model of Cash and Counseling. It also draws lessons from the state's experience. The report is based primarily on in-person interviews conducted in April 2001, about 18 months after the program began enrolling beneficiaries (November 1999). Interviews were conducted with New Jersey state officials, Personal Preference program staff members, officials of organizations representing the personal care industry in New Jersey, and staff members of organizations providing outreach, enrollment, consulting, and

fiscal services under Personal Preference. (New Jersey used the term "consulting" rather than "counseling" in its demonstration.)

# **The Personal Preference Program**

Program "Cashed Out" and Eligibility. New Jersey chose to "cash out" Personal Care Assistance (PCA) under its state Medicaid plan. It did *not* cash out any of its seven Medicaid home- and community-based waiver programs, which provide a variety of services to special populations, such as medically fragile children and frail elderly people with incomes above the usual Medicaid threshold. Only PCA clients were to be eligible for Personal Preference (including current recipients of PCA and those accessed by traditional personal care agencies and found eligible for PCA).

PCA clients who wished to participate in Personal Preference but were unable to manage their own services were eligible to enroll. They could name a family member or friend as a representative to act on their behalf if they were selected for the cash allowance.

**Outreach and Enrollment.** To identify eligible beneficiaries, New Jersey required that a personal care agency send the Personal Preference program a form with contact information for each beneficiary assessed or reassessed for PCA. The form also included information on care plan hours, which was used in determining the amount of the cash allowance for demonstration participants assigned to the treatment group.

The state did not work through personal care agencies to conduct outreach for Personal Preference; instead, it contracted with other organizations and hired state employees. Some agency staff members were concerned that the quality of care would suffer under Personal Preference, others that it would affect agency revenue and their employment. Therefore, New Jersey avoided working through agencies, out of concern that some agency staff might discourage PCA clients from participating in Personal Preference.

Although some agency staff members reportedly did discourage participation in the demonstration, personal care agencies were cooperative overall. The industry believed that the program was appropriate for some people with disabilities, and it had a long history with and professional respect for the director of Personal Preference. Moreover, Personal Preference responded to an important industry concern by discouraging consumers from hiring their agency aides.

During the demonstration, New Jersey both contracted with other organizations and hired state employees to conduct outreach and enrollment for Personal Preference. Concerned about possible delay in hiring state employees, New Jersey initially amended

the contract of an organization that was helping the state market Medicaid managed care, adding responsibility for outreach and enrollment under Personal Preference. State Personal Preference staff mailed materials on the program to PCA recipients. Staff of the enrollment contractor telephoned and visited those who received materials to explain the program and enroll them if they wished to participate. Despite revisions to outreach and enrollment procedures, however, the pace of enrollment consistently fell below the contract target, and costs were higher than anticipated. Therefore, Personal Preference decided to hire state employees to conduct outreach and enrollment and was able to do so after substantial delay. (Because of attrition of state-employed enrollment staff later--after spring 2001--New Jersey again obtained a contractor for outreach and enrollment. It did so this time by amending the contract of the organization providing fiscal services under Personal Preference.)

The day-to-day outreach and enrollment activities of the enrollment contractor and the Personal Preference program were coordinated using a common database. This database was initiated with the contact and care plan information that personal care agencies provided and combined with information from New Jersey's Medicaid program. It was updated weekly as outreach and enrollment continued.

Ultimately, about 1,750 people participated in the Personal Preference Demonstration in New Jersey (in the treatment and control groups combined). They enrolled between December 1999 and June 2002.

**Determining the Amount of Allowance.** New Jersey based the amount of the cash allowance on the current PCA care plan. Hours planned were cashed out at the hourly rates the state paid for weekday and weekend PCA services (in New Jersey, care plans differentiate between weekday and weekend hours). At about the time of our visit, the amount of the allowance ranged from about \$300 to about \$2,800 a month. The average was about \$1,300.

No discounting was applied in determining the amount of the allowance under Personal Preference. (Discounting is intended to ensure the budget neutrality of the cash program when the cost of services received is less, on average, than the cost of services planned--a result, for example, of hospital admission of consumers or an insufficient supply of aides). While planning the demonstration, New Jersey had determined that the historical cost of PCA services received was approximately equal to the cost of services planned--an indication that discounting was unnecessary.

**Reassessment.** PCA recipients are reassessed in New Jersey at six-month intervals. Nurses employed by personal care agencies do the reassessments, using a structured form that the state developed. To mimic this process, Personal Preference arranged for nurses employed by the Medicaid program to conduct reassessments of cash recipients, using the same form as that used for agency service recipients. This

procedure eliminated the possibility that consultants--acting as advocates for consumers--might increase care plan hours (and, thus, the amount of the allowance).

**Cash Planning and Uses of Cash Allowance.** Before receiving a cash allowance, Personal Preference required that consumers develop a plan for its use. A member of the state Personal Preference program staff had to approve all cash management plans and all revisions to those plans.

Nearly all consumers who received a cash allowance hired a worker--usually a family member or friend--with the funds. New Jersey exercised its option under demonstration waivers to allow consumers to hire legally liable relatives. Thus, consumers in Personal Preference could hire their spouses. (Under federal regulations, spouses are the only legally liable relatives of adults.) Some consumers also purchased assistive equipment and personal care supplies (not covered by Medicaid) with the allowance, and a few modified their homes (for example, by adding a ramp).

Consulting Agencies. Initially, New Jersey recruited a large number of human service agencies across the state to provide consulting under Personal Preference. It did this to best serve New Jersey's culturally diverse population and to provide consumers with a choice of consulting agencies. To help in recruiting these agencies, New Jersey made agency participation administratively straightforward by signing memoranda of agreement (rather than contracts) with agencies providing consulting. Ultimately, 34 agencies signed such agreements. Some were public (for example, county departments of social services); some were private, nonprofit (for example, an Independent Living Center); and some were private, for-profit (for example, an agency that provided case management services).

Over time, the number of consulting agencies dropped dramatically (there were 12 in spring 2001). Some agencies withdrew from the cash program (partly because of competing demands on staff time), and Personal Preference dropped agencies not performing satisfactorily. It was possible to have fewer consulting agencies, partly because consumers did not have experience or information to help them choose an agency and did not particularly value having a choice. Therefore, the program began assigning consumers to agencies.

Personal Preference paid the consulting agency a lump sum to complete the cash management plan (initially \$53, later increased to \$75) and thereafter an hourly fee for consulting (initially \$18, increased to \$26). Rates were increased in part because agencies reported that the initial rates covered the cost of salaries, but not the cost of fringe benefits and other overhead.

The approach that Personal Preference adopted for payment for consulting services limited the state's costs for consulting. In addition to the single, lump-sum payment for development of the initial cash management plan (no matter how much consultant time was

required to complete it), the cost of hourly consulting services was capped by limiting payment to 19 hours (later, 20 hours) annually per consumer.

New Jersey did not require Personal Preference consultants to have professional credentials, but most were human services professionals. Given the number of consulting agencies and the overall Personal Preference caseload, the caseload of most agencies was not large. At the time of our visit, most agencies had small caseloads and only one or two consultants, who also had other agency responsibilities.

Consultants had many responsibilities under Personal Preference. They visited treatment group members to help them prepare their initial cash management plans and revisions to plans. They then submitted the plans to the state Personal Preference office for approval. Consultants also advised consumers about the nonfiscal responsibilities of an employer, including recruiting, hiring, training, supervising, and (if necessary) firing workers. In addition, consultants monitored the condition of the consumer, speaking to consumers by telephone at least monthly (for the first six months after enrollment) and visiting them quarterly. Consultants were not responsible for monitoring the uses of the cash allowance.

**Fiscal Services.** Personal Preference issued a solicitation for a single organization to provide fiscal services for consumers across the state. After a long process, including a legal dispute contesting the organization chosen initially, the state signed a contract with a New Jersey organization that provided a variety of human services. (Later, this organization provided consulting, outreach, and enrollment services in addition to fiscal services.)

The major fiscal services provided to Personal Preference consumers were assistance with preparation and submission of payroll tax returns and other documents (including those pertaining to federal and state payroll taxes and state unemployment insurance) and check-writing services. The fiscal agent implemented strict procedures for comparing timesheets and check requests with the cash management plan before checks were cut.

Personal Preference allowed consumers to receive the allowance in cash and accept full responsibility for its management (including payroll taxes), but they had to pass a skills examination before doing so. At the time of our visit, no consumers were managing the allowance themselves. Rather, the fiscal agent was holding the allowance on behalf of each consumer and disbursing funds in accordance with his or her cash management plan.

The fiscal agent earned fees from consumers for fiscal services that they used, and it earned fees from the state for the other services it provided to consumers. A schedule of charges was developed. Consumers were charged for such tasks as cutting a check (75

cents per check) stopping payment on a check (\$28 per stop-payment order), and arranging for a criminal background check (\$15 to \$60, depending on the extent of the search). The fiscal agent charged the state for other tasks, such as processing W-4 forms and other forms in worker employment packets (\$90 per packet).

New Jersey also provided the fiscal agent with start-up funds. Before there was a substantial caseload of cash recipients, the fiscal agent had a serious cash flow problem and obtained an internal loan from its host organization. New Jersey had intended that the up-front payment of start-up funds would limit cash flow problems; however, the fiscal agent used these funds mainly to purchase office equipment for its Personal Preference operation.

# **Lessons from Personal Preference**

New Jersey's experience provides many lessons about operating a consumerdirected cash allowance program under Medicaid. The state was willing to learn from its experience and to try a different approach if problems arose.

#### **Outreach and Enrollment**

- Self-screening of personal assistance recipients for appropriateness is
  workable. It would have been difficult (if not impossible) to develop a formal process
  to identify in advance consumers who would be unable to manage the cash benefit,
  either by themselves or with the help of representatives. Instead, Personal
  Preference relied on consumers and representatives to screen themselves, and this
  approach worked.
- Providing for language diversity is important and possible in a consumer-directed program. Non-English-speaking consumers may be able to benefit greatly from hiring workers who speak their languages and are familiar with their cultures. Personal Preference hired staff members who spoke the languages most common in New Jersey. It also translated program materials into these languages, prepared a notice in 14 languages asking consumers to have materials translated, asked family members and friends to translate, and used the AT&T language line.
- Information should be available in multiple media to accommodate those with vision and hearing impairments and those who are not competent readers in any language.
- To keep program costs down, efficiency in arranging home visits to describe the program and enroll beneficiaries is important. Travel time can be reduced by scheduling visits on the same day to several beneficiaries in the same area. Repeat

- visits can be avoided by having family members present during the initial enrollment visit, even if that visit has to be scheduled outside of business hours.
- To conduct outreach and enrollment, external contracting is not necessarily better than the hiring of state employees (or vice versa). Neither the Personal Preference enrollment contractor nor state employees were able to sustain an enrollment pace that met the sample-size target for the evaluation.
- While both elderly adults and younger adults with physical disabilities found Personal Preference attractive, younger adults appear to have been more likely to enroll. More than half (53 percent) of those enrolled in the demonstration were elderly beneficiaries. However, younger adult participants in the Personal Preference demonstration represented more than 20 percent of the number of non-elderly adult PCA recipients in New Jersey in the year before the demonstration, while the elderly participants represented just over 10 percent of the number of elderly PCA recipients that year. Thus, the cash program in New Jersey attracted a substantial minority, but not a majority, of adult Medicaid personal care recipients.

## **Consulting Agencies**

- Cash allowance programs may find it difficult to recruit enough agencies to
  provide consumers across the state with a choice of agencies providing
  satisfactory consulting services. New Jersey had difficulty recruiting multiple
  agencies in some parts of the state and dropped some agencies out of
  dissatisfaction with the consulting services provided.
- Consumers do not necessarily value having a choice of consulting agencies.
   Most Personal Preference consumers did not have knowledge about or experience with the agencies offering consulting and did not seem to value having a choice.
- Agencies of a variety of auspices can satisfactorily provide consulting.
   Satisfactory consulting services were provided to Personal Preference consumers by public; private, nonprofit; and for-profit agencies.
- A minimum agency caseload is necessary for efficient provision of consulting.
   This caseload needs to be large enough to support a single consultant, working for an appreciable portion of time on the Cash and Counseling program.
   Consultants with small caseloads have difficulty staying up-to-date on program procedures and policies. Having a single consultant per agency can work, although that consultant needs peer support, supervision, and backup.
- Having 10 to 15 consulting agencies statewide can work. Having a large number of consulting agencies consumes program staff time to recruit (and replace)

agencies and to ensure the quality of consulting services. By spring 2001, 12 agencies (down from 34) were providing Personal Preference consulting, and the state program staff was satisfied with their performance.

# **Cash Planning and Consulting Services**

- Careful planning of the initial consulting home visit reduces the need for multiple visits and thus cuts delays in receipt of the cash allowance and program costs. Before the initial consulting visit, Personal Preference consumers were asked to begin to think about how to spend the allowance, whom to hire, and whether to name a representative. Family members and friends who were potential workers and representatives were asked to be present.
- Revisions to the cash management plan are frequently necessary. Because
  expenditures are checked against the cash management plan, the plan must be
  revised to reflect changes in consumer need, and such changes occur frequently.
- The need for revisions to cash management plans can be reduced by writing
  flexible plans, and the cost of revision can be reduced by automation.

  Personal Preference reduced the number of situations in which a revision was
  required by writing flexible cash plans that did not name workers or vendors. The
  program entered the cash plan into an electronic spreadsheet, thereby speeding
  recalculation of plan totals.
- Consumers need varying amounts of consultant assistance, with some needing a substantial amount. Some Personal Preference consumers needed substantially more consulting assistance than others did, and some needed substantially more than expected. The 19 hours of consulting a year for which Personal Preference would pay (after the cash plan was in place) was sometimes exhausted.

#### Representatives

- Many consumers participating in the Cash and Counseling model will name a
  representative to help them manage the cash allowance. Personal Preference
  consultants reported that up to two-thirds of elderly beneficiaries named
  representatives. Sometimes, the family (rather than the consumer) wanted a
  representative named.
- Representatives are usually family members or friends who are already helping the consumer.

- The need for a representative depends on the availability of training from consultants, familiarity with the cash program, and availability of supportive equipment--as well as consumer disability. As consumers learn about the cash program from consultants, the consumer may take over management of the allowance from a representative. Rarely, supportive equipment (such as a talking computer for a blind consumer) can eliminate the need for a representative.
- The role of the representative can vary considerably. While the representative was the sole decision maker for those Personal Preference consumers who were completely unable to communicate their preferences, both the consumer and representative typically contributed to the management of the cash allowance under Personal Preference. In some cases, representatives solicited consumer preferences; in other cases, consumers and representatives functioned as a team. Sometimes, the consumer was the primary decision maker and the representative functioned as a liaison to the outside world.
- Representatives successfully manage the cash allowance on behalf of consumers. Consultants judged that representatives were obtaining input from the consumers when possible and were faithful to the best interests of the consumers.

#### **Amount of Allowance**

• The appropriate discount rate may change over time. New Jersey personal care industry sources reported a substantial reduction during the demonstration in average hours of personal assistance received, as a result of a state utilization review. Hours received also may have been reduced by changes in the supply of aides as the unemployment rate fell. These changes may have resulted in a change in the ratio of the cost of hours of care planned to hours of care received, and possibly a need to introduce a discount rate for Personal Preference.

#### **Uses of Cash**

- Restrictions on the use of the cash allowance need not trouble consumers.

  Most consumers did not find the Personal Preference program's restrictions on the uses of the cash allowance troublesome. While insisting that purchases be related to the need for personal assistance, the program fostered creative use of the allowance and interpreted it broadly.
- The Cash and Counseling model appears to tap a new source of personal
  assistance workers--family members and friends. Program staff reported that
  most workers hired by consumers were willing to assist a loved one but were not
  interested in agency employment. Family members and friends can help people
  whom agencies were not able to serve fully.

## Time to Receipt of Cash Payment

- Streamlined procedures can reduce time to receipt of the first cash payment.
   Under Personal Preference, the time required for each of the steps leading up to receipt of the cash payment could appreciably delay its receipt. Elimination of consumer choice of consulting agencies reduced delay to receipt of the first cash payment, and Personal Preference decreased notice time to personal care agencies regarding consumers about to receive their first cash payments.
- Communication problems can be serious enough to increase the time to receipt of the first cash payment. With four major actors (consumer/representative, consultant, state program staff, and fiscal staff), there were 12 one-way communication pathways under Personal Preference. From this perspective, it is not surprising that communication was sometimes serious enough to delay receipt of the first cash payment.
- Efficient procedures can reduce communication problems. Personal Preference improved communication by training consumers about whom to call for what issue, adding an explanation of errors in exception reports from the fiscal agent, and using multiple-party telephone calls to reduce the need for several one-on-one calls.

# **Consumers as Employers**

- Almost all consumers (including those with representatives) prefer to have a
  fiscal agent handle payroll functions and disburse funds from the allowance.

  Although Personal Preference consumers paid modest fees for the services of the
  fiscal agent, they overwhelmingly chose to delegate responsibility for payroll functions
  and check writing to the fiscal agent.
- Recruiting is critical for those consumers who do not have family or friends
   available to hire as workers. Personal Preference program staff reported that the
   receipt of the cash allowance was delayed for consumers who had difficulty recruiting
   a worker and that those who could not recruit a worker tended to drop out of the
   program. (Since our visit, New Jersey has moved to develop a worker registry to
   assist consumers in recruiting workers.)
- Almost all consumers (who are able to recruit workers) learn to fulfill the non-fiscal responsibilities of employers. Personal Preference consumers (and representatives) varied with respect to the amount of consultant training they needed on hiring, training, supervising and firing workers. However, even those who needed a substantial amount of consultant assistance, learned in time to fulfill successfully the non-fiscal responsibilities of an employer.

## **Avoiding Abuse and Exploitation**

- Abuse of a cash allowance--even an allowance of a substantial amount--can be prevented by careful development of a management cash plan, followed by strict enforcement of procedures to ensure payment for only those goods and services allowed under the plan. Personal Preference cash allowances averaged about \$1,300 a month at the time of our visit. Every cash management plan (including every revised plan) was reviewed to be sure that only allowable goods and services were included, and the fiscal agent rigorously checked every disbursement against the cash plan. Review of receipts was not required in Personal Preference and thus was unimportant in preventing abuse of the cash allowance in New Jersey.
- Neglect and exploitation of consumers can be eliminated before cash payment
  is made if members of the program staff are mindful of the possibility and
  attentive to cases in which something seems amiss. Under Personal
  Preference, the few cases of potential neglect or exploitation were identified before
  or during the first home visit by consultants. These cases were resolved before the
  consumers received their first cash payment.

## **Value of Counseling and Fiscal Services to Consumers**

The tenet of the Cash and Counseling model that most distinguishes it from other benefit and voucher programs is the provision of counseling and fiscal services to help consumers manage the cash allowance. Clearly, consumers valued having fiscal services available. Although consumers were charged fees for fiscal services, almost all chose to use the fiscal agent.

The value of consulting services is more difficult to assess. Consumers varied greatly in the amount of advice and assistance they needed from consultants. Many needed oral explanations of the program and its procedures, in addition to written materials. Some needed a great deal of assistance as they developed a cash management plan, recruited workers, and fulfilled the other responsibilities of an employer. Overall, consulting seems to have been very valuable--perhaps essential--to the success of the cash program in New Jersey. Future analysis of surveys that asked consumers about their satisfaction with consultants will shed more light on this issue.

# **Future Cash Program?**

New Jersey views a consumer-directed cash program as a valuable part of a package of programs within its Medicaid state plan designed to meet the needs of its citizens. The state is working to offer an ongoing waiver program similar to Personal Preference and is interested in expanding the personal assistance benefit to include

equipment so that it can offer a consumer-directed program similar to Pers Preference without waiver authority.	onal

# I. INTRODUCTION

# A. Background

#### 1. A New Model of Medicaid Personal Assistance

About 1.2 million people receive disability-related supportive services in their homes through state Medicaid plans or home- and community-based waiver services programs (LeBlanc et al. 2001; and Kitchener and Harrington 2001). Under state plans, services must be provided by licensed home care agencies and are largely restricted to human help with personal assistance and homemaking. The agencies recruit, train, schedule, and supervise the aides or attendants who assist beneficiaries. Under waiver programs, adult day care, assistive devices, and home modifications can be offered in addition to in-home aide services. However, coverage of these additional services is often limited, and a case manager, not the beneficiary, decides whether they are needed.

In contrast to these traditional service models, states are increasingly offering Medicaid beneficiaries and their families the opportunity to obtain personal assistance from individual providers (Velgouse and Dize 2000). This alternative is called "consumer-directed" care, as Medicaid beneficiaries who use individual providers assume the employer's role of hiring, managing, and (possibly) terminating their workers (Eustis 2000). An expanded model of consumer direction would allow beneficiaries to manage not only their human assistance but other covered supportive services as well.

Cash and Counseling is an expanded model of consumer-directed supportive services. It provides a flexible monthly allowance that consumers can use to hire their choice of workers, including family members, and to purchase other services and goods (as states permit). Cash and Counseling requires that consumers develop plans showing how they would use the allowance to meet their personal assistance needs and provides counseling and fiscal help to aid them in planning and managing their responsibilities. Consumers who are unable or unwilling to manage their care themselves may designate a representative, such as a family member, to help them or do it for them. These features make Cash and Counseling adaptable to consumers of all ages and with all types of impairments.

With funding from the Robert Wood Johnson Foundation (RWJF) and the Office of the Assistant Secretary for Planning and Evaluation (ASPE) of the U.S. Department of Health and Human Services, the Cash and Counseling Demonstration and Evaluation was

implemented in three states--Arkansas, Florida, and New Jersey.<sup>1</sup> The Centers for Medicare & Medicaid Services (CMS) was responsible for the waivers of federal Medicaid regulations required for the demonstration.<sup>2</sup>

Because their Medicaid programs and political environments differed considerably from each other, the demonstration states were not required to implement a standardized intervention, but they did have to adhere to basic Cash and Counseling tenets, as summarized above. The states' resulting demonstration programs differed in their particulars, so each is being evaluated separately, by Mathematica Policy Research, Inc. (MPR).

# 2. Possible Advantages and Disadvantages of Cash and Counseling

When planning for the Cash and Counseling Demonstration began, its designers saw the payment of a cash allowance as having the potential to give beneficiaries the power to purchase services that best fit their long-term care needs and individual values. Beneficiaries were expected to design individual service packages. As alternatives to disability-related supportive services, they could hire workers privately, make home modifications, purchase equipment and supplies, or purchase commercial services (Cameron 1995; and Kane 1996). They were thus in a better position to arrange for services at times that best met their needs. They could also hire their friends and relatives if they believed that doing so was in their best interest. Depending on the level of the cash allowance, the cost of traditional services, and the cost of alternative services, beneficiaries might also be able to purchase more hours of services than they received from the traditional program. These changes were seen as having the potential to increase autonomy, address unmet needs better, and improve satisfaction--changes that might in turn result in improved functioning (Kane 1996).

In addition, at that time, reductions in public expenditures were viewed as possible because traditional case management services and administrative functions were being eliminated under Cash and Counseling, and beneficiaries took responsibility for managing their own services. If the cost of providing counseling was less than that of these traditional functions, savings would accrue. Savings might also accrue if the cost of paying the cash allowance was less than the cost of processing claims. Moreover, the consumer (not the state) was the employer of record under Cash and Counseling--a difference that could

<sup>&</sup>lt;sup>1</sup> For simplicity, we refer to a single Cash and Counseling *Demonstration*. Because each state was expected to design its own demonstration (within the constraints laid down by the funders and federal regulations, including waiver terms and conditions), the program was originally referred to as the Cash and Counseling *Demonstrations*. However, because a single National Program Office provided oversight and guidance to all the states and a single evaluation contractor was selected, references to a single "demonstration" eventually supplanted references to multiple "demonstrations."

<sup>&</sup>lt;sup>2</sup> At the time, CMS was called the Health Care Financing Administration (HCFA).

potentially reduce state costs (see, for example, Jackson 1994; Cameron 1995; Doty et al.1996; and Flanagan 1994). Because the state was not the employer of record, it might avoid collective bargaining with attendant unions (and the costs of that bargaining). In addition, the likelihood of successful liability actions against the state (and costly settlements) might be reduced because it was not the employer of record.

At the time the Cash and Counseling Demonstration was being designed, states had little experience with cash allowance programs. Therefore, public officials were concerned about possible abuse of the program. They feared that relatives or workers might neglect or exploit beneficiaries or that beneficiaries might not use their cash allowance for the intended purpose. Some public officials were also concerned that a cash allowance might create more demand for services (the so-called "woodwork" effect) and strain available resources.

There was concern that traditional providers of supportive services might object to Cash and Counseling because it might reduce their revenue and place them at a competitive disadvantage relative to privately hired workers. Traditional providers might also object that potential workers would not be adequately trained or supervised. Finally, organized labor might not support Cash and Counseling, because no collective bargaining entities existed to represent privately hired workers (Cameron and Lagoyda 1997).

# **B.** Solicitation for the Cash and Counseling Demonstration

The central question RWJF and ASPE posed for the demonstration was: How does Cash and Counseling compare to traditional case-managed supportive services? States interested in participating in the demonstration were free to propose Medicaid personal assistance services (PAS) funded under the optional state plan benefit, Medicaid programs funded under home- and community-based waivers, or programs funded by state general revenues.<sup>3</sup> These were the demonstration "feeder" programs. Existing consumer-directed programs were excluded from the demonstration.

RWJF and ASPE also stipulated an evaluation that employed a rigorous randomized design. Thus, people participating in the demonstration were to be assigned either to a treatment group (to receive the cash allowance) or to a control group (to continue under traditional PAS, home- and community-based waiver services, or a state-funded program).

Page 3

-

<sup>&</sup>lt;sup>3</sup> The Medicaid PAS benefit covers only personal assistance services; as a state plan service, it must be made available to all eligible Medicaid beneficiaries statewide who apply. In contrast, home- and community-based waivers may cover a variety of goods (for example, personal care supplies) and services (for example, chore services and behavioral therapy). Waiver services can be restricted to particular groups (for example, medically fragile children), and the number of slots in waiver programs can be capped.

The effect of the requirement for a randomized design was to limit the demonstration to states with large PAS, waiver, or state-funded programs (or combinations of programs). Only in such states was it possible to obtain the sample sizes needed for the evaluation.

States were expected to include elderly people with disabilities, as well as younger adults with disabilities, in the Cash and Counseling Demonstration. Younger adults with disabilities have long advocated for consumer-directed care. Adoption of a disability model for personal assistance for elderly people was being debated at the time the solicitation for the demonstration was issued, and there was policy interest in extending such care to elderly people with disabilities (see, for example, Simon-Rusinowitz and Holland 1993; and Doty et al. 1996). The states could include children with disabilities if they wished.

The solicitation anticipated that states would seek waivers of the federal regulations restricting cash payments under the Medicaid program. To grant such waivers (as in demonstration waivers generally), CMS required that the cash program affect the federal budget no more than the PAS or home- and community-based waiver program being "cashed out." That is, the cash program was required to be "budget neutral." CMS's traditional approach to calculating budget neutrality involves comparing the monthly cost per recipient of the demonstration program and the traditional program. Another of the terms and conditions for the demonstration waivers limited the potential impact of the demonstration on public costs by restricting the number of new entrants to the cash program. During the demonstration, the ratio of the number of new entrants to the number of current recipients entering the cash program was not to exceed the historical average for the traditional program.<sup>4</sup>

The demonstration solicitation required that the cash allowance cover a broad range of services (such as equipment and home modifications) in addition to personal assistance workers. Furthermore, the solicitation anticipated that legally liable relatives (that is, spouses and parents of minor children) might be hired as personal assistance workers and that states would have to seek a waiver of the federal restriction on such hiring.

Consistent with the Cash and Counseling model, the demonstration solicitation required that counseling and fiscal services be provided. These services were to help personal assistance recipients by giving them information and advice, teaching them skills, and providing support services, including help with payroll and bookkeeping activities. Demonstration states could decide what specific counseling services to offer.

-

<sup>&</sup>lt;sup>4</sup> Only those who were currently receiving PCA or who already had been assessed and approved for PCA were eligible for Personal Preference. Because the latter group was very small, this term and condition had little effect on the New Jersey program.

Seventeen states submitted bids in response to the solicitation for the Cash and Counseling Demonstration. Four states were selected: Arkansas, Florida, New Jersey, and New York. New York dropped out of the demonstration before beginning operations, because its local social service districts had little interest in participating. This report on New Jersey is one of a series of three. Each report tells the story of the implementation of the Cash and Counseling model in a particular state.

# **C.** Design of the Evaluation

The MPR evaluation of the Cash and Counseling Demonstration has two goals.<sup>5</sup> The first goal is to estimate the impacts of provision of a cash allowance in lieu of disability-related supportive services. The second goal is to document and analyze the implementation of the Cash and Counseling model as it unfolded. These two goals are interrelated, as impacts can be interpreted and generalized only in light of how the Cash and Counseling model was implemented.

# 1. Evaluation of Impacts

The evaluation will estimate impacts on consumers, caregivers, and costs to the public. It will also describe participation in the demonstration.

Cash and Counseling is expected to affect consumers' use of, unmet need for, and satisfaction with supportive services. As a result, it may also affect their health and functioning. Because consumers purchase supportive services on their own, rather than relying solely on agencies, they are likely to have more control over who provides their services and how and when these services are delivered. Consumers may use different amounts or mixes of services than they would have received under traditional Medicaid programs. They may also use their funds to buy equipment that increases their independence. This greater flexibility that the cash allowance provides should reduce consumers' unmet need and improve their satisfaction with supportive services. In addition, if the quality of personal assistance improves, consumers' functional independence and disability-related health may also improve. Although the cash program is expected to improve consumer outcomes, the evaluation will also assess whether any outcomes worsen.

Cash and Counseling could affect caregivers in several ways. Family and friends providing unpaid care to consumers prior to the consumer's enrollment in the demonstration could face fewer demands on their time if consumers hire attendants or use the cash allowance to purchase assistive equipment. If consumers mismanage the

<sup>&</sup>lt;sup>5</sup> In addition, the University of Maryland, Baltimore County, is conducting a qualitative study of triads consisting of consumers, their caregivers, and their consultants.

allowance, however, unpaid caregivers may need to provide more care than they did before. The evaluation will also investigate the experience of caregivers who are hired and paid under the demonstration. The working conditions, job satisfaction, and physical and emotional strain that paid caregivers experience will be measured and compared to that of agency workers providing care to control group members.

The evaluation will estimate Cash and Counseling's effects on Medicaid costs for supportive services alone and for all costs paid by Medicaid and Medicare. Costs for personal assistance may increase or decrease, depending on the amount of the monthly cash allowance (on average) and the number of months in which people receive the cash allowance, relative to traditional supportive services. Costs for other health care may also increase or decrease. If consumers receiving the cash allowance are more likely to receive care when they need it, they may have fewer falls or pressure sores (for example) and thus have lower health care costs. On the other hand, if recipients of the cash allowance hire workers, who are less well trained than agency workers, their health may suffer, resulting in higher health care costs.

The major sources of evaluation data for the impact analyses are (1) telephone surveys with demonstration participants and their caregivers, and (2) Medicare and Medicaid eligibility and claims data. People who agree to participate in the demonstration must complete a baseline telephone interview before they are randomly assigned to the treatment or control group. Six months later, MPR interviews treatment group members about their early experiences with the program. Nine months after enrollment, MPR interviews both treatment and control group members to collect information on their satisfaction, quality of care, quality of life, use of other formal and informal care, and health and functional status. Around the same time, unpaid caregivers identified at baseline are interviewed about the type and amount of care they provide, their relationship with the consumer, and their satisfaction with the paid care the consumer receives. Samples of paid workers identified in the nine-month survey are also interviewed about earnings and benefits, job satisfaction, and problems encountered on the job. Medicaid and Medicare claims and eligibility data will be used to study the cost of supportive services, the use and cost of medical services, and the participation rate in the program being cashed out.

## 2. Process Analysis

A second component of the evaluation examines program structure and implementation. This process analysis, of which this report is a part, has two objectives. First, it documents demonstration operations and the context in which the demonstration operated in each of the three states (Arkansas, Florida, and New Jersey) participating in the Cash and Counseling Demonstration. In doing so, it informs the quantitative analyses of impacts. Second, the process analysis develops lessons about designing and managing a Cash and Counseling program.

Specifically, this report seeks to address three major sets of questions:

- 1. How did New Jersey structure its Cash and Counseling program, and what led it to adopt this structure?
- 2. How did New Jersey implement its program? Did it implement it according to its plans? If not, how and why did it depart from its plans?
- 3. What lessons can we learn from the New Jersey experience about structuring and operating a Cash and Counseling program?

The process analysis is based mainly on three data sources, the primary one being in-person interviews conducted in April 2001 with:

- State officials of the New Jersey Department of Human Services, the department that sponsors the Personal Preference program, including cabinet- and subcabinet-level officers.
- State Personal Preference staff, including the program director, the program manager, the coordinator of consulting services, and an enrollment specialist.
- Staff of the enrollment contractor, including the program manager, the community relations manager, and an enrollment specialist.<sup>6</sup>
- Staff of the fiscal agent, including the executive director of the host organization and the program manager for Personal Preference.<sup>7</sup>
- Executive directors of three organizations representing the personal care industry in New Jersey.
- Staff of six agencies providing consulting at the time of our visit, including the director and one or more consultants at each agency.

With the help of state Personal Preference program staff, we identified six agencies, operating under different auspices (public, private nonprofit, and private for-profit) and located in different regions of the state, whose staff members were knowledgeable about

<sup>&</sup>lt;sup>6</sup> We refer to the initial enrollment contractor. After our visit, New Jersey contracted with the organization providing fiscal services to provide outreach and enrollment as well.

<sup>&</sup>lt;sup>7</sup> At the time of our visit, the fiscal agent had recently begun to provide consulting services. We did not interview a consultant there; nor did we interview staff of the fiscal agent about the provision of consulting services there.

Personal Preference. In addition, we interviewed members of the staff of one agency that had dropped out of Personal Preference after helping only a few consumers.<sup>8</sup>

The second source of information for the process analysis is demonstration documents, such as demonstration state protocols prepared for CMS, state quarterly reports required by the National Program Office, and forms and materials for consumers and consultants.

The third data source is information the authors obtained by participating in project meetings and telephone conference calls, which included project status reports and discussion of issues facing the Cash and Counseling states. The authors attended project meetings, which were held twice a year. One of the authors regularly participated in telephone conference calls that were held with state program staff weekly (later biweekly) throughout the demonstration.

A major strength of this process analysis is that it uses the reports of those who were directly involved in Personal Preference. Many people were interviewed in person. The interviews were extensive and the interviewees extremely knowledgeable--at times, insightful. Their reports are certainly credible. Moreover, state Personal Preference staff reported on the cash program in numerous telephone conference calls and meetings.

However, this strength is also a limitation in that this report relies on the perceptions of those involved in Personal Preference. To minimize the possibility of error based on misconception, we have collected information on key topics from multiple perspectives (when possible).

Another project report will supplement this description of the implementation of Personal Preference. It will combine the perspectives of program staff, presented here, with those of consumers and consultants and with descriptive data (such as fiscal agent data categorizing the actual uses of cash). When the quantitative analyses are complete, we will have better evidence about the validity of many of these perceptions. One of the strengths of this evaluation is having multiple data sources based on different methods.

# D. Guide to this Report

This report has 10 chapters. Chapter II describes the goals of the state and other key stakeholders and the approach that New Jersey took to critical issues in designing Personal Preference. Chapter III explains how New Jersey selected organizations to provide outreach and enrollment, consulting, and fiscal services, as well as the

<sup>&</sup>lt;sup>8</sup> Staff members at this agency were interviewed by telephone.

organizations that the state chose. Chapter IV describes outreach to develop community interest in Personal Preference and the enrollment process for it. Chapter V discusses the development and approval of cash management plans and the uses of cash. Chapter VI details the selection and functioning of representatives named to manage the cash allowance on behalf of consumers. Chapter VII considers how consumers, with the help of consultants and the fiscal agent, fulfilled their role as employers. Chapter VIII discusses monitoring and the lack of both abuse of the cash allowance and consumer neglect and exploitation. Chapter IX considers whether the demonstration was implemented as planned, summarizes lessons about the components of the program discussed in Chapters IV through VIII, and describes the lessons of the New Jersey experience that cut across components of the program. Finally, Chapter X looks at New Jersey's plans for the future and for an ongoing Cash and Counseling program.

# II. PERSONAL PREFERENCE: ISSUES AND DECISIONS

The solicitation for demonstration proposals provided the basic outline for the Cash and Counseling Demonstration projects. It stipulated that a cash allowance was to be provided instead of traditional Medicaid PAS or home- and community-based waiver services (HCBS) and was to cover goods and services that would promote independence. Furthermore, the demonstration would provide support services to help a consumer manage the cash allowance and act as an employer. New Jersey called its counselors "consultants," so hereafter we refer to them as such and use the term "consulting" to refer to their work. New Jersey named its Cash and Counseling Demonstration program "Personal Preference."

New Jersey had to make many design decisions to flesh out the program that the state proposed in response to the solicitation. The experience of consumer-directed programs in the United States and abroad influenced these decisions. In addition, New Jersey made many of the decisions in consultation with the other states participating in the Cash and Counseling demonstration and the staff of the demonstration's National Program Office.

Most of New Jersey's made most of its decisions were made during the design phase of the demonstration--that is, beginning in early 1997 (when the Robert Wood Johnson Foundation [RWJF] provided the funding for New Jersey to design its demonstration program) through spring 1998 (when the state began to implement Personal Preference). However, a legal dispute over the selection of the organization to provide fiscal services kept the program from enrolling beneficiaries until December 1999. This delay did create problems, as some data collection and training activities had started months before enrollment began. Table II.1 lists key dates for the design phase and the first months of operation.

In this chapter, we first describe the state's goals for its Cash and Counseling program and the reaction of stakeholders in the state to the prospect that a cash allowance would be available instead of Medicaid personal assistance. The rest of the chapter describes the major issues that arose in designing the Cash and Counseling Demonstration and New Jersey's approach to addressing them.

TABLE II.1. Key Dates of Design and Early Operations of the Cash and Counseling Demonstration in New Jersey				
Date <sup>a</sup>	Event			
February 1996	New Jersey submitted a proposal to RWJF			
November 1996	Submitted a revised demonstration budget to RWJF			
January 1997	New Jersey funded to design its program			
April 1998	Began training consultants			
July 1998	Signed contract for enrollment services			
August 1998	Issued solicitation for a fiscal agent			
January 1999	Submission of data by personal care agencies begins			
November 1999	Enrollment begins			
When the document reviewed indicated only the quarter in which an event				

# A. New Jersey's Personal Assistance Programs and Goals for the Demonstration

occurred, we have given the date as the middle month of that quarter.

New Jersey's goals for the Cash and Counseling Demonstration are best understood within the context of the personal assistance and HCBS waiver programs it was already providing.

# 1. New Jersey's Medicaid Personal Assistance Programs

Early in 1996, when it applied for the Cash and Counseling Demonstration, New Jersey offered personal assistance to eligible Medicaid beneficiaries under its state Medicaid plan benefit, Personal Care Assistance (PCA). The state also offered seven Medicaid home- and community-based waivers. At that time, PCA served about 12,000 beneficiaries annually statewide. Each of the seven waiver programs was offered statewide; together, they served about 4,000 beneficiaries annually. The largest of the waiver programs, Community Care Program for the Elderly and Disabled (CCPED), served about 3,000 of these beneficiaries. The largest of the other six waivers was the AIDS Community Care Alternatives Program. The remaining five waivers served children and adults who were blind and disabled (three separate waivers), children who were medically fragile, and those who had experienced traumatic brain injury.

Most of the recipients of services under these seven waivers did *not* also receive PCA under the state Medicaid plan. The CCPED waiver (serving about three-fourths of all

New Jersey recipients of waiver services) was for individuals with incomes above the usual Medicaid threshold who were not eligible for any other Medicaid services. Other HCBS waivers were intended to help people who needed medically involved care (for example, the waivers for people with AIDS), many of whom did not receive personal assistance as a waiver service.

In addition, at the time that New Jersey applied for the Cash and Counseling Demonstration, it offered one state-funded, consumer-directed personal assistance program, the Personal Assistance Services Program (PASP).9 Operating through Independent Living Centers or a county coordinator (as designated by the county), PASP serves adults with physical disabilities who can direct their own care and who are either working, looking for work, or in school. PASP coordinators help clients develop their own service plans and train them to self-direct their care. While PASP clients are referred to a personal care agency, they can refuse the workers the agency assigns to them and instead recruit their own personal assistance workers. In either case, PASP clients are responsible for directing their workers in such tasks as housekeeping, personal assistance, and communication. The PASP population is similar to part of the Cash and Counseling population, and several of the worker-related features of PASP are similar to those of the Cash and Counseling model. However, PASP does not offer the flexibility to purchase equipment and supplies and home modifications. Moreover, PASP is limited in size, in part because it relies exclusively on state funding. When New Jersey applied for the Cash and Counseling Demonstration, PASP served about 500 beneficiaries statewide. Beneficiaries were typically on the PASP waiting list from 6 to 18 months before they were accepted, which suggested that some beneficiaries found this model of care attractive.

# 2. Goals for the Cash and Counseling Demonstration

In applying for the Cash and Counseling Demonstration, New Jersey sought to expand the options it offered Medicaid beneficiaries--especially elderly beneficiaries--to include programs offering greater opportunity for consumer direction. If feasible, the state wanted to recognize the competence of its citizens by expanding their choice and giving them greater control over their services.

The state wanted to participate in the demonstration primarily to test the feasibility of including a cash allowance model of consumer direction as a state-plan Medicaid service. It wanted to investigate the key issues involved in implementing such a program in

Page 12

<sup>&</sup>lt;sup>9</sup> During the demonstration, New Jersey began another consumer-directed program, Jersey Assistance for Community Caregiving (JACC). Funded entirely with state funds, JACC is designed for people with income and assets above the regular Medicaid thresholds. It allows people to hire their relatives (other than a spouse) as caregiver.

preparation for adding a cash allowance option to its regular PCA program, should the cash allowance prove workable.

New Jersey was also interested in how consumers could use the Cash and Counseling allowance to purchase equipment that Medicaid did not cover. The state office responsible for the Cash and Counseling proposal was keenly aware of equipment needs, since one of its responsibilities was to advise citizens with disabilities about equipment that would best meet their needs.

When it entered the Cash and Counseling Demonstration, New Jersey had a shortage of personal care aides, especially in rural and affluent areas, and was considering alternate ways to provide services. In particular, senior state officials were interested in making aide work more attractive by "professionalizing" it--that is, by reforming the Nurse Practice Act so that aides could perform more professional tasks, rather than primarily housekeeping chores.

Initially, overcoming an aide shortage was not one of New Jersey's primary objectives for participating in Cash and Counseling. During the demonstration, however, the worker shortage became more serious, and some program staff members began to stress the value of expanding the labor supply by allowing consumers to hire relatives and friends under the Cash and Counseling model.

New Jersey did *not* enter the Cash and Counseling Demonstration to generate savings of state funds. However, the state believed that the cash allowance might be more cost-effective than traditional personal care because it allowed consumers to use commercial services, for which competition helped to hold prices down. Moreover, the state believed that the consumer's ability to hire multiple workers at different skill levels would foster cost-effectiveness because it would create a better match between the demands of the job and worker skills and compensation.

#### **B.** Reaction of Stakeholders

Both governmental and private stakeholders had important roles as New Jersey was designing its Cash and Counseling program.

#### 1. Governmental Stakeholders

Because it was the state host agency, the New Jersey Department of Human Services (DHS) was a major stakeholder in the Cash and Counseling Demonstration. DHS, Division of Medical Assistance and Health Services (DMAHS), submitted the Cash and Counseling Demonstration proposal and applied for and received the waivers needed for the demonstration. The director of DHS/DMAHS delegated authority for administration

of the program to the New Jersey State Office on Disability Services (ODS) within DHS. The executive director of ODS became the project director for the Cash and Counseling Demonstration. As of July 1999, ODS became the Division of Disability Services (DDS); we refer to it as DDS hereafter.

New Jersey's governor was not directly involved in designing the Cash and Counseling Demonstration. However, the Commissioner of Human Services, the cabinet-level official leading DHS, strongly supported the demonstration. The DMAHS director, who felt that the cash model had wide applicability, also strongly supported the demonstration.

A few months after receiving demonstration funding, the state formed an interdepartmental work group to help implement the Cash and Counseling Demonstration. Among those in the work group were staff members of the PASP program, who shared their expertise on self-directed care. The work group also included a representative from the Office of Information Services (OIS) within DMAHS. Staff members of OIS developed and implemented project-specific software for New Jersey's Cash and Counseling Demonstration to track eligible beneficiaries, demonstration enrollees, and cash recipients, and they worked with a contractor to identify cash recipients on the state's Medicaid Management Information System.

The state legislature in New Jersey has supported community care in general, but it was not a major stakeholder in the Cash and Counseling Demonstration. As there was no need for revision of state legislation to implement the demonstration, DHS did not involve the state legislature.

Nor were unions a major stakeholder in New Jersey during the design of the demonstration, so union concerns did not shape the procedures of the Personal Preference program. However, unionization of personal care began in New Jersey while the Cash and Counseling Demonstration was under way.

#### 2. Private Stakeholders

DDS involved advocate organizations and providers of home- and community-based services in the design of the Cash and Counseling Demonstration to garner the support of private stakeholders for the demonstration. Such stakeholders were not involved in the decision to apply for the grant or the preparation of the initial proposal. However, New Jersey began to develop an advisory council before the state received funding from RWJF to design its Cash and Counseling Demonstration. After funding was assured, New Jersey quickly completed establishing the council. The state invited members of advocacy organizations for nonelderly adults with disabilities and the elderly, as well as representatives of traditional personal care agencies and other agencies providing home-

and community-based services, to join the council. The council held its first meeting in spring 1997.

#### a. Advocates

Advocates for elderly people and nonelderly adults with disabilities had no major concerns about the Cash and Counseling model and strongly supported the demonstration. During the early planning for the demonstration, the Alzheimer's Association was concerned that beneficiaries with cognitive impairment might be discriminated against by being excluded from the cash program. This concern was resolved when Personal Preference decided to allow those who could not manage the cash allowance (including those with Alzheimer's disease and other cognitive impairments) to participate if they had a representative to plan and arrange care services on their behalf.

# b. Personal Care Industry

When the demonstration began, the personal care industry in New Jersey consisted of about 250 personal care service agencies, of which more than 200 were for profit. These agencies are licensed by the state to provide personal care and private-duty nursing. There were also about 50 home health agencies that served Medicare beneficiaries. The demonstration had less effect on the home health agencies than on the personal care service agencies.

According to the organizations representing the personal care industry of New Jersey, its members had mixed feelings about the Cash and Counseling Demonstration. The industry saw some advantages to a cash allowance program. First, it believed that some clients--especially adults with disabilities who could work if they had help--needed a more flexible personal care program and could benefit from a cash program. Moreover, the industry recognized that clients who were difficult to serve in traditional programs might be referred to the cash allowance program. The Personal Preference staff marketed the program to the industry by suggesting that agencies refer their "hard" cases to Personal Preference. Finally, although personal care agencies were not allowed to bid on the provision of consulting services, the industry recognized that its member agencies could become suppliers of services (such as housecleaning and chore services) to cash recipients.

On the other hand, the personal care industry did not believe that the demonstration was appropriate for less capable beneficiaries, including most elderly ones. The industry was proud of the quality of personal care it provided in New Jersey and genuinely concerned about the well-being of less capable beneficiaries. It recognized that the fiscal agent and consultants helped to safeguard the allowance and the well-being of cash recipients. Nevertheless, the industry was concerned about potential abuse of the cash

allowance by consumers and possible exploitation of consumers by family members who were hired as workers but did not provide agreed-upon care. In addition, the industry was concerned about the potential for increased medication errors because nurses would not supervise workers who had been hired with the cash allowance.

The personal care industry also was concerned about the effect of a cash allowance program on its business. To some extent, the industry was concerned about losing clients to the Cash and Counseling Demonstration. However, its major concern was that cash recipients might hire their agency aides, thereby reducing the agency labor force--at a time when agencies already faced a shortage of personal care workers--making it more difficult for agencies to serve their clientele as a whole. This concern was heightened because a consumer might offer an aide a higher hourly wage than the agency would pay.

The industry took the position that consumer hiring of agency aides was illegal. They argued that agencies in New Jersey must invest considerable time and resources in training and certifying aides and in completing hiring paperwork. Referring to the demonstration, one industry representative put the concern this way: "As I see it, we have to do the paperwork and then they'll take the client and the client will take the aide with them." Without concurring with the industry position that it was illegal for consumers to hire their aides, the state Personal Preference staff did respond to the industry's concern by adopting a policy that discouraged consumers from hiring agency aides, including writing to cash recipients to discourage such hiring. Turnover of agency aides may have limited such hires, since consumers were unlikely to try to hire aides with whom they had not developed personal relationships. Another factor that limited such hires may have been that it was generally not in an aide's interest to give up the greater income and security of agency employment to work for one individual. Moreover, if an aide worked exclusively for a consumer, he or she would not be in a position to maintain the certification required to work for an agency and might have a difficult time finding other work if and when the consumer no longer needed personal assistance.

-

<sup>&</sup>lt;sup>10</sup> Perhaps the impact of the Cash and Counseling Demonstration on their revenues was dwarfed by a general reduction in hours of personal assistance rendered. Industry representatives reported that average hours of personal care dropped from 22 to 16 hours per week during the demonstration. Industry representatives attributed the drop to state efforts to stem growth in the PCA program following cutbacks in Medicare home health care mandated by the federal Balanced Budget Act of 1997. Senior program staff reported that state policy on the provision of PCA services did not change during the demonstration, but noted that hours of personal care likely fell as a result of the state's utilization review activities. As part of these activities, the state reviewed care plans for clients of agencies that were providing appreciably more hours of personal care per client than the typical agency (that is, "outliers"). As a result of these reviews, care plan hours (and, thus, hours of service) were reduced for some beneficiaries.

The personal care industry also was concerned that agencies be given adequate notice that a personal care client was to become a cash recipient. In response, New Jersey agreed to provide notice 30 days before a consumer switched to cash.<sup>11</sup>

In addition, the industry was concerned about the costs to agencies of supplying information needed to administer the cash allowance program. Agencies were required to complete a consumer data form) that included information from the care plan of every client they assessed. The state used the information on care plan hours from the form to determine the amount of the cash allowance to which a consumer would be entitled. In addition, personal care agencies had to provide a copy of their most recent assessment for clients who were assigned to the treatment group. The industry perceived submission of this information as a condition of participation in the PCA program and unsuccessfully sought to have the state reimburse these costs. Moreover, as the enrollment period for the demonstration was extended again and again, the personal care industry became frustrated that its members had to submit consumer data forms and copies of assessments for far longer than originally planned.

Finally, the personal care industry resented the fact that the extensive regulations that the state had imposed on its agencies would not apply to workers hired directly by consumers. Industry representatives described New Jersey personal care as "very regulated," with supervision, certification, and time-consuming criminal background checks involving the FBI, state police, and state nursing practice board. None of these regulations applied to workers hired with the cash allowance. Case Example II.1 presents the explanation one industry representative gave for the resentment of differences in regulations.

#### Case Example II.1: Resentment of Differences in Regulations

Under the Cash and Counseling program, the state says, "We'll wipe that out for Cash and Counseling, but *you* still have to do it." Moreover, the state does not pay us to comply with the regulations. We have had only one rate increase in 10 years, and that was 50 cents an hour. The state created the rules, now the state won't have to follow them.

The industry did not substantially impede the implementation of the Cash and Counseling Demonstration, partly because the state Personal Preference project staff carefully considered the industry's concerns and responded to some of them. Indeed, the industry supported the demonstration with timely submission of consumer data forms and copies of assessments and by referring clients. Organizations representing the personal care industry also agreed to notify member agencies about the evaluation's worker survey and encourage them to cooperate with it.

\_

<sup>&</sup>lt;sup>11</sup> The notice period was later reduced to 14 days.

Moreover, the industry greatly respected the project director of Personal Preferencethey knew him well through his work as executive director of DDS. This respect made the industry more willing to cooperate with the Cash and Counseling Demonstration. Case Example II.2 presents the description one industry representative gave of this respect and its importance to the demonstration.

#### Case Example II.2: Trust and Respect for DDS Executive Director

The state has good people trying to make the program work fairly for everybody. The person chosen to head this up has made a positive difference. He has experience with disability, is open, fair, has a history in personal care and community-based services, and knows what clients need. He is the least bureaucratic of the state people. He sincerely cares. We knew we could trust him to be honest. If it were not for him, we would have hated the program more.

Some of the industry's most serious concerns were not realized. There were no instances of serious abuse of the cash allowance or of exploitation of consumers. Agency labor supply was affected very little, because consumers hired few agency aides. In addition, some of those hired may have continued to work for agencies while working for cash recipients "on the side," perhaps without the knowledge of their agency employers.

# C. Design Issues and Decisions

Decisions of the funders (RWJF and DHHS/ASPE) and federal law and regulations (including the terms and conditions of the demonstration waivers) set parameters for the Cash and Counseling Demonstration. Within these parameters, New Jersey made many decisions that shaped the Personal Preference program. As noted earlier, these decisions were influenced by the experience of other consumer-directed programs and shaped by the work of the Cash and Counseling National Program Office. Many were made jointly with the other Cash and Counseling programs. These decisions involved all of the major components of the Cash and Counseling model: eligibility and appropriateness, outreach and enrollment, services to be covered by the cash allowance, determination of the level of the cash allowance, and consulting and fiscal services.<sup>12</sup>

# 1. Eligibility and Appropriateness

The solicitation for the Cash and Counseling Demonstration permitted states to cash out either Medicaid state-plan PCA or Medicaid home- and community-based waiver services.

<sup>&</sup>lt;sup>12</sup> For a discussion of the decisions the Cash and Counseling states faced, see Mahoney et al. 2000.

As mentioned earlier, New Jersey had both state-plan PCA and waiver services when it responded to the solicitation. It proposed to cash out only state-plan PCA. Later, under pressure to expand the pool of eligible beneficiaries to increase the likelihood of reaching the targets for enrollment needed for the Cash and Counseling Evaluation, New Jersey considered cashing out some of its home- and community-based waivers. However, only one waiver program--CCPED--served enough clients (about 3,000 annually) to materially expand the pool of eligible beneficiaries. The senior program staff was reluctant to cash out the CCPED waiver, for two main reasons. First, state-plan PCA participants are entitled to other Medicaid benefits, while CCPED clients are entitled only to the services the CCPED waiver covers. Second, authorization procedures for stateplan PCA and CCPED waiver services differed in that services authorized under CCPED (and most of the other waivers) were subject to cost caps. Because of these differences, CCPED participants were expected to use a different proportion of their care plan services than were PCA participants, making it more complex to cash out the care plan while ensuring that the cash program would cost no more (on average) than did services actually delivered. Moreover, those responsible for the CCPED program apparently had little interest in participating in the Cash and Counseling Demonstration. Ultimately, New Jersey decided to cash out only state-plan PCA services under its Personal Preference program.

Both current recipients of PCA services and PCA applicants were eligible to enroll in Personal Preference, except that New Jersey excluded beneficiaries who were receiving both PCA and home- and community-based waiver services. New Jersey was concerned that including recipients of both PCA and waiver services would cause confusion, as these beneficiaries would have both Cash and Counseling consultants and waiver case managers. Moreover, New Jersey was concerned that, if beneficiaries receiving both PCA and waiver services were included in the demonstration, the difference in authorization procedures for PCA and waiver services would make it difficult to interpret the results of the evaluation. As the number of beneficiaries receiving both PCA and waiver services was small, their exclusion had little effect on the size of the eligible population.

Given the decision to cash out state-plan PCA, New Jersey needed to decide which, if any, beneficiaries of that program would be inappropriate for Personal Preference and how to identify them so that they could be excluded. From the beginning, New Jersey wished to exclude as few PCA beneficiaries as possible. So it could more readily meet the target sample sizes for the evaluation, the state wanted to retain the largest possible pool of eligible participants. Moreover, New Jersey pointed out that those not capable of

<sup>&</sup>lt;sup>13</sup> PASP recipients wre ineligible for Personal Preference.

<sup>&</sup>lt;sup>14</sup> Later, New Jersey allowed beneficiaries who received both PCA and services under the waiver for developmentally disabled adults to participate in the demonstration. This waiver was the only one of the seven that did not impose cost caps on the care plan. Five such beneficiaries enrolled in Cash and Counseling.

managing the cash allowance for themselves could have representatives. New Jersey realized that developing a structured screening process would be problematic. For example, a history of alcohol abuse did not necessarily mean that a beneficiary was currently inappropriate for the cash allowance program.

The decision about a structured screening process offers a good example of the work of the National Program Office and the collaboration of the Cash and Counseling states during the design phase of the demonstration. All of the demonstration states shared New Jersey's concerns about a structured screening process. The National Program Office for Cash and Counseling formed a task force on screening. While deliberations on screening were under way, an attorney on Arkansas's advisory council said that a structured process that denied participation might not be legally defensible. If the process was not legally defensible and a consumer chose to contest exclusion from the program, a state might be held liable for such exclusion.

Ultimately, New Jersey decided to implement only a single criterion pertaining to appropriateness for Cash and Counseling. PCA recipients who were not expected to be living in a community setting for at least six months were excluded on the grounds that consumers typically require several months to develop a cash management plan and hire workers. Thus, those who were not expected to be living in a community setting for at least six months were not likely to benefit from the cash program. To implement this criterion, the consumer data form asked personal care agency nurses to indicate whether they expected PCA services to be required for at least six months.

Except for this one structured criterion, New Jersey relied on a self-screening process for Personal Preference for both consumers and representatives. The materials provided to consumers described their responsibilities, as well as their rights, under Personal Preference and stated that only those who were willing to assume these responsibilities were to enroll in the cash program. Similarly, New Jersey developed materials describing the responsibilities of the representative and asked that potential representatives review these materials before making a commitment to serve in that capacity. (Legal guardians and other court-appointed representatives were expected to serve as Personal Preference representatives if the person for whom they acted enrolled in the program.) Appendix A presents these materials, and Chapter VI discusses representatives in detail.

While New Jersey did not adopt a formal process to determine whether a given person was appropriate as a representative, it did specify that the same person could not serve as both a worker and a representative. Serving as both could create a conflict of interest, since the representative's responsibilities would typically include signing worker time sheets and supervising worker activities.

#### 2. Outreach and Enrollment

Outreach to the community was necessary to create awareness of the Cash and Counseling Demonstration and to increase interest among beneficiaries in receiving a cash allowance, as well as interest from agencies in providing consulting services. The state Personal Preference staff was responsible for the marketing phase of outreach. Most marketing involved presentations to potential participants, as well as to advocacy, community, and professional groups.

New Jersey had to make two related design decisions about enrollment: (1) when it would take place, and (2) who would be responsible for it. These decisions in turn affected procedures for outreach to individuals. The basic options for *timing* of enrollment were to allow consumers to enroll at any time or to require them to enroll only at the time of their assessment for PCA--that is, as ongoing recipients were reassessed to determine the need for continuing care and as new applicants were assessed. The basic options for *responsibility* for enrollment were for state employees (or state contractors) to contact beneficiaries or to rely on organizations already serving PCA recipients.

Enrollment at assessment had two advantages. First, the care plan developed from an assessment was to be the information source needed to determine the amount of the cash allowance. If enrollment were timed to follow assessment, that care plan would remain in force for about six months (barring a material change in the beneficiary's condition or circumstances requiring a new assessment). If enrollment were divorced from assessment, a new care plan might be required before the consumer could complete the development of the cash plan. Second, enrollment at assessment would spread the volume of enrollment-related work out over time, making the workload for enrollment staff more manageable. If consumers could enroll at any time, the enrollment staff might find it impossible to meet demand at peak times--especially following a major outreach effort. On the other hand, restricting enrollment to the time of assessment could be problematic if outreach could not target those who were about to be assessed or had been assessed recently. If beneficiaries who expressed interest in Personal Preference had to wait several months before being allowed to enroll, they might lose interest in the cash program altogether.

As noted earlier, the state might vest responsibility for outreach in state employees or contractors (contractors might be independent assessors) or work with other organizations, including providers of traditional services. Providers of traditional services might readily couple enrollment with assessment, as the latter is often among their responsibilities. For example, those conducting assessments might explain the cash program and leave reading material about the program with the consumer. Consequently, relying on traditional providers to conduct outreach might be more efficient than hiring state employees or a state contractor--other things being equal.

However, objectivity, as well as efficiency, was an important consideration in selecting an approach to enrollment in Personal Preference. Staff members at traditional agencies who were resistant to the cash program might influence consumers not to participate. In addition, independent assessors might perceive the cash program as a threat to their professional norms (for example, norms about the need for nurses to supervise the administration of medication).

Concerned about the possible resistance of traditional personal care agency staff to Personal Preference, New Jersey decided not to give responsibility for either outreach or enrollment to these providers. Instead, the state vested responsibility for outreach and enrollment in its own staff and an external contractor. Thus, it separated outreach and enrollment from the assessment process for which traditional personal care agencies were responsible.

Initially, New Jersey rejected hiring state employees for other aspects of outreach or for enrollment. Instead, it contracted with a for-profit firm for these tasks. There were three major arguments against hiring state employees for outreach and enrollment. First, New Jersey would be able to hire only a small state staff for Personal Preference. With a small staff, it would be nearly impossible to have staff members fluent in Spanish and Russian, foreign languages spoken by a substantial number of Medicaid beneficiaries in New Jersey. Second, it would be difficult for a small staff to visit the homes of consumers throughout the state. Third, New Jersey was concerned about the possibility of lengthy delays (of six months or more) in hiring state employees. It did not want to be forced to postpone the implementation of the demonstration due to delays in hiring state employees for enrollment. However, many months after implementing the demonstration, New Jersey hired state employees to take over the outreach and enrollment responsibilities from the for-profit firm. (Chapter IV provides a detailed description of Personal Preference outreach and enrollment.)

Although New Jersey separated responsibility for assessment and enrollment, it initially timed enrollment to follow assessment. State receipt of the consumer data form for a given beneficiary (submitted after assessment) triggered outreach to see if that person was interested in enrolling in Personal Preference. New Jersey rejected the idea of having assessment nurses leave brochures about Personal Preference with the consumer at the time of the assessment home visit. The state did not want these nurses to respond to questions about the cash program because it feared they would not answer objectively, and the agencies pointed out that it did not make sense for their nurses to leave brochures if they were not allowed to respond to questions consumers might have about Personal Preference. Later, New Jersey mailed materials about Personal Preference to all consumers listed on a database developed from the consumer data forms who had not previously expressed interest in the cash program (see Chapter IV).

#### 3. Permissible Uses of the Cash Allowance

The solicitation for the Cash and Counseling Demonstration did not envision that a state would provide unfettered cash, as in an income supplement program. However, it insisted that states permit the cash allowance to cover a range of goods and services that would help consumers function more independently.

Two key issues that arose were (1) the breadth of goods and services that could be purchased with the cash allowance, and (2) the method (if any) by which their purchase would be authorized. New Jersey could opt for a relatively narrow list of covered goods and services, or it might interpret coverage broadly. Furthermore, New Jersey could require state Personal Preference staff to review and approve cash plans or delegate authorization to do so to consultants.

In developing an approach to covered goods and services, New Jersey was torn between concern about the possibility of abuse of the cash allowance and pressure from the National Program Office, RWJF, and DHHS/ASPE to broaden the goods and services covered under the cash allowance. Facing skeptics and critics who initially equated cash with abuse, New Jersey was concerned about the potential harm to the cash program if abuse of the cash allowance led to adverse publicity. The director of Personal Preference recalled that he dreaded the possibility of a newspaper comment that a consumer "went to Atlantic City on the money."

New Jersey decided to use the opportunity afforded by the demonstration to develop lists of approved goods and services for the demonstration and of unapproved goods and services for use in a future program (if the cash allowance program were continued beyond the demonstration). To maintain control over the approval process during the demonstration, New Jersey decided that the state Personal Preference staff would be responsible for reviewing and approving cash management plans.

The Personal Preference program staff did not want to give consultants the power to approve cash plans because it was concerned that doing so would place consultants in an adversarial role and damage their relationships with consumers. New Jersey preferred that the state be the "bad guy" when it was necessary to disapprove a consumer request to include a good or service in the cash plan.

Another issue was who would be eligible to be hired as a worker. The waivers for the Cash and Counseling Demonstration permitted payment of legally responsible relatives-that is, spouses and parents of minor children. New Jersey decided to avail itself of this provision of the waivers--it would allow consumers to hire their spouses with the cash allowance. (No cash recipients in New Jersey were minors, as the PCA program did not serve minors.)

As a cash management plan is being developed, the availability of goods and services from other public sources must be taken into account. New Jersey (and the other Cash and Counseling states) recognized that it was not in the consumer's interest to use the cash to purchase services already available at no cost (or reduced cost) to the consumer. The consumer would incur an unnecessary expense if he or she purchased such goods or services. In addition, another program might be better able to help the client choose the most appropriate good or service. For example, a program specializing in assistive equipment could offer expert advice on what type of equipment was best under different circumstances. To address this issue, New Jersey had Personal Preference consultants advise consumers about the goods and services available under other public programs during the development (and any later revisions) of the cash management plan. In addition, staff members at DDS were knowledgeable about technology to help people with disabilities and made their expertise available to the Personal Preference consultants.

## 4. Determining the Level of the Cash Allowance

In cash programs, the level of allowance may vary, depending on the consumer's level of need (or a related concept such as disability) or on his or her history of benefits in the traditional program. One advantage of basing the level of the cash allowance on the historical benefit level is that doing so reduces the likelihood that demand for PAS will increase because beneficiaries not interested in agency services apply for the cash allowance because they find it attractive, thereby increasing public costs. Such induced demand is sometimes labeled the "woodwork effect." 15

On the other hand, basing the level of the cash allowance on the historical level of benefits is problematic, for two reasons. First, the condition of the beneficiary or his or her situation can change--perhaps dramatically; the historical level of benefits may be inadequate in such situations. Second, those not already enrolled in the traditional program will not have a track record of historical benefits on which to base the amount of the cash allowance. If beneficiaries are to participate in the cash program without first participating in the traditional program until they establish a claims history, the amount of their allowance must be based on need, and a mechanism must be developed outside of the traditional program for assessing that need.<sup>16</sup>

New Jersey based the level of the cash allowance on approved PCA care plan hours. Consumers were not required to have a history of PCA benefits, but they could not enter Personal Preference directly. To be eligible for Personal Preference, a Medicaid

<sup>&</sup>lt;sup>15</sup> To prevent an increase in public costs due to the woodwork effect, the special terms and conditions for the waiver for the Cash and Counseling Demonstration stipulated that the ratio of the number of new clients to the number of current clients was not to exceed a historical average.

<sup>&</sup>lt;sup>16</sup> Level of need might be based on professional judgment or on an algorithm that takes into account the beneficiary's functional impairment and informal care.

beneficiary had to be a current recipient of PCA or be assessed and approved for PCA by a traditional agency, which then developed a care plan. However, to reduce the potential for demand induced by the attractiveness of a cash allowance, New Jersey deliberately focused outreach for Personal Preference on *current* recipients of PCA services.

Channeling enrollment of those not already receiving PCA through traditional agencies had another advantage. By doing so, New Jersey facilitated arrangement of traditional services on an interim basis (if needed) while beneficiaries developed their cash management plans.

The formula for determining the level of the cash allowance in New Jersey involved calculating the cost to the state of the hours in the PCA care plan (separately for weekday and weekend hours as the hourly cost differed), then deducting 10 percent to cover the costs of Personal Preference consulting and part of the cost of fiscal services. (Consumers were also charged for certain services provided by the fiscal agent if they used these services, such as cutting checks and conducting criminal background checks). Information on hours in the PCA care plan was obtained from the consumer data form, which was completed by staff of traditional agencies and instituted expressly for this purpose.<sup>17</sup> The hours on the care plan the traditional agency prepared indicated the number of hours the agency planned to deliver. However, if the traditional agency planned to deliver more than 25 hours of care a week, it first had to obtain authorization from the state.

When using care plans to set cash allowance levels, a major issue is the potential for differences between the hours of traditional service planned and the hours actually delivered. The hours of service delivered generally do not exceed the number planned, since the care plan typically represents the maximum authorized. Rather, the number of hours of care received is often less than the hours planned, for several reasons. Agencies sometimes may plan for somewhat more care than they expect to provide so that they can increase the amount without revising the care plan. That is, total hours planned may include a hedge against possible future increases in need. More frequently, events preclude providing all the hours of care planned. A client may be unexpectedly hospitalized and thus not available when an aide arrives. A personal care aide may not appear for work when expected. An agency may be unable to find enough workers to provide the planned care. If the worker shortage is prolonged, an agency may reduce care plan hours to lessen consumer expectations.

Because the number of hours of service received may be less than the hours planned, a "discount rate" may be required to ensure budget neutrality--that is, it may be necessary to discount the hours in the care plan before cashing them out to ensure that the

Page 25

-

<sup>&</sup>lt;sup>17</sup> Prior to requiring the consumer data form, New Jersey did not have ready access to data on care plan hours, as it was necessary for the state to ask agencies for data on care plan hours on a case-by-case basis.

cost of the cash program does not exceed the cost of the traditional program. Such budget neutrality was a condition of the Cash and Counseling Demonstration waivers.

To assess the need for a discount rate, New Jersey collected information on care plan hours for a historical sample of PCA clients and compared those hours to hours of care received, as indicated by Medicaid claims. As there was little difference between hours planned and hours received for the sample, New Jersey decided against discounting care plan hours for the cash allowance.

# 5. Consulting and Fiscal Services

In designing Personal Preference, New Jersey faced several major issues concerning consulting and fiscal services. It had to decide what consulting and fiscal services would be offered, how these services would be organized, and how they would be paid for and monitored.

## a. Organization of Consulting and Fiscal Services

New Jersey had several options in deciding how to organize assistance for consumers receiving the cash allowance. First, it could either combine consulting and fiscal functions in a single host organization or separate these functions, assigning them to different organizations. The major argument for separating consulting and fiscal functions is the difference in the types of expertise required, especially expertise for preparing and filing payroll taxes and related documents.

Second, New Jersey could choose to have one host organization serve the entire state (for consulting, fiscal services, or both) or to have multiple host organizations-- (perhaps serving different regions or different populations). Having one host organization might take better advantage of economies of scale. If the consumer is to be able to choose among providers of assistance, however, more than one host organization must be available. Moreover, a single host organization might not be able to cover all regions of the state or serve all population groups equally well. Finally, the state could opt for different numbers of host organizations for consulting and for fiscal services.

New Jersey chose to separate the fiscal and consulting functions. It contracted with a single organization to provide fiscal services for the entire state and established agreements (called "partnerships") with 34 organizations to provide consulting services across the state. (Agencies providing traditional personal care services were not permitted to provide consulting services.) New Jersey has a diverse population, and the state wanted to have a variety of human services agencies providing consulting to different ethnic communities. It also wanted to give consumers a choice of agencies that provide consulting. Moreover, the consultants needed to be located throughout the state, since they were required to visit consumers' homes periodically. Having a large number of

organizations providing consulting also helped to ensure that the human services organizations in the state would support Personal Preference.

Given the large number of consulting agencies, it was expected that each would typically have a small volume of Personal Preference cases. Agency staff members were expected to continue to have other duties and to provide consulting part-time or even after hours.

New Jersey judged that it would be inefficient for agencies serving only a few consumers to provide fiscal services under Personal Preference. To keep costs as low as possible, a fiscal agent needs to process a volume of checks and payroll documents. It might also be inefficient if multiple organizations were required to become expert in acting as the agent of the consumer. Moreover, the fiscal agent does not need to be located close to the consumer, since its activities can be handled primarily by telephone and mail. Finally, New Jersey would have had to devote more resources to monitoring if it selected multiple host organizations to provide fiscal services.

To provide checks and balances, New Jersey also separated the consulting and fiscal functions. The Personal Preference program staff was concerned about increasing the potential for abuse if staff of the same organization helped to develop the cash plan, then approved the checks under that plan. For example, such abuse might arise if a consumer hired a member of the consultant's family as a worker at the consultant's urging. Arguably, staff of a separate organization might be more likely than staff in the same organization to question the appropriateness of payment in a situation such as one in which the payee had the same last name as the person reviewing the cash plan.

# b. What Consulting and Fiscal Services to Offer

In the Cash and Counseling model, many aspects of consulting and fiscal services are intended to help consumers and may be used at their discretion. Others are intended to prevent abuse of the cash allowance and exploitation of the consumer and are mandatory. New Jersey had to decide which aspects of consulting/fiscal services would be offered and which would be mandatory. In particular, consultant assistance with the development of the cash management plan might be discretionary, while consultant review of the uses of cash might be mandatory. Consultant assistance with recruiting, hiring, training, and supervising workers might be discretionary, but New Jersey might require that the fiscal agent prepare the appropriate tax and unemployment insurance forms (or ensure that the consumer did so). Furthermore, during the design of the Cash and Counseling program, New Jersey had to decide whether a consumer who wished to manage the cash allowance without help from the fiscal agent would be required to pass a skills examination on preparation of payroll documents to ensure that federal and state tax liabilities would be met. Finally, New Jersey had to decide how frequently consultant contact with consumers would be required to prevent abuse and exploitation.

In addition to these basic services, a consulting agency or fiscal agent might perform other services for consumers. These include maintaining a worker registry to help consumers identify workers to hire, assisting in securing criminal background checks for potential workers, keeping staff to serve as back-up workers if a consumer's regular worker were unable to care for him or her, and hosting a peer support group. Consultants might also be responsible for enrollment in the cash program and for reassessment of the care needs of participants in the cash program.

The New Jersey design for Personal Preference made three aspects of consulting/fiscal services mandatory. First, the consultant had to review (but not approve) the initial cash management plan and all subsequent revisions to it. A revised plan was required when the amount of the cash allowance changed or if the consumer proposed changes in the use of cash that were inconsistent with the current cash plan. After consultant review of the cash management plan, the consultant was to forward the plan to the state Personal Preference program staff in Trenton for approval.

Second, consultants were required to visit consumers face-to-face at least quarterly and speak to them at least monthly for the first six months after their assignment to the treatment group. In this way, they could monitor consumers' circumstances and condition and their use of the cash allowance. New Jersey did not require consumers or representatives using the fiscal agent to retain receipts to document expenses for review by consultants. The state reasoned that such consumers had direct control of only small amounts of cash for incidental expenses, which were limited to no more than 10 percent of the allowance. Moreover, any use of the cash allowance for incidental expenses had to be included in the cash management plan.

The third mandatory feature New Jersey adopted was fiscal review for consumers who chose not to use the fiscal services for payroll taxes and related documents. (This decision was made in conjunction with the other Cash and Counseling programs and the National Program Office.) Consumers who chose not to use fiscal services were required to pass a skills examination on the preparation of payroll documents and present their payroll records for review by the fiscal agent if requested to do so.

Optional consulting services for participants in New Jersey included:

- Training and support services related to the development of cash management plans.
- Training and support services related to recruiting, hiring, training, and firing workers.
- Contacting the fiscal agent or the state Personal Preference office on the consumer's behalf.

- Providing referral information on services that might assist the consumer, including
  public services (such as transportation from the local Area Agency on Aging) and
  private services (such as an insurance agent who would write a rider for workmen's
  compensation to a renter's policy).
- Making initial contacts with service providers on the behalf of the consumer.
- Identifying equipment, assistive technology, and other goods that could increase independence.

The amount of optional consulting service available to a given consumer was limited by the system New Jersey adopted for payment of consultants (discussed further in Subsection c below). Specifically, hourly payments to consultants were capped at no more than 19 hours a year for a given consumer, with a separate lump-sum payment for the development of the initial cash management plan. New Jersey adopted this cap on consultant hours partly to contain the costs of consulting. In addition, it wanted to avoid having consultants assume the role of case managers and thereby help ensure that the philosophy of consumer direction was indeed implemented in Personal Preference.

The consulting agencies and fiscal agent provided some services beyond the basic consulting and fiscal services. New Jersey required that consulting agencies facilitate peer support among participants in the Personal Preference program. Some of the consulting agencies did sponsor peer support programs for other groups or refer their clients to peer support programs sponsored by other organizations. However, no consulting agency that we visited reported special programs to foster peer support among the Personal Preference consumers to whom it provided consulting services. Specialized peer support programs probably were not a priority for organizations providing consulting agencies to only a small number of consumers. The Personal Preference program did develop a list of consumers who were willing to speak to interested beneficiaries about enrolling in the cash program. In effect, these consumers were engaged in a peer support activity. The fiscal agent provided criminal background checks as an optional service (consumers were charged an additional fee for this). However, neither the consulting agencies nor the fiscal agent provided worker registries or back-up workers as optional services.<sup>19</sup>

Consultants were not responsible for enrollment or reassessment in New Jersey. As described earlier, the state initially contracted with a private firm to enroll beneficiaries in

<sup>&</sup>lt;sup>18</sup> The cap was later increased to 20 hours a year for a given consumer.

<sup>&</sup>lt;sup>19</sup> After our visit to New Jersey, the state received a "Systems Change" grant for Community Living from the Health Care Financing Administration (now the Centers for Medicare & Medicaid Services). New Jersey planned to devote some of the grant funds to the development of a worker registry.

Personal Preference and later hired state employees as enrollment staff. New Jersey arranged for Medicaid nurses to reassess cash recipients.

# c. Payment for Consulting and Fiscal Services

Another issue that New Jersey faced in designing Personal Preference was how to pay for consulting and fiscal services. First, it had to decide whether consumers would be charged for nonmandatory services. For example, were they to be charged a monthly fee for bookkeeping or a service charge for each check issued? The major argument in favor of such consumer charges is that they work to ensure that those who use the services bear the cost. The major argument against such charges is that consumers might be discouraged from using a service from which they would benefit.

New Jersey opted to have consumers charged service fees for the tasks that they requested the fiscal agent to perform. A schedule of charges was developed, covering such tasks as cutting a check, stopping payment on a check, and arranging for a criminal background check.

Second, New Jersey had to decide how to structure the state's payment for consulting and fiscal services. The basic options were a fee-for-service or capitated approach. A fee-for-service approach might pay by the hour or establish payment for a particular task, while a capitated approach might pay the consulting or fiscal agent a monthly fee for every consumer on its caseload. One issue in designing a capitated approach was whether the level of services a given consumer required was likely to decrease over time as he or she (or a representative) mastered the responsibilities of an employer. Under either a fee-for-service or a capitated approach, it might be feasible to pay for some specific services under another authority. For example, reassessment might be rendered outside the cash program and paid for as a separate Medicaid service.

New Jersey decided in favor of a fee-for-service approach for consultants, with a lump-sum payment for the development of the initial cash plan (originally \$52), followed by payment of \$16 an hour (up to a maximum of 19 hours annually).<sup>20</sup> Consumers were not charged directly for consulting services. Rather, all consulting costs were paid from the funds generated by taking the top 10 percent off all "cashed out" care plans before arriving at the consumer's allowance level.

New Jersey adopted a fee-for-service approach to paying the fiscal agent. In addition to the fees paid by consumers, the state paid the fiscal agent for some services rendered to consumers. For example, the state paid the fiscal agency for processing the papers in a worker employment packet. A fee schedule was also developed for these

Page 30

<sup>&</sup>lt;sup>20</sup> The lump-sum payment was later increased to \$75 and the hourly payment for consulting services was later increased to \$26 an hour (up to a cap of 20 hours per consumer per year).

services. The top 10 percent of all "cashed out" care plans was also the source for these payments from the state to the fiscal agent.

## d. Monitoring Consulting and Fiscal Services

New Jersey planned to monitor consulting services in several ways. First, state program staff members were to review documents submitted by consultants pertaining to individual consumers (primarily cash management plans and reports on quarterly visits to consumers). Second, consultants were required to submit detailed invoices naming the consumers they had assisted. Third, consultants were required to document their contacts with consumers and to maintain a minimum data set about each consumer. Fourth, New Jersey planned to conduct on-site quality assurance audits of consulting services. As a part of those audits, state program staff members were to review the consultant's documentation. In addition, New Jersey planned to survey consumers about their satisfaction with Personal Preference, and consumers were to receive a toll-free number that they might use to report difficulties with the program. (Monitoring to ensure the quality of consulting services is considered in Chapter III.)

New Jersey planned to monitor the work of the fiscal agent in several ways. First, the fiscal agent was required to submit reports of its activities. Second, New Jersey set standards for the performance of fiscal services and mandated the notification of state program staff if these standards were not met. Third, New Jersey planned audits of the fiscal agent. (Monitoring of the fiscal agent is also discussed further in Chapter III.)

# **D.** Poised for Implementation

Working with the National Program Office, RWJF, DHHS/ASPE, and the other Cash and Counseling states, New Jersey completed the design of its Personal Preference program by mid- 1998, about a year and a half after it received funding to do so and was poised to begin implementation. By early 1998, New Jersey had developed agreements with agencies throughout the state to provide consulting and had developed a training curriculum for consultants. In April 1998, Personal Preference began to train consultants. A few months later, New Jersey contracted with an organization to provide outreach and enrollment and began the procurement process for an organization to serve as fiscal agent. Because New Jersey expected to implement the cash program early in 1999, the state required that personal care agencies begin to submit consumer data forms in January of that year. However, a legal dispute arose over the selection of the fiscal agent. This dispute was protracted and delayed full implementation of Personal Preference until November 1999. The delay caused problems. Personal care agencies were submitting information about PCA recipients that was of limited value, as the Personal Preference program could not begin to enroll participants. Some of the consultants who had been trained resigned before enrollment began, and the others needed refresher training.

Nevertheless, when the legal dispute was finally resolved, New Jersey's Personal Preference program was up and running quickly. In the next several chapters, we describe its operation.

# III. ORGANIZATIONS PROVIDING OUTREACH AND ENROLLMENT, CONSULTING, AND FISCAL SERVICES

A variety of organizations--private and public--are partners with the state of New Jersey in the Personal Preference program. These organizations provide outreach and enrollment, consulting, and fiscal services. Their selection marks the transition from the design of Personal Preference to its operation.

This chapter discusses organizations that provide outreach and enrollment, consulting, and fiscal services. We describe how New Jersey selected these organizations, the organizations themselves, and the procedures the state used to ensure the quality of their services.

### A. Outreach and Enrollment Contractor

As Chapter II indicated, New Jersey initially used an external contractor for outreach and enrollment services. However, Personal Preference was displeased with the services of the contractor and did not renew that contract. The program staff then brought outreach and enrollment activities in-house, 15 months after enrollment began. Here, we describe the initial contractor and how it was selected. In Chapter VI, we discuss the shortcomings of external contracting that led New Jersey to discontinue that service.

#### 1. Criteria

New Jersey used four major criteria in selecting a contractor for outreach and enrollment. First, it sought a contractor with a good record in human or health services programs. Outreach and enrollment staff members needed to be sensitive to the needs and concerns of elderly and disabled Personal Care Assistance (PCA) recipients and their families. Second, the state sought a contractor that already had a multilingual staff. New Jersey is an ethnically diverse state, and PCA recipients represent a variety of ethnic groups. Because many PCA recipients speak a language other than English, the state sought a contractor with staff members able to talk with them in at least Spanish or Russian (and ideally in other languages as well) to explain the demonstration and seek their participation. Third, New Jersey sought a contractor with staff located throughout the state. Personal Preference was to be implemented statewide, and outreach and enrollment staff members would need to visit the homes of potential demonstration participants in all parts of the state. New Jersey also sought to minimize travel costs by seeking an organization that had staff throughout the state. Finally, New Jersey sought a

contractor with a toll-free telephone number, as potential participants would have to telephone the outreach and enrollment contractor. To ensure that participants would be able to call without cost to themselves, New Jersey sought a contractor that had a toll-free telephone number in place (or at least that was willing to install one and maintain a staff to answer it).

#### 2. Outreach and Enrollment Contractor

As its outreach and enrollment contractor, New Jersey selected a for-profit firm already under contract with the state to enroll Medicaid beneficiaries in managed care. Headquartered in the Trenton area (in the center of the state), the firm met the state's selection criteria. It had successfully enrolled a Medicaid population in a health services program, and the state had been pleased with the pace of enrollment. The firm had a multilingual staff located throughout the state and also had a toll-free telephone number. An added benefit was that the current contract could simply be amended to cover outreach and enrollment for Personal Preference. The state thus avoided the full procurement process, which can take a long time.

Unbeknownst to the state at the time, the contractor had conducted all its previous marketing by telephone. It had no experience conducting home visits, which may have exacerbated its later problems with conducting enrollment.

# 3. Assuring the Quality of Outreach and Enrollment Services

To help assure the quality of outreach and enrollment services, New Jersey established standards, several of which pertained to the timeliness and extent of outreach and enrollment activities. The most important of these standards required that the outreach and enrollment contractor do the following:

- Contact a Medicaid PCA beneficiary within 10 days of receiving his or her name from the Personal Preference program.
- Make three attempts (on different days of the week and at different times of the day) to reach a beneficiary by telephone.
- Arrange a home visit within 10 days of a beneficiary agreement for a visit (unless unusual circumstances prevailed).
- Make two attempts to conduct a home visit.
- Forward all signed consent forms to the Personal Preference program office within one week of their receipt.

New Jersey also set a standard for the number of enrollments to be generated: 30 a week. In addition, the outreach and enrollment contractor was required to file a weekly report on the status of its activities.

# **B.** Consulting Agencies

New Jersey cast a broad net for consulting services. As Chapter II indicated, the state sought human services agencies--most of which would have a small caseload of Personal Preference participants--from across the state to provide consulting services. (However, agencies that provide traditional PCA were excluded.) The state specified minimal educational requirements for consultants and left agencies free to determine how best to staff and supervise consulting for Personal Preference. Finally, to minimize reporting burden, New Jersey streamlined standards for documentation and the reporting requirements imposed on consultants and consulting agencies.

# 1. Recruiting Consulting Agencies

To recruit consulting agencies, New Jersey relied heavily on presentations to human services executives. Shortly after the state received funding to design its Cash and Counseling program, Personal Preference staff began giving presentations at regional meetings of human services executives, with the goal of fostering interest in providing consulting services. Early in 1998, New Jersey intensified these efforts, launching a major campaign to recruit agencies to provide consulting. The director of Personal Preference wrote a letter to every health and human services organization in the state (public, private nonprofit, and private for-profit agencies), inviting the organizational leadership to regional meetings. At these meetings, Personal Preference program staff provided information about expectations and reimbursement for consulting services, as well as about the Personal Preference program in general.

New Jersey set minimal criteria for participation as a consulting agency. The agency had to demonstrate interest in Personal Preference by sending its staff to training at no expense to the state, and had to sign an agreement (in the form of a simple letter) with the state. This approach to procurement minimized the administrative effort required to participate as a consulting agency. An executive of one consulting agency reported that she was pleased that the state had made that aspect of participation so straightforward.

A total of 34 agencies from across the state signed the letter of agreement. These agencies included county boards of social services, Independent Living Centers, adult day care centers, private case management agencies, and Area Agencies on Aging (AAAs). Personal Preference program staff especially welcomed the participation of several boards of social services, because these boards have contact with many PCA recipients and are knowledgeable about the resources available in their communities. While the

number of AAAs participating as consulting agencies was smaller than the state had hoped, other agencies that serve elderly people participated.

The Personal Preference staff members were satisfied with the overall level of interest in providing consulting under Personal Preference. One agency executive speculated that more agencies--especially smaller ones--might have participated if the state had offered a higher payment rate for consulting services. (Later, New Jersey increased the payment for consulting.) Agency executives reported that while the initial consulting payments did cover the direct cost of staff time to provide consulting services, they did not cover the cost of staff benefits. Thus, if agency staff were to receive benefits for the portion of their time devoted to Personal Preference, the agency had to draw on other resources to cover the cost of those benefits. Small agencies may have had fewer resources than larger ones from which to cover these costs and thus may have found participation in Personal Preference less attractive.

The executives of agencies that chose not to participate in Personal Preference also may have been concerned about competing demands on the time of their staff. While we did not speak to the executives of any agencies that declined to participate, we did interview the staff of one agency that dropped out after working with only a few consumers. This agency was instituting two other new programs at the same time as Personal Preference, and an executive there cited competing demands on staff time as the main reason for dropping out.

While agencies considered a variety of factors when deciding whether to participate in Personal Preference, a common theme was a desire to transcend the limitations of the traditional personal care system. Agency executives saw the Cash and Counseling model as a possible solution for some of their clients, especially younger adults with disabilities who wanted to have more control over their own services. In these agencies, the leadership supported the philosophy of consumer direction. Case Example III.1 presents the comments of three agency executives on why their agencies chose to provide consulting under Personal Preference. They represent public, private nonprofit, and private for-profit agencies.

At the time of our visit, 12 agencies were providing consulting services under Personal Preference, just over a third of the 34 that sent staff to consultant training. Recognizing the diversity of its population, the Personal Preference staff had hoped to attract enough consulting agencies to give consumers a choice. However, the program found that consumers generally had little or no information or experience to draw on in selecting a consulting agency and did not seem to value having a choice, at least not in these circumstances. Over time, Personal Preference whittled the list to 12 agencies that were providing effective consulting services.

#### Case Example III.1: Becoming a Consulting Agency

**Public Agency.** Our agency has provided case management since 1993. The case manager plays a pivotal role. We wanted to know if people were capable of taking on this role themselves. Also, we thought that we might be better able to meet the needs of younger adults with disabilities through a cash program. The younger disabled are really independent minded, so a consumer program is really perfect for them.

**Private Nonprofit Agency.** We have had a good, long-standing relationship with the Personal Preference project director. His enthusiasm was a big factor in our decision to participate. We also liked the fact that the program empowered beneficiaries--making them consumers rather than clients. Our agency likes to be involved in programs that make people as independent as possible. The philosophy in our case management program has long been to give people the information they need but not to make their decisions for them. Also, our experience is that people pay more attention to the quality of their services if they are paying for them.

**Private For-Profit Agency.** This agency has been doing case management for 13 or 14 years. It wanted to establish a working relationship with the state. Also, agency leadership felt that the cash program was right for some clients. It is common for the personality of the client and the worker to clash. The cash program offered a way to solve that. Also, agency leadership knew that some clients would be interested in the program. Some younger disabled clients have asked why they couldn't just get the money the state was spending and purchase their own services.

# 2. Recruiting Consultants

New Jersey set only minimal educational criteria for consultants. However, most of those who became consultants were experienced human services professionals.

#### a. Criteria to Become a Consultant

New Jersey required that, to become a Personal Preference consultant, a person merely have a high school diploma and complete a training course. (Section B.3 describes training.) The state did not require that consultants be human services professionals, in part because it was concerned that the norms of many professionals would conflict with the philosophy of consumer direction. Many professionals were accustomed to taking charge themselves and solving problems, instead of empowering consumers to take responsibility for their own care.

Executives of consulting agencies identified several attributes necessary for success as a consultant: (1) listening skills, (2) the ability to help consumers realize their goals, (3) the patience to allow consumers to proceed at their own pace, (4) the ability to handle budgets, and (5) "people" skills in general.

#### Case Example III.2: What Makes a Successful Consultant?

A consultant has to be practical--definitely has to be able to deal with budgetary concepts. Has to have good listening skills. "What is the consumer really saying?" Needs to be able to control the urge to "fix it" for the person. For the pilot project, a consultant has to be flexible, as there will inevitably be changes.

Most agency executives thought that a consultant should have a college degree. One commented that a college degree indicated general proficiency in reading and in mathematics. Another suggested that a college degree in a human services field should be required, as those who pursue such a degree are "self-selected" as people who enjoy working with others.

#### b. Who Are the Consultants?

With few exceptions, consultants for Personal Preference were members of the existing agency staff, rather than newly recruited staff. Two factors led consulting agencies to assign members of their existing staff to provide consulting. First, as indicated, the volume of Personal Preference cases at many agencies was low. Low volume made it possible for existing staff to assume responsibility for Personal Preference along with their other duties. In addition, given low volume, agencies may have wanted to avoid recruitment costs. Second, agencies may have been reluctant to have consulting provided by someone they did not know. One agency executive noted that a consultant holds a position of trust because he or she helps consumers with their financial affairs (at least to the extent of assisting with the cash management plan). The activities of a consultant who abused this trust could damage the consulting agency's reputation.

Despite the state's willingness to accept consultants without college degrees, most had them. Only one agency of the six we visited had a consultant without college training.

Most of the consultants were human services professionals. That is not surprising, as most had been full-time employees of human services organizations, with responsibilities other than consulting for Personal Preference. An example of a consultant who was not a human services professional is a young woman who was trained in accounting and was working for the agency as an assistant to the chief financial officer. She is featured in Case Example III.3, along with a senior consultant at the same agency.

#### 3. Training Consultants

New Jersey set minimal criteria for consultant educational credentials. However, a great deal of effort was devoted to developing consultant training.

## a. Initial Training of Consultants

New Jersey contracted with Rutgers University School of Social Work to develop a curriculum for training consultants. After a pilot test of this curriculum, New Jersey substantially revised both the curriculum and the design of Personal Preference. Before the pilot test, New Jersey had planned to make consultants responsible for outreach and enrollment, as well as for helping consumers after they had enrolled in the demonstration. However, the human services staff participating as trainees in the pilot training did not want to take on the outreach and enrollment functions, because they did not see themselves as social marketers. Because of their reaction, New Jersey decided to separate the outreach and enrollment functions from the other consultant functions and to contract for the former.

Initially, Rutgers University staff ran the training sessions, with state Personal Preference staff giving presentations on selected topics. However, because of the extension of the enrollment period for the demonstration and consultant turnover, more training sessions were required than had been anticipated, so the budget for Rutgers training was exhausted. Personal Preference staff felt that they themselves could conduct training more efficiently than the Rutgers University staff members, who were not involved in the Personal Preference program day to day. Thereafter, Personal Preference staff members conducted all aspects of the training.

Initially, the training lasted five days and was divided into several sessions so that agency staff would not have to be out of their offices for an entire week. The five-day curriculum covered the philosophy of consumer direction and the procedures for the Personal Preference program. It also devoted substantial time to role-playing to reorient trainees who had a case manager background and to provide experience to consultants who did not have a human services background.

Some early trainee consultants felt that the role-playing unnecessarily lengthened the training session. The state agreed, and the session was reduced to three days. Extensive role-playing seems to have been superfluous, for two major reasons. First, most trainees were experienced human services professionals. Second, most did not need extensive orientation to convince them of the value of the consumer direction philosophy. A few trainees reacted negatively to it, but most accepted consumer direction as a good choice for some beneficiaries, even if they sometimes had to struggle with a desire to fix problems themselves. Most human services professionals with a strong negative reaction to consumer direction probably avoided the Personal Preference program altogether.

All the consultants we visited reported that the training had been good or excellent. Some described it as "very effective"; another noted, "I enjoyed it."

Nonetheless, they did have ideas for improvement. Some consultants suggested more training on how to help the consumer develop the cash management plan. One

specific idea was to prepare the consumer in advance of the home visit so the visit could be as productive as possible. They also suggested teaching consultants about the type of official correspondence that consumers should expect. For example, consultants should know to warn consumers that they are likely to receive a letter from the Internal Revenue Service about the amount of payroll taxes due for their employees and that the letter should be forwarded to the fiscal agent for payment (assuming the consumer was using the fiscal agent). In addition, the fiscal agent suggested that a member of its staff participate in training sessions to teach consultants about the information that must be collected to complete the cash management plan and employment documents.<sup>21</sup>

## b. Ongoing Training

Personal Preference offered refresher training because the initial consultant training sessions (in mid-1998) took place more than a year before operations began. The program staff had expected to enroll the first demonstration participants shortly after the initial sessions, but a legal dispute over the selection of the fiscal agent forced the state to postpone enrollment for many months.

Even with the refresher training, the initial group of consultants reported that they had to teach themselves by going through the manual and that they had to learn some things by trial and error. One noted, "Once I made a mistake, I learned from it and never made that mistake again."

Over time, many consultants received on-the-job training from the Personal Preference consultant coordinator. She spent a substantial proportion of her time "coaching" consultants, continuing to do so throughout the demonstration.

At the time of our site visit in the spring of that year, New Jersey was planning meetings of practicing consultants to be held shortly thereafter. This was to be the first formal training session since the refresher training. One goal of this meeting was to provide an opportunity for consultants to learn from each other.

#### 4. Supervision of Consultants

The local supervision available to Personal Preference consultants varied considerably, as it depended largely on the size of the Personal Preference caseload at a given agency. When the caseload was small, an agency often had only one staff member trained in consulting. New Jersey did not require that consultant supervisors complete the Personal Preference training, and agencies did not voluntarily send supervisors to training. The lack of trained supervision did not seem to be a serious problem, as consultants were

<sup>&</sup>lt;sup>21</sup> The state Personal Preference staff indicated that, because the state would have incurred additional charges for their time, they had not asked fiscal agent staff to participate in consultant training sessions.

able to function somewhat independently from day to day. In addition, the senior executive of the agency usually had a basic understanding of the Personal Preference program from regional briefings and state-supplied materials and could draw on that understanding to advise the consultant if questions arose.

The most serious problem caused by having only one trained consultant at an agency involved responding to questions that arose when the consultant was not available. An executive at an agency that had only one trained consultant reported having to scramble to respond expeditiously to consumer requests that came in when the consultant was out of the office.

As Personal Preference caseloads grew and agencies had enough volume for two or more trained consultants (each of whom was part-time), one of the consultants might be named as the local Personal Preference supervisor. However, the typical relationship between consultants at the same agency seemed to be a collaborative one, rather than that of supervisor and subordinate. Case Example III.3 describes one such relationship. Collaboration did not typically extend to visiting consumers as a team, although a joint home visit might be made when something about the case generated suspicion or when a newly trained consultant was making a first visit.

#### Case Example III.3: Local Supervision of Consulting at One Agency

Consultant A is the supervisor. She has the Personal Preference files in her office. She is a case manager, but she runs an emergency services unit--so she is used to dealing with the problem at hand and not trying to solve all the problems in the case. Consultant B is an assistant to the financial officer. She is "a whiz" on the budget side. However, she wants to "make it right." One cannot do that in Personal Preference, as the consumer has to make the decision. These consultants make a very good team, as they bring different strengths to the process. They also have senior agency executives at their disposal--the executive director of the agency for social work matters and the financial officer for business matters.

# 5. Assuring the Quality of Consulting

To limit the burden on organizations providing consulting, as well as on the consultants themselves, New Jersey imposed minimal reporting requirements, which focused on interaction with individual consumers. Before submitting the cash management plan to the state program office, consultants were required to sign it to indicate that they had reviewed it. They also had to submit a report to Personal Preference about each consumer's condition and situation following the quarterly visit to that consumer. In addition, when they submitted an invoice, consultants had to include the names of the consumers assisted and itemize the services provided to each. Personal Preference staff checked that the required reports (cash management plans and quarterly reports) had been received before approving invoices for payment.

These reporting requirements were important in enabling members of the state Personal Preference staff, particularly the consultant coordinator, to become quite knowledgeable about the strengths and weaknesses of consultants. Program staff first got to know consultants in Personal Preference training. Then the coordinator reviewed every cash management plan after consultants had reviewed it. The coordinator also reviewed quarterly reports. In addition, state program staff members and consultants sometimes interacted while troubleshooting problems on behalf of consumers. During the demonstration, Personal Preference staff drew on their knowledge of the consultants to channel new enrollees in Personal Preference to agencies that offered strong consulting services.

Initially, New Jersey imposed only a single formal standard on consultants. They were required to maintain case notes on their contacts with consumers and to include a specified minimum data set in these notes. The documentation was to be made available for review during planned on-site audits of consultants.

Several months into demonstration operations, the state added a standard on the timeliness of initial contact with the consumer. Some consultants were apparently not prompt enough in contacting consumers who had been referred for consulting, so New Jersey stipulated that consultants were to contact a consumer to begin to arrange for a home visit within 48 hours after receiving a referral. Consultants were expected to telephone on weekends as necessary to meet the 48-hour requirement.

When demonstration operations began, the Personal Preference program did have a standard about the timeliness of the development of the initial cash management plan, and this standard had implications for consultants. Initially, the program expected consumers to have completed a cash management plan and be receiving cash within 90 days after their assignment to the treatment group (with 15-day extensions provided with the approval of staff). This requirement imposed an implicit standard on the work of the consultant in helping the consumer develop the cash management plan. For reasons explained in Chapter V, development of the plan often took much longer than 90 days, and the Personal Preference program did not enforce the 90-day standard. Senior program staff feared that doing so might lead consultants to develop the cash plans themselves instead of assisting the consumer in doing it.

# C. Fiscal Agent

As mentioned in Chapter II, New Jersey contracted for a single fiscal agent to serve Personal Preference cash recipients across the state.

#### 1. Procurement Process

The procurement process to secure a fiscal agent was protracted, extending over more than a year. In August 1998, New Jersey issued a formal request for proposal (RFP) to solicit an organization to provide fiscal services for the Cash and Counseling Demonstration under a three-year contract. Although the state expected to have the fiscal agent contract in place by early 1999, this did not happen until November 1999.

New Jersey prepared the RFP for the fiscal agent carefully, even hiring an expert in fiscal services to help prepare the statement of work and holding a bidder's conference. The state sought an organization with a minimum three-year history as a financially viable business and with a philosophy compatible with consumer direction. The state also laid out a number of programmatic criteria in the RFP, including:

- Ability to produce payroll checks (including doing so in certain emergency situations), pay nonlabor-related invoices, and disburse cash grants quickly and accurately.
- Ability to automate key payroll, accounting, and data collection functions to achieve economies of scale.
- Knowledge of federal and state tax, labor, unemployment, disability, and workmen's compensation rules and regulations pertaining to the employment of household/domestic service employees and independent contractors.
- Ability to communicate effectively with consumers (including accessing translation services) regardless of disability, and to respond to all participant requests and inquiries.
- Ability to develop training and skills-examination tools pertaining to payroll for domestic service/household employees.
- Ability to train and assess consumers using these tools.
- Ability to develop a code structure for all types of labor and nonlabor expenditures and to create expenditure reports that contain the needed data elements, are accurate and clearly laid out, and are produced in a timely manner.
- Ability to hire and train professional and nonprofessional staff to respond to a staggered schedule as the caseload builds.
- Ability to broker a life/health insurance option package.

Using these criteria, the state evaluated the bids received in response to the RFP and rejected them all as unsatisfactory. Citing technical errors in the RFP, the state withdrew its initial solicitation and revised and reissued it.

An award was made to one of the bidders on the reissued RFP, but only after a time-consuming legal and administrative process. Two companies submitted satisfactory bids to the revised RFP, and New Jersey awarded the contract to one of them. However, the other appealed the decision, and a final award was delayed for months by the ensuing legal proceedings. Ultimately, in November 1999, the state awarded the contract to the bidder that had appealed.

# 2. Host Organization for Fiscal Services

The host organization for fiscal services under Personal Preference is a community human services organization in a large city in the northern part of New Jersey.

Staff of the organization believes that the philosophy of consumer direction is consistent with the philosophy of its organization--helping people to help themselves--and see consumer direction as the "wave of the future." Although the organization already had a program in which consumers could hire their own workers, Personal Preference offered even more flexibility, which intrigued the staff. An executive indicated that the organization also realized the need to increase the supply of personal attendants and saw the Cash and Counseling model as tapping a different labor supply. This labor supply could help ease a critical shortage of personal care attendants in New Jersey, one that extended beyond personal care to community institutions, such as group homes.<sup>22</sup>

#### a. Variety of Services

The host organization for fiscal services provides a wide variety of other services, many of them for people with disabilities. These services include:

- **Residential Services.** Supportive living, supervised apartment, foster home, group home, home sharing, and supervised transitional living for at-risk youth.
- **Counseling and Job Placement.** Crisis intervention, family counseling, case management, and employment placement.
- **Education and Information.** Recreation and education, enrichment programs for parents of children who are developmentally delayed, conferences and training for

<sup>&</sup>lt;sup>22</sup> This executive reported that 20 group homes that had planned to open in 2000 were unable to do so, because of their inability to hire attendants to staff them.

people with disabilities (including training to help them manage their own personal care attendants), and information and referral.

- Personal Care. Respite services for caregivers of people with disabilities, and a stipend program for people with disabilities to allow them to hire their own workers.
- Other. Retired senior and volunteer program.

### b. Consulting Services Under Personal Preference

At the time of our visit, the host organization for fiscal services was also providing consulting services. However, this had not been the case when demonstration operations began. The organization had responded to the original letter from the director of Personal Preference soliciting human services agencies to provide consulting, and it had been approved. However, once the fiscal agent contract was awarded, the state withdrew its approval for the host organization. As Chapter II indicated, the state program staff was concerned about the absence of checks and balances if one organization provided both consulting and fiscal services.

When the demonstration had been operating for more than a year, the state agreed to allow the host organization that was providing fiscal services to provide consulting as well, as long as it maintained a "firewall" between the two services. At the time of our visit, the organization had two consultants on staff and planned to hire two more. Fiscal staff at that organization confirmed that they did not interact with the Personal Preference consultants. After our visit, the host organization providing fiscal services also became responsible for outreach and enrollment.

#### 3. Staffing for Fiscal Services

At the time of our visit to New Jersey in spring 2001, about 250 consumers were receiving cash, and the fiscal agent had two full-time staff devoted to Personal Preference fiscal services. The host agency's finance director supervised them.<sup>23</sup> One of the full-time staff members for Personal Preference was a consumer liaison, taking consumers' telephone calls and responding to their questions. The other was responsible for (1) reviewing cash plans, employment forms (such as for federal income tax withholding), and other materials as they were received from consumers to make sure that they were completed correctly; and (2) reconciling check requests against cash management plans. Administrative staff members, who worked on a number of programs, were responsible for cutting and mailing Personal Preference checks and for preparing payroll reports for the federal and state governments.

<sup>&</sup>lt;sup>23</sup> At the time of our visit, the finance director had just resigned, and the Personal Preference staff was temporarily reporting to the host organization's executive director.

# 4. Assuring the Quality of Fiscal Services

New Jersey monitored the quality of fiscal services through monthly and annual reports and semiannual audits. The reports showed the total number of packets of employment forms processed and the number of checks cut. A contractor hired by the National Program Office for the Cash and Counseling Demonstration visited all three state demonstration fiscal agents every six months to audit their operations. The New Jersey fiscal agent viewed these contractor visits as very helpful.

New Jersey also required that the Personal Preference program be notified immediately if payroll or disbursements were not made in a timely way.

In addition to this formal reporting, frequent interaction between the staff of the fiscal agent and of the state staff helped ensure the quality of fiscal services.

## **D.** Conclusion

New Jersey was able to attract organizations to provide enrollment, fiscal services, and consulting under Personal Preference. A for-profit organization already under state contract to market a Medicaid managed care program was willing to expand the scope of that contract to include enrollment for Personal Preference. Two nonprofit organizations in the state were keenly interested in providing fiscal services--so much so that they were willing to engage in a protracted legal dispute to determine the successful bidder. Even though payment was reportedly not adequate initially to cover the full cost of providing consulting services, many agencies from across New Jersey came forward to do so.

How can we account for New Jersey's success in attracting organizations to Personal Preference? It is in part due to the respect its director commanded. More fundamentally, interest in Personal Preference stemmed from the human services community's recognition of the flaws of the traditional system of providing personal care--specifically, characteristics that engendered dependence among clients, lack of responsiveness to client need, and inability to secure an adequate labor supply. Organizations were attracted to Personal Preference because it offered a potential solution to these problems. As such, Personal Preference was perceived as the "wave of the future."

# IV. OUTREACH AND ENROLLMENT

New Jersey initially decided to make state Personal Preference staff responsible for outreach and an external contractor responsible for enrollment. The state did this because there was a possibility that the staff of agencies providing traditional personal care might not explain Personal Preference objectively. Later, because the pace of enrollment was lagging far behind that necessary to meet the evaluation sample-size target, the state hired employees to be enrollment specialists, and the Personal Preference staff assumed responsibility for all aspects of outreach and enrollment.<sup>24</sup> In this chapter, we describe how Personal Preference implemented outreach and enrollment, including how it provided information to the community and how it reached out to potential participants and enrolled them.

# **A.** Community Information

The state Personal Preference program staff informed the community about the program. A major goal of its marketing campaign was to provide information about the cash program to human services agencies statewide in the interest of recruiting them to serve as consulting agencies. (We described this aspect of the campaign in Chapter III.)

Another goal was to inform the provider community about the Personal Preference program. To do this, the director and assistant director of Personal Preference gave presentations at meetings and conferences, such as those held by the state nursing board and organizations representing the personal care industry.

At the time of our visit to New Jersey in spring 2001, the Personal Preference program was working with the demonstration's National Program Office to develop a color brochure on Personal Preference. They planned to distribute copies to interested parties (such as beneficiaries and their families and those working in the personal care industry) to emphasize the successes of Personal Preference.

New Jersey did not conduct a mass community information campaign to recruit demonstration participants. Only current recipients of Medicaid Personal Care Assistance (PCA) were eligible for Personal Preference, and the state did not want to generate demand among people who were not eligible. Therefore, the state did not seek publicity for Personal Preference through mass media (such as newspaper articles or television public service announcements). New Jersey had learned from the experience of

Page 47

<sup>&</sup>lt;sup>24</sup> In fall 2001, several months after our visit, one state enrollment specialist resigned, and Personal Preference again moved to have outreach and enrollment services provided under contract.

Arkansas, which drew many inquiries from Medicare and Medicaid beneficiaries who were not eligible for Medicaid PAS and thus not eligible for the cash program. Arkansas had difficulty meeting a surge in demand for enrollment in its cash program immediately following a direct mass mailing. Moreover, as described in Chapter II, Personal Preference was designed to identify potential participants at the time of assessment (or reassessment) for Medicaid personal care. Initially, the state program staff did not want to generate demand among PCA recipients who had not recently been assessed and who would have to wait several months to be enrolled. Later, Personal Preference relaxed the plan to enroll recipients shortly after assessment.

When enrollment began, the contractor was contacting many individuals for whom agencies had submitted consumer data forms many months before. (As Chapter II indicated, agencies began to submit consumer data forms in January 1999, but the legal dispute about procurement of a contractor for fiscal services delayed implementation until November 1999.) When the enrollment contractor tried to contact individuals for whom consumer data forms had been submitted, some could no longer be reached at the telephone number given on the forms, some were no longer community residents, and some were no longer eligible for Medicaid.

When it became apparent that enrollment was not proceeding at the pace required in the enrollment contract (30 signed consent forms a week), the staff of the enrollment contractor made presentations to small groups of people who might be interested and eligible (without regard for their assessment dates or current receipt of PCA). For example, enrollment staff sometimes asked social workers in public housing projects to invite a group who might be receiving Medicaid personal assistance to hear a description of the program. According to the enrollment contractor, these presentations usually were unsuccessful. Most such groups included only a few people who were eligible for Medicaid. Those who were eligible often did not acknowledge their Medicaid eligibility-seemingly because they were reluctant to do so in a public meeting.

In addition, we speculate that Medicaid beneficiaries attending such meetings who were *not* current PCA recipients may have been discouraged because they would have had to enroll in PCA before becoming eligible for the demonstration. In addition, Personal Preference did not offer them help in applying for PCA.

With enrollment lagging, Personal Preference also encouraged providers of traditional PCA services to refer some of their clients to Personal Preference--another departure from the plan to enroll participants only after assessment. The director of the Personal Preference program wrote to the directors of all Medicaid PCA providers urging them to refer PCA clients they found difficult to serve. This letter was intended to decrease the resistance of providers of traditional personal care to the Personal Preference program, as well as to generate referrals. In response, providers did refer many clients. PCA providers felt that clients who did not like the structure or rules involved in working

with an agency might be better suited to self-directed care and often referred such clients to the Personal Preference program. Agencies also referred clients who lived in areas (including rural ones) in which it was difficult for agencies to recruit employees. New Jersey did not ask advocacy groups to help recruit demonstration participants.

### **B.** Outreach and Enrollment

State Personal Preference program and external contractor staff were responsible for initial outreach to individual consumers. State program staff prepared a database of Medicaid PCA recipients eligible for Personal Preference and mailed them flyers about the cash program. The enrollment contractor staff was responsible for the remaining steps in outreach and enrollment. In this section, we describe the roles of the state Personal Preference staff and of the enrollment contractor.

## 1. Determining Eligibility and Initial Flyer

Potential participants in Personal Preference were first identified at the point of assessment (or reassessment) for Medicaid PCA. After completing an assessment, the nurse for the provider of traditional services completed a consumer data form, which was sent to the state Personal Preference program. This form contained the following information:

- Beneficiary Identifying and Contact Information. Name, social security number, Medicaid number, telephone number, and address.
- Relative Identifying and Contact Information. Name and telephone number.
- Care Plan Data. Number of personal care hours authorized for weekdays and for weekends for the upcoming six-month care plan.
- **Other.** Primary language spoken, diagnosis, whether the person was expected to be in the community in six months, and names of the agency, agency nurse, and client physician.

The completed consumer data forms were sent to the Personal Preference state office, where staff members verified eligibility against Medicaid records. They identified anyone who was not a Medicaid PCA recipient or who had lost eligibility for Medicaid since the assessment was completed.

After eligibility was confirmed, the state Personal Preference program staff mailed an introductory letter and flyer to the PCA recipient. These briefly explained the Personal

Preference program and said that someone would telephone the recipient to schedule a home visit to provide more information about it.

Later, because enrollment continued to lag, Personal Preference added a Spanish translation of the description, as well as a toll-free telephone number, to the flyer. The flyer invited people to call if they were interested in learning more about the cash program, and state program staff answered the toll-free number. (Appendix A shows the revised flyer.)

New Jersey also experimented with printing the flyer on colored paper. State staff reported that pink paper seemed to be the most effective in generating interest in the cash program, although the reason for its apparent success was not clear.

#### 2. Databases

Drawing on the consumer data forms, the Personal Preference staff created a database to support the outreach and enrollment process. Information from the consumer data forms was entered and sent to the Medicaid Information Services (MIS) office, where it was combined with information from the state's Medicaid Management Information System to create a master, cumulative database of PCA recipients on whom consumer data forms had been received. This master database was used to check eligibility and served as the basis for the tracking systems that the enrollment and evaluation contractors used.

An extract from the master cumulative database was distributed weekly to the enrollment contractor, and an enrollment database was created from it. Each week, the enrollment contractor identified the names of PCA recipients who had been newly added to the database. Enrollment contractor staff added the new names to a separate database that was used to track the status of all contact attempts and contacts. Cases were tracked on this database until the number of contact attempts required in the contract with the state had been completed or until an individual had decided whether to participate in the demonstration (and had signed a consent form if he or she had decided to participate). The master database was also updated with information on the outcome of the enrollment process.<sup>25</sup> A code for the outcome was entered on the current extract of the master database, and a copy was returned to the MIS office, where the master database was updated.

In addition, the evaluation contractor received an extract from the master database weekly. This extract identified the names of people who had just signed consent forms to participate in the demonstration. The evaluation contractor then telephoned these people

<sup>&</sup>lt;sup>25</sup> Participants in the demonstration signed consent forms, usually during the home visit from the enrollment contractor staff. However, the evaluation adopted the date of random assignment as the demonstration enrollment date. Those who signed a consent form but withdrew before random assignment were not considered enrolled.

to conduct baseline interviews. After a baseline interview was completed, the evaluation contractor randomly assigned the individual to the treatment or control group. Random-assignment status was communicated daily to the Personal Preference program, which notified the individual of the assignment in writing. The evaluation contractor staff also added random-assignment status to its extract of the current master database and returned a copy of it weekly to the New Jersey MIS office, where the master database was updated.

The Personal Preference office also received a copy of the weekly extract of the database, which it used to troubleshoot problem cases. For example, if a consumer called about his or her random assignment, Personal Preference staff could determine whether the case had been randomly assigned and, if so, whether the assignment was to the treatment or control group.

In addition, at the request of the Personal Preference program, the MIS office compared the master Personal Preference database to Medicaid claims to assess whether consumer data forms were being received for all PCA recipients. The program staff asked for the comparison because it was concerned that providers might be "prescreening" clients they judged inappropriate for the Personal Preference program. They could do this simply by not submitting a consumer data form. However, the comparison of the master database and claims showed that consumer data forms were routinely being completed for all PCA recipients.

To prevent delays in transmission, the database file was typically distributed each week as an attachment to an e-mail message. If e-mail systems were down, a disk containing the file was hand delivered (the offices of the Personal Preference program, enrollment contractor, and evaluation contractor were only a few minutes driving time from the MIS office).

#### 3. Procedures of the Enrollment Contractor

Enrollment contractor staff members contacted PCA recipients by telephone and in person. Although the number of staff members varied over the course of the contract, the enrollment contractor reported typically having three telephone staff members and seven or eight field staff members dedicated to the project.<sup>26</sup> According to the director of Personal Preference, enrollment staff turnover was high.

<sup>&</sup>lt;sup>26</sup> The Personal Preference program staff reported that the enrollment contractor had up to 12 individuals staffing enrollment for Personal Preference but described them as part-time.

### a. Telephone Outreach and Enrollment

As the enrollment contractor received names of recipients, they were separated by primary language spoken (as indicated on the consumer data form). Those whose primary languages were English, Spanish, or Russian were assigned to telephone staff members who spoke their language. Others were assigned to English-speaking telephone staff members. If necessary, the enrollment staff identified English-speaking family members or friends to translate for the PCA recipients.

Members of the telephone enrollment staff called each new referral, using scripts that the enrollment contractor had developed and the state Personal Preference staff had approved. The telephone staff asked if the PCA recipient had received the materials mailed from the state Personal Preference program, briefly explained the program, and tried to schedule a home visit so a member of the field enrollment staff could give a presentation about the program.

The visits were typically scheduled between 10:30 A.M. and 6:00 P.M., at a time when family members could attend but the personal care aide would not be present. The time for the visit was tailored to fit a schedule that many recipients followed. As an enrollment staff member noted, "Seniors get up late and eat an early dinner." However, visits would be scheduled later in the evening if that were more convenient for family members.

Enrollment staff scheduled visits when family members could be present because consumers often sought advice from their family about whether they should participate. In addition, family members might become representatives or employees under Personal Preference. Enrollment staff also felt that family members usually would support participation. One noted, "For seniors, you need a son or daughter there. If the person is close to the aide, they won't change unless the son or daughter is there to argue the benefits."

Enrollment staff tried to schedule visits when the personal care aide would not be present. As discussed in the next section, aides sometimes joined the discussion and discouraged participation in the cash program. One member of the enrollment staff noted, "I'll ask 'from what time to what time will your aide be there?' and then I try to avoid that time." Some consumers have close relationships with their aides and may feel that it is appropriate for the aide to join in the discussion.

#### b. Home Visits

After the home visit was scheduled, the case was transferred to an enrollment field staff member, who made the visit. Usually, the staff member telephoned the PCA recipient two days before the appointment to confirm the date and time.

PCA recipients often broke their appointments. Despite having confirmed their appointments, field staff members reported often "going to the home, only to find no one there." Some other recipients would make appointments, then call to say that they had changed their minds and were no longer interested.

The enrollment staff believed that many recipients who did not keep their appointments had been dissuaded from participating by their personal care aides. One reported that aides had participated in discussions about demonstration participation "on at least a dozen occasions." Another reported that aides discouraged participation in her presence. These aides had spoken in Spanish, unaware that she understood that language.

Like the telephone enrollment presentation, the home visit presentation was based on a script that the enrollment contractor had developed and the state Personal Preference staff had approved. Often, the PCA recipient and family members viewed a videotape about Cash and Counseling in New Jersey during the home visit. One member of the enrollment field staff estimated showing the videotape in 30 to 40 percent of cases, sometimes leaving a copy for others to view later.

The discussion following the home visit presentation focused on issues that those in attendance raised. As one enrollment staff member said, "They want to know, 'How will this program benefit me?" During the discussion, the enrollment staff indicated what the dollar amount of the cash benefit would be if a recipient were randomly assigned to the treatment group. (The amount was based on the number of hours of care listed on the consumer data form.)

If the PCA recipient and his or her family decided to participate in the demonstration, the field staff visitor read and explained the consent form, section by section, before requesting a signature. If the recipient was unsure about participating or wanted time to discuss the decision with someone else, the enrollment staff tried to schedule another home visit or called back later to arrange one.

### 4. Changes in Enrollment Contractor Procedures over Time

The pace of enrollment consistently failed to meet the contractual obligation of the enrollment contractor to produce 30 signed consent forms a week (the pace that was necessary to meet the evaluation sample size targets). In response, the state Personal Preference staff and the enrollment contractor began reviewing enrollment procedures and developing new approaches. They held many meetings for several months to do this.

They identified two major issues. First, the enrollment contractor often had difficulty reaching PCA recipients by telephone due to incorrect telephone numbers and disconnected telephone lines. Second, enrollment staff members were making more than

the required three attempts to schedule a home visit, sometimes trying five or six times. The director of the Personal Preference program judged that the enrollment contractor was pursuing reluctant people for too long and should give up and move on. In addition, the enrollment contractor was sometimes making multiple home visits to try to get signed consent forms. These additional contacts increased the costs incurred by the enrollment contractor with little payoff.

In January 2000, Personal Preference program staff and the enrollment contractor agreed upon revised procedures. The revisions included relaxing the standard for the first telephone contact from 5 to 10 business days following identification of a PCA recipient, as well as instituting a procedure for writing to those who were not readily contacted by telephone. If the recipient did not have a working telephone number or the enrollment contractor was unable to reach the recipient after the third telephone attempt, the contractor mailed a letter explaining the demonstration and asking the recipient or a family member to call the enrollment contractor within 10 days if they were interested in learning more about Personal Preference. If the recipient or family did not call, these cases were not pursued further. In addition, to reduce travel costs, recipients were grouped by zip code before assignment to field enrollment staff for home visits.

Perhaps the most fundamental revision of the enrollment procedures involved shifting, in selected cases, from in-person to telephone presentations of the Personal Preference program. The telephone enrollment staff began explaining the cash program. Consumers (usually those already familiar with Personal Preference) were occasionally enrolled over the telephone, without a home visit. If the consumer or family member understood the features of the cash program, declined a home visit, and found telephone enrollment acceptable, the enrollment contractor would mail a consent form, as well as written material explaining Personal Preference, and ask the recipient to sign the consent form and return it by mail. If the recipient did not return the consent form, the enrollment contractor would follow up by telephone. If the recipient and his or her family wanted a home visit or were undecided following the telephone presentation, a home visit was scheduled.

These revisions were somewhat successful. The time required for a case to reach a final enrollment outcome was decreased. The pace of enrollment also increased; however, this improvement proved short-lived.

When the pace of enrollment slowed once more, the Personal Preference program and the enrollment contractor again revised their procedures. Instead of the program sending out a letter and flyer to be followed up by the enrollment contractor, information from the consumer data form was made available to the contractor without delay, and telephone enrollment staff began making cold calls to the PCA recipients who had recently been assessed. This revision to procedure also generated a spurt in enrollments. However, this improvement also was short-lived.

Finally, the enrollment contractor established a team responsible for the Personal Preference program. The team member who made the initial telephone enrollment call also made the home visit (unless the consumer declined the visit). The assumption underlying this procedure was that recipients would be more comfortable during the home visit (and thus more likely to participate) if they were discussing the cash program with someone with whom they had already talked.

The enrollment contractor reported that all these methods of outreach and enrollment worked for some people and recommended that any enrollment effort use several approaches. The enrollment contractor staff also suggested that New Jersey should market the Personal Preference program more broadly, arguing that word-of-mouth can be powerful: "Everybody should hear about it; then consumers would be hearing about it from their friends and neighbors."

The costs the enrollment contractor incurred were higher than it had projected. New Jersey initially responded by adding funds to the enrollment contract. However, the director of the Personal Preference program, frustrated by the slow pace of enrollment month after month and the high cost per enrollee, decided to hire state employees as enrollment specialists and to have Personal Preference staff members supervise them. The Personal Preference staff hoped the pace of enrollment would improve if it could supervise the enrollment process more closely than had been possible with a contractor. We discuss the work of the state enrollment specialists in the next section.

### 5. Hiring and Training State Employees as Specialists

Midway through the demonstration, the director of Personal Preference began the process of hiring three full-time state employees (in temporary positions) as enrollment specialists. Getting permission to create state jobs, even temporary ones, took several months. Permission was finally granted when it became apparent that one or two similar positions were open in other parts of state government. The potential workers were identified months before the positions were approved; fortunately, the three were still available when approval came. The state enrollment specialists began work in February 2001, shortly before our visit.

All three of the state enrollment specialists were bilingual in Spanish and English, and their backgrounds were in social work and marketing/communications. One lived in northern New Jersey, the others in central New Jersey. The three specialists shared two state automobiles, which they used to make home visits.

The state enrollment specialists were trained by staff from the Personal Preference program, the enrollment contractor, and a social marketing firm working under contract to the Cash and Counseling National Program Office. Senior Personal Preference staff members explained the cash program and their responsibilities to the state specialists.

The specialists then met with enrollment contractor staff members, who gave them advice on marketing and on health and safety issues involved in making home visits. Finally, staff members of the social marketing firm trained the state specialists in communication skills. The social marketing training was extremely well received; the state enrollment specialists left the training enthusiastic about the cash program.

When we visited, the state specialists had been in their positions for only about two months. Senior Personal Preference staff members were pleased with their progress and reported that they were "becoming a good team." The specialists were also engaged in a healthy competition among themselves to enroll large numbers of PCA recipients in Personal Preference.<sup>27</sup>

### 6. Enrollment Procedures Under State Specialists

As responsibility for all outreach and enrollment activities moved from the enrollment contractor to the Personal Preference program, some procedures were revised. We describe these revisions next.

#### a. Database

The database in the state Personal Preference office was revised to provide more detail for tracking the status of enrollment contacts. Obviously, the enrollment contractor no longer needed to receive information from the master database. Having one less recipient for this information eliminated an occasional source of delays in the weekly transmissions of copies of the database.

At the time of our visit in spring 2001, the Personal Preference staff anticipated that it would soon assume all responsibility from MIS staff for the maintenance of the database. In New Jersey, as programs move from demonstration to permanent status, programs typically become responsible for their own databases. In this case, however, the shift in responsibility came before the program achieved permanent status because the MIS staff member who had been maintaining the Personal Preference database was promoted and no longer able to maintain the database.

#### b. Outreach and Enrollment Procedures

At the time of our visit, the state enrollment specialists were focusing on New Jersey counties where the most PCA recipients had recently been added to the database. They

<sup>&</sup>lt;sup>27</sup> A few months after our visit, one of the state enrollment specialists resigned. Mindful of the delays in hiring state employees, the state Personal Preference staff decided to amend the contract of the fiscal agent to provide outreach and enrollment services in northern New Jersey. The remaining two state employees were to continue to provide outreach and enrollment in central and southern New Jersey.

did this to increase enrollment quickly (to meet evaluation sample-size targets) and minimize travel time. Because of this strategy, however, referrals to consulting agencies in counties with fewer recipients were lagging. One consulting agency we visited was particularly eager for new referrals.

The state enrollment specialists followed the procedure adopted by the enrollment contractor of having the same staff member responsible for both the enrollment telephone call and the home visit. The specialists telephoned potential participants from the Personal Preference office one or two days a week. Over the telephone, they explained the cash program and tried to schedule home visits. They also entered the results of their contacts into the database. The state enrollment specialists spent the rest of the week making home visits. They worked from home, sending an e-mail to their supervisor indicating where they would be on a given day and calling in twice a day. If a home visit fell through, the specialists called to get a referral of another PCA recipient they might be able to visit in the area. (To make this process easier, the names of recipients newly added to the database were sorted by county of residence.)

The state enrollment specialists developed some enrollment techniques of their own. One specialist stopped using the telephone script that the enrollment contractor had developed. Instead, he tried to establish rapport with the PCA recipients so they would feel more comfortable during his visit to their home. This approach worked well--recipients enjoyed visiting with him and sometimes called the office and asked him to come back for another visit.

The state enrollment specialists also relied less on written materials than did the enrollment contractor, which was useful since some PCA recipients did not understand written material and felt overwhelmed by it. One specialist gave recipients a basic understanding of the cash program when he telephoned. He believes the telephone is particularly important because many people do not, or cannot, read letters. "They'll say they are waiting for their grandson to read it," he said. Another way to avoid written material is to send the video before the home visit. However, this often does not work for elderly PCA recipients, as few of them have a video player. Yet another technique is the use of a PowerPoint presentation instead of written material. Case Example IV.1 describes the PowerPoint presentation that one of the state enrollment specialists developed.

At the time of our visit, another change to outreach and enrollment procedures was about to be implemented. After unsuccessful attempts to reach a PCA recipient by telephone, specialists were to begin mailing another copy of the flyer describing the Personal Preference program and enclosing the specialist's business card, which listed the program's toll-free telephone number. The state Personal Preference staff had suggested this approach to the enrollment contractor, but the contractor did not implement it.

#### Case Example IV.1: PowerPoint Presentation

One specialist developed a PowerPoint presentation that is very simple; it just lists "talking points" about Personal Preference. He said, "I like to keep it simple. Families are rushing; there is not enough time." He takes his laptop with a big screen to the home and uses the presentation to go through the features of Personal Preference, point by point. "It walks me through the details. That way, I don't go off track." He answers questions at the end. Some of the typical questions are answered in the PowerPoint presentation. The presentation is in English, but the specialist translates it into Spanish, as necessary. He also uses the video but reports that people seem to like to see the Power Point presentation first.

# C. Attractive and Unattractive Features of Personal Preference

The enrollment staff--both those employed by the enrollment contractor and the state-were in a good position to identify the features of Personal Preference that potential demonstration participants found attractive or unattractive. (The evaluation will report on participant satisfaction with various aspects of the cash program based on interviews with consumers, but we do not draw on those data here.)

#### 1. Attractive Features

Both the enrollment contractor and state enrollment specialists listed several features of Personal Preference as attractive to consumers. First, enrollment staff reported that consumers liked being able to hire someone they knew, like a relative, friend, or neighbor. This was especially true if the consumer had identified someone they wished to hire. Although most consumers did not want to hire a stranger, enrollment staff reported that some did not want to hire a family member. Such consumers sometimes expressed satisfaction that they could help their communities by providing employment and liked the idea that they were helping their community at the same time that people in the community were helping them.

Enrollment staff reported that having control over the funds was a second attractive feature of Personal Preference. Some consumers were dissatisfied with the amount of traditional personal care they received and thought they could get more hours of care than they had been receiving if they controlled the funds. Some thought that they could get more care by stretching the funds farther than traditional providers did. Others were surprised to learn the number of hours of personal care included in their care plans--they had been receiving far fewer hours. A few had not seen their aides in months.

A third attractive major feature was the flexibility of the cash benefit. Consumers liked being able to schedule care when it was needed. They also liked being able to purchase goods and services. These goods and services included equipment (such as a small appliance or medical equipment that Medicaid did not cover) and home modifications (such as a ramp) that consumers needed to become more self-sufficient. They particularly valued the ability to purchase transportation services.

Finally, enrollment staff reported that the availability of bookkeeping services from the fiscal agent was another attractive feature.

#### 2. Unattractive Features

The enrollment contractor and state enrollment specialists identified three unattractive features. Two of these are artifacts of the demonstration and evaluation and would not be present in a regular program. First, enrollment staff agreed that consumers found the 50/50 chance of being randomly assigned to the control group unattractive. They reported that almost all consumers reacted negatively to this feature of the demonstration and evaluation. Some consumers questioned the rationale for random assignment, asking, "How come if the state is willing to give the money, we have to do the 50/50?"

Deferring description of random assignment may have contributed to its unattractiveness. The written materials about Personal Preference clearly indicate that half the participants will be assigned to continue to receive traditional services. However, enrollment specialists reported sometimes deferring oral descriptions of random assignment until the later part of their presentations--perhaps after the consumer had already made an informal decision about participation. One state enrollment specialist reported waiting until toward the end of a contact with a consumer to mention the 50/50. This specialist also reported that disillusionment with random assignment gave rise to a negative image of the cash program, noting, "So word of mouth spreads that you shouldn't talk to these people."

The length of the demonstration was a second unattractive feature. Some consumers were apparently dissatisfied that they were assured of receiving cash for only two years.

The third unattractive feature (which would be part of a regular program) was the effort required to manage the cash benefit, especially the paperwork. One enrollment staff member put it this way: "Cash is easy, but the paperwork of the cash management plan is off-putting." Some consumers apparently were overwhelmed by the prospect of managing the cash benefit, in spite of being able to get help from the fiscal agent.

The enrollment specialists agreed that those who refused to participate in the demonstration generally did so because they were satisfied with their services under the traditional system. Some consumers had been with one agency and aide for a long time.

The aide was their friend or like part of the family, and they did not want to lose that relationship.

### D. Lessons about Outreach and Enrollment

In this section, we present lessons gleaned from the experience of the Personal Preference program staff in outreach and enrollment.

### 1. Should Personal Preference Adopt a Formal Screening Process?

Like the other Cash and Counseling states, Personal Preference did not adopt a formal screening process to exclude those inappropriate for a cash program (except that those not expected to remain in the community for six months were excluded).

One lesson of Personal Preference is that a policy of "taking all comers" can work. Moreover, program staff indicated that it would be very difficult (if not impossible) to identify in advance consumers who are unable to manage the cash benefit, either by themselves or with help from a representative and the fiscal agent. The staff of the enrollment contractor enthusiastically endorsed the idea that everyone should be given all the facts about the cash program and then allowed to decide whether they wanted to participate. State Personal Preference staff members were more ambivalent about this issue. While they believed everyone should have a chance to try to manage a cash benefit, they also knew from working with the consultants that some consumers had great difficulty doing so. They cited two examples: a person with a psychiatric diagnosis who was not following her medication regimen and a person who lived in deplorable circumstances and was receiving Adult Protective Services. Nonetheless, state Personal Preference staff members were at a loss to define a basis by which applicants could be screened out.

### 2. Providing for Language Diversity

Diversity--ethnic, cultural, and language--characterizes New Jersey's population of PCA recipients. While the state Personal Preference staff was aware of New Jersey's diversity from the beginning of the program, the extent of the diversity was unexpected. For example, so many PCA recipients were assigned to the "other" language category on the consumer database that the state added additional categories for specific languages.

Providing for language diversity is particularly important because non-English-speaking consumers should be good candidates for a cash program, since aides who spoke their language and were familiar with their cultures were often not available from traditional providers. With the cash benefit, a person who did not speak English could hire someone who spoke their language and shared their ethnic or cultural background. As a result, a consumer might be more satisfied with their care.

Mounting a successful outreach and enrollment effort in a diverse community is a challenge, and Personal Preference used several techniques to try to meet it. First, New Jersey insisted that the enrollment specialists (both the contractor and state employees) include those who spoke the languages most common in New Jersey (English and Spanish). Many program materials were translated into these languages and into Russian, the third most common language in the state.

Since it was not possible to translate all the materials into a number of languages, Personal Preference relied on translators, who often were family members of the PCA recipient. If no family members were available, someone from the community might be called upon to help translate during a telephone call or home visit. Because Personal Preference sometimes had to send people written material in a language they could not read, the program included a notice in those mailings that informed the addressee in 14 languages (including English) that the material in the mailing was important and requested him or her to ask someone to translate it immediately. (Appendix A shows this notice.)

In addition, Personal Preference used the translation services on the AT&T language line. To use this service, a program staff member would put a telephone call on hold and dial an access number, after which a translator fluent in the requested language would join the call. While expensive, this service was frequently needed. On average, the Personal Preference program staff used the language line for 10 calls a day.

New Jersey enrollment specialists emphasized that information should also be available in several media. Accommodations were needed for people with vision and hearing impairments, as well as for the many people who are not competent readers in any language.

### 3. Importance of Family Participation

Enrollment specialists emphasized the importance of including family members in home visits and of being flexible in scheduling these visits so family members could attend. PCA recipients seemed to feel more comfortable having someone they trusted in their homes when the enrollment staff member visited. They also seemed to want a family member involved in making the decision about participation. Often, a family member would be serving as a worker or representative if the recipient chose to participate.

Flexibility in scheduling was necessary to meet with relatives, who often worked. One enrollment specialist noted, "If you only work nine to five, they think you don't care. I am successful in asking people to meet early in the morning before work or during the early part of the workday."

### 4. Difficulty in Meeting Enrollment Goals

In an ongoing Medicaid program, New Jersey might allow caseload to build gradually (at least once caseload was large enough to support the efficient provision of consulting and fiscal services.) However, to secure a relatively large sample for the evaluation, the Personal Preference program was asked to enroll a large number of people in a relatively short time. Consistent with this requirement, New Jersey set an initial goal of 30 new enrollees a week, or more than 120 a month. While the pace of enrollment varied over time, it consistently fell far short of the goal. In response, the target for the evaluation was reduced and the enrollment period extended from a year to 30 months. With these changes, enrollment of 70 a month would have been enough to meet the evaluation sample-size target. However, Personal Preference often did not enroll 70 participants in a month.

Can we account for the difficulty that Personal Preference experienced in meeting its enrollment goals? Next, we assess whether the basic organization of outreach and enrollment, staffing levels, marketing techniques, and consumer interest might account for New Jersey's difficulty in meeting the enrollment goals for Personal Preference.

### a. Organizational Structure of Enrollment

New Jersey's two approaches to organizing outreach and enrollment--external contracting and internal state employees--provide an opportunity to assess the relative advantages and disadvantages of these approaches, as well as their relative success in meeting enrollment goals.

Contracting for enrollment in Personal Preference proved very expensive per case enrolled. To review, New Jersey was able to obtain the services of an external contractor quickly by amending a current state contract, and the enrollment contractor met the selection criteria the state had developed. However, the enrollment contractor was unable to meet the weekly target of 30 enrollees called for in its contract. Moreover, it experienced costs beyond those budgeted. (Perhaps its initial budgets were unrealistic due in part to its lack of experience with home visits.) New Jersey increased the funding for the enrollment contractor, but the pace of enrollment was still below the contractual target. As a result, cost per case enrolled was high.<sup>28</sup> State Personal Preference staff attributed the high cost per case to poor contact information and the continued pursuit of reluctant PCA recipients. They concluded that contracting for outreach and enrollment had not been successful.

Page 62

-

<sup>&</sup>lt;sup>28</sup> New Jersey's policy of visiting *all* PCA recipients to explain Personal Preference also increased enrollment costs relative to those of the Arkansas program, where only those who expressed an interest in Cash and Counseling were visited before enrollment.

Hiring state employees as enrollment specialists also proved problematic. To review, New Jersey initially avoided using state employees due to concerns about delays in hiring. This concern was well founded--state enrollment specialists were not on the job until many months after the Personal Preference program first sought approval to hire them. Moreover, one resigned to take another job after only nine months. On the other hand, after state employees were on the job, delays caused by having the enrollment contractor involved were reduced. Moreover, although Personal Preference program staff found it difficult to supervise staff members who did not come into the office, they had closer control of the enrollment procedures with state employees. Closer control increased the state's ability to experiment with different approaches and quickly discard those that did not prove successful. Such flexibility is especially valuable in a new program that is attempting to "shake out the bugs." However, state employees also were unable to meet the goal for the pace of enrollment. Six months after they began work, the monthly enrollment had fallen to less than 50 consumers, compared to 70 a month needed to reach the revised evaluation target.

We conclude that the basic organizational structure of outreach and enrollment in New Jersey does not seem to explain the failure to meet the targeted enrollment pace. Neither contractor staff nor state employees were able to do so.

### b. Was State Enrollment Staffing Adequate?

Personal Preference sought to use the state enrollment specialists' time efficiently. The director gave state enrollment specialists equipment that would let them keep in touch and allow them to use their time productively. They had laptops and beepers, and they could call the program's toll-free line from a consumer's home. (At the time of our visit, the state enrollment specialists did not have mobile telephones, but the director felt that they needed them.) The specialists also had state vehicles.

To minimize travel time, the state enrollment specialists initially focused on areas of the state with the largest concentrations of newly assessed PCA recipients. The numbers of cases enrolled in a month likely would have increased had outreach covered the entire state. It seems likely, however, that three full-time staff members were not enough to enroll recipients across the state. In contrast, the enrollment contractor had employed three telephone staff members and seven to eight field staff members to cover the state. Even if the contractor field staff members worked on Personal Preference part-time, they lived in different parts of the state, so their travel time might actually have been less than that of the state-employed enrollment specialists.<sup>29</sup>

<sup>&</sup>lt;sup>29</sup> At the time of our visit in spring 2001, state enrollment specialists had been on the job for only two months. It is possible that with time they would have developed ways to cover enrollment in the entire state. However, by fall 2001, one of the enrollment specialists had resigned, and Personal Preference was again considering contracting for enrollment services.

### c. Outreach Techniques

A change in outreach techniques might have increased the pace of enrollment in the Personal Preference program.

Letters and Testimonials. One outreach technique used successfully in Arkansas (and later in Florida) was a letter from the state governor. At the time of our visit in spring 2001, the Personal Preference program was aware of the Arkansas experience but had not used a letter from the governor. The Personal Preference staff was considering a letter from either the governor or a member of his cabinet, the commissioner of Human Services. However, the director of Personal Preference had several concerns. He thought that getting the state bureaucracy to approve a letter could be time-consuming and result in a delay. He also felt that, given the diversity of languages spoken in New Jersey, the purpose of the letter might not be readily understood and might even create anxiety among some Medicaid recipients. Finally, the director felt that a letter from New Jersey's governor would not be as successful as one from Arkansas's governor. The Arkansas governor was well known and popular with his constituents, while the New Jersey governor had assumed that position only a few months before our visit (when the former governor accepted a cabinet position in the Federal Government).

At the time of our visit, Personal Preference had also made relatively little use of testimonials from satisfied cash recipients as an outreach technique. Arkansas had made extensive use of testimonials from consumers who had enrolled in its cash program--for example, by enclosing copies in direct mailings to PCA recipients. Greater use of testimonials would be consistent with the suggestion of the enrollment contractor that New Jersey rely more on word of mouth to generate interest in Personal Preference. When we visited New Jersey in spring 2001, the Personal Preference program had made a list of participating consumers willing to speak to potential participants but was not mailing testimonials to PCA recipients. In addition, at that time, a social marketing firm, with funding from the National Program Office, was preparing a multicolor brochure on the Personal Preference program. This brochure was to feature consumers participating in Personal Preference.

#### d. Consumer Interest

Finally, it is possible that Personal Preference recruited most of the PCA recipients in New Jersey who were interested in participating in the cash program during the sample intake period. From the inception of the demonstration, New Jersey's ability to reach the sample-size target for the evaluation had been a concern, given that its Medicaid PCA program served only about 12,000 beneficiaries annually--the smallest number of any Cash and Counseling state. This concern was somewhat assuaged by the results of a telephone survey of a randomly selected sample of Medicaid PCA recipients, conducted while Personal Preference was being planned. These results indicated that more than 40

percent of New Jersey PCA recipients were interested in a cash program (Simon-Rusinowitz et al. 1998). Nonetheless, despite concerted efforts by the Personal Preference program, enrollment fell far short of the level that the results of the telephone survey predicted. Personal Preference enrolled about 1,750 participants in the demonstration by the time enrollment for the demonstration closed at the end of June 2002.30 These 1,750 enrollees represent about 14 percent of the number of PCA recipients in the year prior to the demonstration--a percentage that is roughly the same as the comparable percentage in Arkansas (between 10 and 15 percent). Enrollment percentages in New Jersey and Arkansas are also roughly comparable if one separates enrollment among elderly adults and among nonelderly adults with physical disabilities. During the year before the demonstration, a larger percentage of Medicaid personal care recipients were under age 65 in New Jersey than in Arkansas (about 30 percent, versus about 20 percent). In each state, however, the number of elderly enrollees is roughly 8 to 10 percent of the number of elderly PCA recipients in the year before the demonstration. Similarly, in each state, the number of nonelderly enrollees is roughly 15 to 20 percent of the number of nonelderly PCA recipients in the year before the demonstration.<sup>31</sup>

#### e. In Conclusion

Would Personal Preference have been able to recruit more demonstration participants if it had adopted different outreach and enrollment procedures? It is not possible to answer this question definitively, but it seems unlikely that different procedures would have produced a materially larger number of enrollees. A letter from the governor or testimonials might have been useful, but perhaps only marginally so. Coverage of the entire state by state-employed outreach specialists probably would have increased the number of enrollees in Personal Preference. This strategy, too, might have been only marginally helpful since the state-employed specialists focused on the most promising areas of the state--those with concentrations of newly assessed PCA recipients.

The fact that Arkansas was able to recruit roughly the same percentage of annual recipients of Medicaid personal assistance as New Jersey leads us to the tentative conclusion that it was unrealistic to expect a much larger number of participants in New Jersey and thus that outreach and enrollment under Personal Preference was generally successful. Arkansas used a letter from the governor and testimonials to good effect, and Arkansas's outreach and enrollment staff were able to cover the entire state for the entire sample intake period (see Phillips and Schneider 2002). Even so, that state enrolled no

\_

<sup>&</sup>lt;sup>30</sup> The exact number in the evaluation research sample is 1,714. Approximately another 50 people participated in Personal Preference but not in the evaluation research sample. They were members of a household with another person in the research sample.

<sup>&</sup>lt;sup>31</sup> This cross-state comparison of enrollment rates is preliminary. Error would arise because enrollment occurred in different periods in the two states and because the populations of personal care recipients may have been changing at different rates in the two states. Planned analyses of claims data will support a more accurate comparison.

larger a percentage of the eligible population (as of the year before the demonstration), albeit during a slightly shorter intake period (29 months for Arkansas, versus 32 for New Jersey). It is possible that no more than about one in five of the eligible population finds a cash program attractive enough to enroll--at least in the context of a randomized experiment of a new social program. Thus, the simple fact that Arkansas had a larger population of personal assistance recipients may explain why that state was able to meet the sample-size targets for the evaluation while New Jersey was not.

On the other hand, the New Jersey and Arkansas situations differ in other ways that presumably should have made the New Jersey cash program more attractive than that of Arkansas. In particular, the amount of the average cash benefit was much higher in New Jersey than in Arkansas. Thus, our conclusion that the number of New Jersey enrollees could probably not have been increased materially must remain tentative.

Although the pace of enrollment in Personal Preference did not match original expectations, the cash program attracted large numbers of beneficiaries relative to the size of a typical pilot program. As a result of the demonstration, New Jersey confirmed that the cash program was attractive to a sizable minority of PCA recipients. Moreover, the demonstration proved that a cash program is attractive to elderly consumers in that state. As of the time of our visit, just over half (53 percent) of those enrolled in the demonstration were elderly beneficiaries.

### V. CASH PLANNING AND MANAGEMENT

After a demonstration participant was randomly assigned to the treatment group to receive cash, the state Personal Preference staff notified the consumer and referred the case to a consulting agency. This chapter describes the referral process. It then discusses the process for developing and revising the plan to manage the cash allowance.

### A. Referral to Consulting Agency

As Chapter II indicated, New Jersey chose to have multiple consulting agencies. A member of the state Personal Preference staff referred consumers to one of these agencies.

### 1. Choice of Consulting Agency

Initially, a member of the state Personal Preference staff contacted consumers assigned to the treatment group and offered them a choice of consulting agencies in their area. If the chosen consulting agency declined a referral, the program staff contacted the consumer and asked that he or she choose another agency. These procedures delayed the start of consulting and lengthened the time it took for the consumer to receive the initial cash allowance. (Consumer choice was not an issue in referral for fiscal services, as a single organization served the entire state.)

After a few months, Personal Preference began to assign consumers to a consulting agency based on geographic area and on the agency's capacity. This reduced the delay in getting consulting under way. Few consumers had experience with these agencies or knowledge of their performance on which to base a choice. Consequently, few found it important to have a choice. Finally, not all consumers had a choice. New Jersey recruited 34 consulting agencies, but in some areas there was only one.

The state Personal Preference staff contacted the consulting agency to complete the referral. In addition to the consumer's name and contact information, the referral included demographic information on the consumer and care plan hours, both from the consumer data form.

### 2. Selecting a Consultant

After the state Personal Preference office gave the referral information to a consulting agency, the agency assigned a consultant to the case. The method of selecting a consultant varied, depending in part on the volume of Personal Preference cases. In some

agencies, there was no decision to be made, because only one staff member was trained in Personal Preference consulting. Other agencies reported rotating assignments of new referrals. In one such agency, consulting fees were passed on to the consultants, and the agency wanted to make sure that each consultant had equal access to this source of extra income.<sup>32</sup> Finally, an agency with a large caseload reported dividing its catchment area into zones and assigning a consultant based on the zone in which the consumer lived. This approach reduced travel time when consultants were making home visits. An exception to the zone approach was made for Spanish-speaking consumers, who were assigned to the only Spanish-speaking consultant.

### **B.** Developing the Cash Plan

After receiving a referral, the consultant telephoned the consumer to schedule a home visit to train the consumer (and a representative, should the consumer name one) on how to develop the cash management plan.

### 1. Arrangements for the Initial Home Consulting Visit

Consultants reported different approaches to preparing the consumer for the initial home consulting visit. They also reported revising their procedures as they learned how to use the home visit time as efficiently as possible.

At the time of our visit in spring 2001, consultants differed in their approach to mailing materials to consumers before the initial home visit. Some consultants believed that people often mislaid pre-mailed materials, so they brought materials with them when they came for the visit. Other consultants reported mailing materials in advance. One reported that she asked consumers to call her when they had reviewed the materials and were ready to discuss them. If she did not hear from them in a week, she called them. By using this approach, this consultant may have reduced the chance of the materials being mislaid.

Consultants also used the home visit scheduling call to prompt consumers to prepare for the visit. They asked consumers to think about how they would use the cash allowance. Some asked consumers to start to identify a potential worker and to think about what hourly wage rates they would pay. To ensure that consumers considered payroll taxes when they thought about hourly wage rates, one consultant (who mailed materials in advance) reported specifically mentioning the tables on payroll deductions included with the training materials. Other consultants asked consumers to think about what purchase they would like to save for if they had any funds left over after paying their workers.

Page 68

-

<sup>&</sup>lt;sup>32</sup> Of those we visited, only this one agency reported passing the consulting fees on to consultants.

Consultants suggested that, to make the home visit as productive as possible, others besides the consumer be present. Some consultants asked consumers to have potential workers and potential representatives present. Others asked that *all* family members involved with the care of the consumer attend, whether or not they had been identified as a potential worker or representative. Home visits were scheduled at times when family members could be present. As Case Example V.1 indicates, having several family members present at the initial home visit could reduce the need for additional training if workers changed. Finally, a consultant reported asking consumers who did not speak English to have a translator present when she came for the home visit.

#### Case Example V.1: Value of Having Family Members Present

In most of my cases, everyone is waiting for me when I get there for the home visit. They have the worker there. They have the family members there. In one case, there were three daughters and four granddaughters at the home visit. Two of them became workers. Since that visit, one of the original workers has been replaced by one of the others also present at the home visit. That person already knew about the program, so switching workers was easy.

### 2. Initial Consulting Home Visit

Initial consulting home visits lasted two or three hours. During these visits, consultants reviewed any features of Personal Preference with which the consumers were not already familiar.

To begin to develop cash management plans, consultants sat down with consumers (and/or representatives) and carefully reviewed the financial aspects of the program.<sup>33</sup> Consultants reviewed the amount of the cash allowance that the consumer would receive. At the time of our visit, the values of Personal Preference cash plans ranged from \$262 to \$2,630 a month, with an average value of \$1,300 a month. Consultants also reviewed the allowable uses of the cash with consumers. One consultant reported that she stressed how the cash could be spent, because the letter consumers received notifying them of assignment to the treatment group was "a little vague" on that subject.

Personal Preference designed a form to help consumers and consultants think about uses of the cash. Entitled "Decision Tree for the Development of the Cash Management Plan," the form presents a list of activities in four categories (personal care, nutrition needs, mobility/transportation, and other). Consumers are asked to (1) check the activities with which they need assistance; (2) indicate whether they plan to use the cash allowance to meet each need they check; and (3) if so, whether they plan to hire workers, purchase

Page 69

\_

<sup>&</sup>lt;sup>33</sup> For brevity, we do not distinguish between consumers and representatives in the rest of this discussion of development of the cash management plan.

agency services, purchase assistive devices or other equipment or supplies, and/or modify their homes. Program staff reported that the decision tree was useful in training consultants. However, they estimated that it was completed during the development of only about a quarter of initial cash management plans.

During these discussions, consultants reported introducing other programs that provide goods and services that consumers might be able to obtain at no cost (or reduced cost), which could reduce the demands on the cash allowance. Consultants who worked for agencies that offered a variety of human services seemed especially aware of the importance of considering other services that might be available to the consumer. However, none of the consulting agencies we visited had developed a handbook of available community resources. Sometimes, services available from the community were still included in the cash management plan to avoid long waiting lists. Using Area Agency on Aging services to build a ramp was cited as an example of a service with a long waiting list.

Consultants and consumers next discussed plans to hire workers. They began with the number of workers to be hired and the hourly wage rates and cost after taxes and other expenses (such as fringe benefits). Consultants reminded consumers that they needed to identify back-up workers. One consultant reported having recommended an agency as a source of back-up workers; this consultant acted on the consumer's behalf to negotiate a rate with the agency. In her experience, agencies wanted to stay involved with these clients.<sup>34</sup>

Consultants and consumers next discussed purchases of equipment or home modifications to be included in the cash management plan. They asked consumers to identify the goods or services they wanted to save for, sometimes urging them to think creatively about how to use the cash.

In a few cases in which the consumer's requested purchase was not allowable but nevertheless had merit, the program staff suggested "trade-offs." The staff identified purchases made with personal funds that might instead be covered by the cash allowance, thus freeing personal funds for uncovered purchases. Trade-offs were rare and occurred only in cases of unallowable but meritorious requests. The possibility of a trade-off was not mentioned to most consumers.

Consultants also advised consumers on how to make purchases. For example, since the fiscal agent required a purchase order to issue a check to a vendor, consultants advised consumers to pool several small items into a single order (frequently, a catalog

<sup>&</sup>lt;sup>34</sup> However, agencies that were having difficulty hiring enough staff members to serve their regular clientele would not be a good source of back-up workers.

order). One consultant reported reminding a consumer who was bedbound that she did not need to be able to leave her home to shop--she could use the telephone to do so.

If time permitted during the initial home visit, the consultant might next discuss the forms in the employment package. This package included the forms that the fiscal agent needed to initiate employment, such as W-4 forms for federal tax withholding. The consumer, as an employer, was responsible for making sure that these forms were completed and submitted to the fiscal agent.

Often, after the initial home visit, telephone calls would be required to complete the planning process. Sometimes, additional home visits were needed. This was often true for consumers who were recruiting workers other than family members and friends.

Consultants varied in how much they helped consumers during recruitment. Some kept in close touch with consumers by telephone. Others waited for consumers to contact them when they were ready to hire. One consultant said: "Consumers are accustomed to professionals taking care of all the details and paperwork. They have to learn to be responsible for it themselves." Some consultants reported uncertainty about how much they should be helping with hiring, which probably contributed to the variation in how much help they gave consumers.

At the end of the planning process, the formal cash management plan form was completed. Consultants reported that the cash plan form (which appears in Appendix A) was not "user-friendly." Consumers and consultants found it difficult to follow. Some consultants also reported being uncomfortable with performing the calculations required to complete the form, although they did become more facile over time with doing the arithmetic. The consumer and consultant each kept a copy of the completed cash management plan, and the consultant sent a copy of the completed form to the state Personal Preference office for its review.

### 3. Program Review of the Cash Plan

Consultants reviewed cash management plans before submitting them to the Personal Preference office, but they did not formally approve the cash plans. State-level program staff members reviewed all cash plans to determine whether the goods and services requested were allowable. (See discussion in Chapter II.)

Personal Preference adopted a broad interpretation of allowable goods and services, permitting a purchase if it was demonstrated that the purchase would contribute to personal independence (if that were not self-evident). For example, the program allowed the purchase of a new mattress because an old mattress was so broken down that it hampered the consumer's ability to get out of bed without help. In contrast, a request to cover the cost of demolishing a wall in a consumer's house was denied because the

modification was not linked to her independence. Other examples of disallowed goods and services included a suite of furniture, cosmetics, and pantyhose.

Consultants sometimes advocated a purchase on behalf of a consumer. One consultant said, "I call the state program office and make the case and have not been denied."

### C. Uses of Cash

Consultants reported that nearly all consumers used the cash allowance to hire a worker (or workers). The next most commonly reported use of cash in New Jersey was to purchase equipment. Consumers often purchased medical equipment (such as lift chairs) or appliances (such as microwave ovens). As the examples in Case Example V.2 show, some consumers were creative in selecting equipment for purchase. One consultant reported that she was trying to arrange for a consumer to purchase a specially trained monkey as a service animal. She noted, "That monkey could do everything he needs."

Consultants reported that some consumers used the cash allowance to modify their homes, primarily by building ramps. The number of home modifications was likely limited because many Personal Preference participants were renters, and owner permission is usually required to modify rental property. Moreover, consumers might be reluctant spend their cash allowances to modify rental property since they would be unlikely to benefit from any increase in the value of the property. In addition, major home modifications are expensive--consumers probably would not have been able to save enough funds from the cash allowance for major modifications.

On rare occasions, consumers obtained loans instead of waiting to accumulate enough savings to make a purchase. Consultants reported that none of the consumers had applied to obtain a bank loan to be repaid from the cash allowance. However, two had borrowed money from family members and were repaying the loans in installments. One had borrowed money to have a ramp built, the other to purchase an adjustable bed.

#### Case Example V.2: Creative Equipment Purchases

A young man purchased a walker-like support for his voice synthesizer so that he could speak to people he met on the street without stopping and taking the synthesizer out of his backpack.

A blind woman bought a scanner and talking computer to allow her to read mail and check worker time sheets. She saved a portion of her cash allowance for a year to purchase this equipment.

A person with quadriplegia purchased a fax machine to use for sending medical papers to doctors and insurance companies, as well as for Personal Preference correspondence.

Under Personal Preference, consumers could use the cash allowance to cover the costs involved in managing their own services. For example, workmen's compensation was available as a rider to homeowner's or renter's insurance, and consumers who did not already have a policy were allowed to include the full cost of such insurance, not simply the cost of the rider. In addition, consumers were allowed to purchase personal post office boxes to reduce the chance of a worker's check mailed to their home being stolen.

### D. Time to Receipt of Cash

Many Personal Preference participants were enrolled in the program for several months before they received their first monthly cash allowance. Data from evaluation interviews with an early cohort of more than 200 Personal Preference participants randomly assigned to the treatment group indicated that only about a third received their first cash payment during the first six months following enrollment. Another quarter received their first cash payment between six and nine months after their enrollment. The rest of this early cohort had not received a cash payment by the ninth month following enrollment. (Future reports based on interview data will provide similar information for the entire treatment group.) Here, in this process analysis, we describe factors that contributed to delays in receiving the cash allowance.

Some factors were structural. First, allowing consumers a choice of consulting agencies added another step to a complex process, delaying the average time it took to receive the cash allowance. Second, in response to concerns from the personal care industry, Personal Preference initially gave Personal Care Assistance (PCA) providers 30 days notice to discontinue services for a client who was to become cash recipient. Partly in response to the urging of the National Program Office to reduce delay in receipt of the

<sup>&</sup>lt;sup>35</sup> Memorandum from Leslie Foster dated November 29, 2001.

first cash allowance, New Jersey later reduced the agency notice period to 14 days.<sup>36</sup> Third, consumers had to wait until the beginning of a month to receive their first cash payment, because a variable was set on the New Jersey Medicaid Management Information System (MMIS) to indicate that a beneficiary was a cash recipient (and thus not eligible to receive traditional PCA services), and the MMIS was updated only monthly. New Jersey did not develop a procedure to provide a cash payment for a partial month before the MMIS had been updated.

Consumers who did not have family members or friends to hire as workers typically had more difficulty than other consumers with completing a cash management plan fairly quickly, because they needed more time to identify and recruit a worker. In addition, they sometimes had difficulty meeting the obligations of an employer, which delayed receipt of the cash allowance. For example, one consultant cited the case of a consumer who had unrealistic expectations about the wages necessary to attract a worker. With no one available to serve as a representative, the consultant worked with this consumer to revise her expectations so that she could recruit a worker.

Finally, the development and review of every cash management plan required the participation of several people and groups. The consumer (perhaps a representative's help) and the consultant developed the cash management plan. The completed cash plan form was sent to the state Personal Preference program office for review and approval of the included goods and services. The Personal Preference office then sent approved cash plans to the fiscal agent, where the calculations were double-checked for accuracy before the consumer's cash account was established. The fiscal agent also reviewed the forms in the employment package for completeness and consistency.

If the state program or the fiscal agent identified problems with the cash management plan or the employment forms, the paperwork was returned to the consumer. A consumer who did not understand the problem and how to resolve it would have to contact the consultant. Some consultants with whom we visited complained that the fiscal agent returned paperwork without indicating what was in error. Consultants said that they themselves sometimes were unable to identify the error and had to call the fiscal agent for an explanation.

Both state Personal Preference staff and consultants reported frequent communication problems between consumers or their representatives (on the one hand) and consultants, state program staff, and fiscal agent staff (on the other). The number of people involved made it difficult for consumers to keep track of which one to call about a particular type of problem. Consumers often would call the state Personal Preference staff

<sup>&</sup>lt;sup>36</sup> The shorter notice period was apparently acceptable to personal care agencies. Senior Personal Preference staff reported that they did not receive any complaints from agencies (or, indeed, any comments at all) about the reduction in the notice period.

when they should have called the fiscal agent. Some consumers placed a call to another person without giving the first a reasonable time to respond. Other consumers called the consultant, state program office, and fiscal agent, one after the other, perhaps thinking that this would get the problem resolved more quickly. In Section F, we describe some suggestions for improving communication in the Personal Preference program.

Consultants reported that fiscal agent staff members were slow to return telephone calls both to them and to consumers. State Personal Preference staff members indicated that they, too, had noticed that telephone calls to the fiscal agent were not returned promptly when one of the two full-time fiscal staff members was out of the office.

### E. Changes to the Cash Plan

To revise a cash plan, all those originally involved (consumer or representative, consultant, state Personal Preference staff, and fiscal agent), as well as all the steps taken for the original submission of the cash plan, were required. Revisions were often necessary, as the circumstances and conditions of consumers changed. Over time, some consultants, working with the fiscal agent, developed ways to reduce the number of times the cash plan had to be formally revised and to streamline the process when revision was necessary.

### 1. When Was a Cash Plan Revision Necessary?

The cash plan had to be revised if the total amount of the cash allowance changed or if the amount to be authorized for a given type of good or service (line item) changed. Changes arose as consumers' circumstances changed, as in the following examples. A worker left a consumer's employ. A consumer had saved enough to purchase one piece of equipment and wanted to start saving for something else. A consumer wanted to change vendors. Consumers' care plan hours were increased at assessment due to a change in their condition or in the informal care available to them.

Changes were also sometimes required because state or federal policies or regulations changed. For example, changes were required in all Personal Preference cash plans when New Jersey increased the hourly rate paid for PCA services, thereby increasing the amount of the allowance for all cash recipients. A change in federal rules regarding withholding required the revision of most cash plans, since the costs of employees (and the associated line items in cash plans) changed under the revised rates.

### 2. Writing Flexible Cash Plans

Consultants and the fiscal agent learned how to write the original cash plan in a flexible way and reduce the number of circumstances in which a formal cash plan revision

was required. Instead of naming a specific individual in the cash plan, the plan would list "Worker 1" at a specified rate for a specified number of hours a month. With a such cash plan, formal revision was not necessary if a worker changed, provided that the replacement was to be paid at the same rate and work the same number of hours a month. (A new employment package would of course be required for the replacement worker.)

Similarly, the cash plan could be made more flexible by listing a type of good or service, rather than naming a vendor. For example, the cash plan might list "laundry services" rather than specify "Tom's Spotless Laundry." With such a flexible cash plan, a formal revision was not required for the consumer to change vendors (perhaps to take advantage of a better price). The fiscal agent agreed to honor purchase orders for goods and services under flexible cash plans if the cost did not exceed what was listed for that line item in the cash plan.

Not all the consultants we spoke with in spring 2001 were writing flexible cash plans. Apparently, some did not understand that it was acceptable to write a cash plan that did not name specific workers or vendors.

#### 3. More Efficient Revision of the Cash Plan

Consultants and the state Personal Preference office developed several techniques to make revision of the cash plan more efficient. One consultant created a tickler file to remind her when a consumer would have about enough money saved for a planned purchase. She would then call the consumer to ask him or her to begin to think about what good or service to purchase next and to schedule a home visit to develop the revised cash plan.

Another consultant asked consumers to give her signed, blank cash plan forms. The consultant then did not need to visit the consumer's home or to use the mail to get the revised plan signed. Instead, the consumer and consultant could discuss the necessary revisions to the cash plan over the telephone. A problem with this technique is that it is prone to error and possibly open to abuse. Consultants could inadvertently make the wrong changes to the cash plan, and the consumer would not be able to identify the error before the revised cash plan was submitted to the fiscal agent. In addition, consultants could conceivably make inappropriate changes in the cash plan without the consumer's knowledge. The consumer might not learn of an erroneous or inappropriate change until some time later when he or she received a financial statement from the fiscal agent--after funds had been expended.

At the time of our visit, a member of the state Personal Preference staff was introducing a spreadsheet version of the cash plan. She planned to distribute the spreadsheet to consultants--and some consumers--who had access to computers, as well as to use it at the state Personal Preference office. While entering the initial cash plan into

the spreadsheet was time-consuming, its use was expected to expedite cash plan revisions because it would be unnecessary to copy material that was unchanged and because the spreadsheet software would automatically do the math.<sup>37</sup>

### F. Keeping Consultants Informed

Consultants to whom we spoke indicated they needed a better way to be informed of revisions to program rules. (In a demonstration program, changes in program rules are especially likely.) One consultant suggested a monthly briefing for this purpose. Consultants also suggested meetings to learn from Personal Preference staff and from one another. Such meetings seem particularly important when consultants work alone or in small groups and when they have a small volume of cases. With experience, consultants often would develop tricks to make the work go more smoothly and quickly. At meetings, they could share these with other consultants.

### **G.** Major Lessons

Many lessons may be drawn from New Jersey's experience with Personal Preference.

### 1. Choice of Consulting Agencies

New Jersey tried to give consumers a choice of consulting agencies. However, doing so proved problematic. Some areas of the state were never served by more than one agency, and the number of agencies was greatly reduced over time. Consumers in New Jersey had little experience and knowledge on which to base their choice of a consulting agency and did not seem to place much importance on having a choice. Moreover, allowing for choice delayed the receipt of the cash allowance. While appealing in theory, in practice, allowing consumers a choice of consulting agencies under Personal Preference created problems, with little or no gain.

### 2. Advance Planning for the Home Visit Is Essential

Advance planning is essential to make the initial consulting home visit as useful as possible and to minimize the need for additional home visits. Providing written materials before the visit is not critical. It may even be counterproductive, since materials may be mislaid. Moreover, PCA recipients may find the paperwork especially off-putting if no one

<sup>&</sup>lt;sup>37</sup> We later learned that the spreadsheet version of the cash plan had been successful. Not only did it expedite cash plan revision, but it also reduced arithmetic errors in the initial cash plan and in revisions.

is there to explain it. However, before the home visit, consumers need to think about how they would like to spend the cash allowance, whom they would like to hire as a worker, and whether a representative should help them manage the cash allowance. The potential representative should attend the initial home visit, and potential workers should attend if they have been identified. It is also advantageous if *all* family members involved with the care of the PCA recipient are present during the visit. With good planning, the need for multiple home visits can be minimized, thereby eliminating a source of delay in the receipt of the cash allowance.

### 3. Techniques to Minimize and Streamline Cash Plan Revision

Consultants, state Personal Preference staff, and fiscal agent staff learned a technique to reduce the number of circumstances that required a formal cash plan revision. This involved writing a flexible cash plan that did not name workers or vendors but instead indicated the type of good or service and key cost parameters--total cost for a given type of good or service and, for workers, hourly rates and hours.

Techniques were also developed to streamline revision of cash plans when necessary. These included (1) using spreadsheets to eliminate the need to copy material that had not changed and to do the math on the cash plan, (2) using tickler files to remind the consultant to begin to plan for expected changes in the consumer's circumstances, and (3) allowing the consultant to retain blank cash plan forms that the consumer had already signed.

While the use of blank forms seems open to error and abuse, all these techniques appear to be useful in expediting revision of the cash plan. Putting the cash plan on a spreadsheet appears particularly promising. If consultants had access to laptop computers, the initial cash plan might be entered directly into the spreadsheet. To expedite review of draft cash plans, the spreadsheet might be sent electronically to the state Personal Preference office and the fiscal agent, with copies with original signatures to follow by mail when all errors had been resolved.

#### 4. Time to Receipt of Cash Allowance

One lesson learned from Personal Preference is that the time required for each step in cash planning adds up and can appreciably delay the consumer's receipt of cash. Consumers can become discouraged if they are unable to participate in the cash program as they had expected. It is important, therefore, to prevent delays in cash receipt by eliminating unnecessary steps and by minimizing the time required for the remaining steps.

To reduce delay of the cash allowance, New Jersey (partly at the urging of the National Program Office) eliminated some steps and reduced the time required for others.

It stopped giving consumers a choice of consulting agencies, as that proved to be of little value to them. Traditional providers must be notified to discontinue service to new cash recipients, but New Jersey reduced the notice time to 14 days, which seems to be adequate.

More may be possible. For example, while it may not be feasible to update an MMIS more often than monthly, Arkansas developed a procedure to allow consumers to receive a partial cash payment if they had completed the cash planning process by, say, the middle of a month. This procedure used a special revolving fund to cover the cost of a partial cash payment for one month.

### 5. Resolving Communication Problems

Personal Preference experienced serious communication problems. These problems contributed to delays in completing the cash management plan and getting the cash allowance. Resolution of such problems could reduce the time to receipt of cash.

Those we interviewed identified several ways to improve communication in Personal Preference. One suggestion was to train consumers about whom to call regarding what issues, including providing them with a "Who Does What" list. Another suggestion was that, when the state Personal Preference office or the fiscal agent returned documents for correction, exception reports be produced. These reports would indicate the error requiring resolution. Another consultant suggested using multiple-party telephone calls to reduce the need for several one-on-one calls. Case Example V.3 presents this consultant's views on the value of these calls. The potential for communication problems is high in Personal Preference because so many people are involved in its day-to-day operations. We return to this issue in Chapter IX.

#### **Case Example V.3: Multiple-Party Telephone Calls**

When I have an issue with the fiscal agent, I do a three-way call. The consumer should call, but I want to see how to handle the issue. The consumer is involved and hears the transaction.

I say to the consumer, "Let's call together. I'll put you on hold and get the fiscal agent on the line." When I reach the appropriate member of the fiscal agent staff, I say, "I have a consumer on the other line and we want to talk to you." The call helps my credibility with both the consumer and the fiscal agent.

I do these calls with the state office, too. They are very effective.

The Arkansas experience suggests a second approach to resolving communication problems. This involves limiting the role of the state staff in reviewing cash management plans. Under the Arkansas cash program, a list of allowable goods and services was

prepared. The Arkansas state program staff reviews only those cash plans that call for unlisted goods or services. Through the Personal Preference demonstration, New Jersey has learned how to develop a list of allowable goods and services.<sup>38</sup> Particularly if New Jersey adds a cash model to its permanent Medicaid program, there would seem to be little or nothing to gain from continued state-level review of all Personal Preference cash plans.

\_

<sup>&</sup>lt;sup>38</sup> After our visit, we learned that New Jersey plans to create a list of goods and services that would be clearly allowable in a future cash allowance program. Some would be allowable only in specified circumstances. For example, blenders and food processors would be allowable only for consumers who had difficulty with eating.

### VI. REPRESENTATIVES

Many of those receiving the cash allowance under Personal Preference had representatives to help them manage it. Consultants reported that up to two-thirds of the elderly consumers had representatives, while the proportion was considerably smaller among nonelderly adults. Ethnic and language diversity likely contributed to the extensive use of representatives among elderly consumers in New Jersey. Those not fluent in written English probably were not comfortable handling the paperwork, such as the development of the cash management plan that Personal Preference required.

In this chapter, we describe the Personal Preference procedures pertaining to representatives. We also present lessons we have drawn from New Jersey's experience with the cash program.

### A. Selection of Representatives

In general, consumers could decide whether they wished to have a representative help them manage the Personal Preference cash allowance. Consumers who had legally appointed surrogates (such as legal guardians and social security payees) had to name their surrogates as representatives, but no one else who entered Personal Preference was required to have a representative. Rarely, New Jersey required that a representative be named if the consumer was to continue to participate. By the time of our visit in spring 2001, the state had mandated representatives in only three cases. For example, the state mandated a representative for a consumer who was making unreasonable demands on her employees.

The Personal Preference enrollment staff discussed the possibility of naming a representative when they contacted a Personal Care Assistance (PCA) recipient. However, representatives were not formally identified during an enrollment visit--the consultant did this after random assignment to receive the cash allowance. By delaying identification until after assignment, New Jersey avoided naming representatives for those PCA recipients later assigned to the control group.

During the initial home visit, the consultant discussed and assessed the consumer's need and desire for a representative. The consultant looked at the types of assistance the consumer was currently receiving. Consumers who were already receiving assistance with their affairs, such as help maintaining a checking account, were viewed as likely to need a representative. Those who were accustomed to assistance with their affairs usually were happy to name a representative when a consultant recommended that they do so.

Usually, the choice of a person to serve as representative was obvious and grew out of the current relationships of the consumer. It was usually someone who already had been helping manage the consumer's financial affairs or services. Most had been giving informal help; few consumers had legal guardians or other surrogates who had been legally appointed.

Nearly all the representatives were family members--usually, close family members. Consultants reported that elderly consumers seemed more comfortable with the cash program if a family member was named as a representative.

Consultants helped prospective representatives decide whether they wished to undertake this role. They described the responsibilities that representatives were expected to assume and assessed whether prospective representatives had realistic expectations. They were asked to complete a self-screening questionnaire that the state Personal Preference staff had developed (see Appendix A). The items in this questionnaire raised pertinent issues, such as the frequency of contact between the consumer and the prospective representative and whether the latter was paid to care for the consumer. Another item asked prospective representatives to confirm their understanding that they could not pay themselves in that role. Many representatives were pleased that their willingness to serve as a representative allowed their loved ones to participate in Personal Preference.

If a representative agreed to serve after completing the self-screening questionnaire, both the representative and the consumer signed a separate form designating the representative. This form could be completed during the consulting home visit if the representative was present.

Infrequently, representatives were named after the initial home visit. A consultant might recruit a representative for a consumer who was having difficulty developing the cash management plan or managing the cash allowance.

### **B.** Major Lessons

New Jersey's Personal Preference program offers several lessons about representatives.

### 1. Selection of Representative Is Extension of Natural Process

The representative selection process was an extension of the relationships that consumers already had when they joined Personal Preference. Most representatives were relatives (usually close relatives) of the consumer. As indicated earlier, many consumers were already receiving help with their affairs, and the naming of a Personal Preference

representative was a formal extension of that assistance. Some may have been receiving assistance because of their limited facility with English. In general, those who had been providing assistance simply assumed the role of representative.

The fact that representatives were generally close relatives probably made it easier for them to interact with providers and vendors. Many institutions (for example, banks and hospitals) are used to having close relatives act on behalf of their elderly or disabled loved ones. None of the consultants or other program staff members we interviewed reported any problems that arose because of questions about the legal standing of representatives.<sup>39</sup>

Representatives were usually happy to serve. Often, they wanted a loved one to participate in Personal Preference, and acting as a representative made that possible. In addition, the availability of fiscal services (at low cost to the consumer) may have made the selection of representatives easier by limiting the duties expected of them.

### 2. Need for Representative Depends on Several Factors

The need for a representative depends on the consumer's type and level of disability. For example, a consumer with a cognitive disability is likely to need help managing his or her affairs.

Other factors also affect the need for a representative. They include familiarity with the cash program, the availability of consultant training about how to manage cash, and the availability of equipment to support management activities. The familiarity of the consumer and of the consumer's family with the cash program affected the need for, and choice of, a representative. Many consumers originally named representatives for security, since the cash program was new and they were new to it. As they became more familiar with the program, they were able to assume more responsibility for managing the cash allowance. In some cases, a son or daughter might have pressed to be a representative early, out of concern about a parent's ability to manage the cash allowance without assistance. This same son or daughter might have felt more comfortable once the cash management plan was completed and might have allowed the consumer to assume more responsibility for his or her own care.

A sense of insecurity probably explains why some consumers named representatives when they could have managed without them, at least in the view of the program staff. The state Personal Preference staff estimated that about half of all consumers who named a representative would have been able to manage the cash allowance independently had they chosen to do so.

\_

<sup>&</sup>lt;sup>39</sup> After our visit to New Jersey, the National Program Office arranged for an attorney to review legal issues pertaining to representatives. It did so at the request of some of the Cash and Counseling programs.

The need for a representative could also depend on the availability of consumer training from the consultant. For example, one consultant asked a son to be a representative. When the son did not complete the forms, the consultant taught the consumer how to treat a worker so that she could manage the cash allowance herself.

Rarely, the need for a representative can depend on the consumer's environment. The blind woman mentioned in Case Example V.2 offers an example of how the environment affects the need for a representative. She purchased an electronic scanner and talking computer with the cash allowance so that she could personally read worker timesheets, as well as correspondence regarding the Personal Preference program. Her consultant had suggested naming a representative to review documents, but she resisted because she wanted to be independent. The equipment made that independence possible.

### 3. Role of the Representative Varies

Consultants encouraged representatives to involve consumers in decision making as much as possible. Typically, both the consumer and the representative contributed to the management of the cash allowance, although the nature and extent of their contributions varied.

Representatives generally asked consumers about their preferences. The representative was the sole decision maker only in the rare cases in which the consumer was unable to communicate his or her preferences or unable to act in his or her own best interests. For example, a representative for a New Jersey consumer who was nonverbal and had profound mental retardation was reportedly the sole decision maker. However, representatives sometimes involved consumers with severe disabilities in decision making. One consultant reported an example of a parent who was the representative of a young man with cerebral palsy. Although others could not understand his speech, his parent could and kept bringing him into discussions about his care, turning to him and asking, "What do you think?" For example, he was consulted about whether he would like to use some of the cash allowance to buy a cordless phone.

Despite disability, a consumer may be the primary decision maker, while the representative serves as the consumer's liaison with the outside world. In these cases, the designation of a representative may not be strictly necessary, but it may be more efficient. The representative can reduce the time required from the consultant and reduce the burden on the consumer.

# Case Example VI.1: Representatives When the Consumer Is Primary Decision Maker

A man whose speech was severely impaired after a stroke (but whose other mental abilities were unaffected) selected his wife as his representative. She is the only one who can understand his speech and has even gone into the operating room with him to translate for his physicians.

A young man who is attending college selected his mother as a representative, not because his disabilities prevented him from undertaking the task, but because he was busy going to class and studying during the day. The mother was at home, with more time to interact with the consultant and the fiscal agent.

### 4. Representatives Are Functioning Well

All the consultants with whom we spoke reported that representatives were obtaining input from the consumers when possible and were faithful to consumers' best interests. All these representatives were relatives of consumers. No incidents were reported in which consumers had changed representatives. Consultants described some of the representatives as "very close" to the consumers.

## VII. CONSUMERS AS EMPLOYERS

For consumers who hired workers under Personal Preference, the success of consumer direction depended on the consumer's ability to be a good employer--to make decisions about recruiting, hiring, supervising, and (if necessary) firing workers. <sup>40</sup> Although consumers could not delegate these tasks (as they could fiscal tasks), they did have help carrying out their role as employers. Consultants provided much of this assistance, and the fiscal agent helped with particular tasks. In this chapter, we discuss consumers' performance of their employer responsibilities.

### A. Assistance with Employer Responsibilities

Consultants trained consumers on how to recruit, hire, train, supervise, and fire workers. However, consultants did not perform these tasks for consumers.

#### 1. Initial Assistance

Consultants began to train consumers on employer responsibilities during the initial consulting home visit. As consultants helped consumers develop the cash management plan, they told them the prevailing wage rates in their community. For example, a consultant in the northern part of the state said that she told consumers that traditional agencies were raising hourly wage rates for their aides to \$9 to \$10 an hour and offering them a benefit package. Occasionally, a consumer had unrealistic views about wage rates. For example, one consumer wanted to hire a live-in worker for \$700 a month and charge that worker \$500 a month for rent.

During the initial home visit, consultants stressed how much payroll deductions would add to a given hourly wage. Consultants reported that the "tax tables" the state Personal Preference staff had developed were helpful. These tables presented the cost to the consumer to pay a given hourly wage, with all payroll deductions taken into account (for example, an hourly wage of \$11.50 would cost the consumer \$12.35 an hour).

Personal Preference workers were considered domestic workers. State law in New Jersey requires that employers (including employers of domestic workers) carry workmen's compensation insurance, and insurance companies must offer this under low-cost riders to homeowner and renter insurance policies. Early in the demonstration, it was difficult to find an agent who would write the riders for renter insurance policies, which no

<sup>&</sup>lt;sup>40</sup> For brevity, we do not distinguish between consumer and their representatives with respect to fulfilling the responsibilities of an employer.

longer were common in the industry. Because the cost for the rider was low, an agent received little compensation for writing it. However, consultants helped consumers obtain the riders. One consultant reported that she finally convinced an insurance agent to write workmen's compensation riders to renter policies by arguing that he would receive a substantial volume of business from consumers whom she and other consultants referred to his agency. Eventually, consultants identified a handful of agents in the state who were willing to write the riders.

### 2. Recruiting Workers

Consultants used different approaches to helping consumers during recruitment. Some kept in close touch with consumers by telephone. Others depended largely on the initiative of the consumers, expecting that they would contact the consultant when ready to hire. As noted in Chapter V, some consultants reported uncertainty about how much help with recruiting they should provide, which probably contributed to the variation in the amount of assistance they did provide with this activity.

As mentioned in Chapter V, most consumers hired family member or friends as workers. Consultants advised consumers who wanted to hire a family member that doing so might not always work out. Consumers needed to understand that it could be difficult to discipline a worker who is a family member. One consultant reported that she also advised consumers to be careful not to hire a family caregiver who already was "burning out" from providing unpaid care.

Some consultants gave suggestions on recruiting techniques to those who did not have a family member or friend they wanted to hire. They shared their personal experiences with consumers and gave them copies of materials on recruiting. They copied materials on advertising, interviewing, and other aspects of recruiting from the Personal Preference consultant training manual and gave them to consumers. One consultant also gave consumers copies of materials developed by another program because she thought they were helpful. Because a newspaper ad could cost as much as \$100, she found the manual's suggestions about places to post free ads (for example, churches, colleges, Laundromats) especially useful. Another consultant reported frequently recommending that consumers seek workers through their churches.

As Case Example VII.1 suggests, recruiting a worker could be difficult, and this difficulty sometimes led to delay in receipt of the cash allowance.

# Case Example VII.1: Difficulty Recruiting a Worker Leads to Delay in Cash Allowance

One young man living in a supportive housing facility had no family members or friends whom he wished to hire as a worker. All his neighbors also had disabilities. The lack of bus or other public transportation made it difficult for potential workers to travel to the supportive housing facility to care for him. He tried to hire the janitor in the facility as his worker, but the housing management refused to allow that. He then tried to hire the janitor's wife, but the management disapproved of that as well, leaving him without a worker at the time of our visit.

Personal Preference consultants pointed out to consumers that they could use agency services if they were unable to identify a worker. One consumer hired an agency that charged \$15 an hour--much less than traditional agencies usually charge. Another consultant reported that she had negotiated special rates with a traditional agency to provide back-up care for a consumer.

### 3. Hiring Workers

Consumers had to ensure that the workers they recruited completed employment packets. Consultants left one or more packets with the consumer at the initial home visit for the worker(s) to complete. The fiscal agent provided consumers with instructions on how to complete forms in the package. Although state and federal regulations require that these forms be completed in English, the fiscal agent prepared the instructions in both English and Spanish.

The fiscal agent also reviewed the completed employment packages. As with the cash management plans, consultants reported that they occasionally had to intervene with the fiscal agent about employment paperwork returned to the consumer for corrections. In these cases, the fiscal agent had not specified the nature of the error, and it was not obvious to the consultant.

#### 4. Time Sheets

Consultants trained consumers and workers on the importance of submitting time sheets on schedule. Consultants advised workers that they were responsible for making sure the time sheet was mailed on time. Many consumers were homebound and had difficulty getting to a mailbox or post office. Consultants taught consumers and workers to leave enough time for the mail to travel to the fiscal agent's office in the northern part of the state.

Fiscal agent staff reinforced the training about time sheets on an ad hoc basis. When consumers called the fiscal agent to complain that worker paychecks had not been received as expected, the fiscal agent investigated. Often, the time sheets had not been

submitted on schedule. This situation afforded an opportunity to retrain the consumer about the importance of prompt submission of time sheets. After the consumer and the worker learned what was necessary, they usually submitted time sheets on schedule.

A small minority of consumers and workers repeatedly failed to submit time sheets on schedule. However, a fiscal agent executive reported that the kinds of problems she had seen with time sheets in Personal Preference were no different from those in the agency's other human services programs.

#### 5. Firing Workers

The consultants with whom we spoke did not report helping any consumer to fire an unsatisfactory employee. One consultant reported counseling a consumer about giving appropriate notice when terminating an employee. Case Example VII.2 describes this.

#### Case Example VII.2: Firing Workers Without Notice

A consumer had fired a worker (who had been performing well) without notice so that she could hire her daughter-in-law. The worker complained to the consultant, but there was nothing the consultant could do. When the consumer's arrangement with the daughter-in-law fell through soon thereafter, the consumer wanted to rehire the worker and called the consultant to discuss doing so. The consultant told the consumer that the worker was upset about being fired without notice, to which the consumer replied, "No one ever told me I was supposed to give a worker notice before firing them."

# **B.** Consumer Management of Payroll Functions

As indicated in Chapter II, New Jersey permitted consumers to take personal responsibility for managing their cash allowance--provided that they passed a skills examination covering preparation of payroll tax returns and related payroll documents. The skills examination was prepared by the fiscal agent and revised based on comments from a contractor the National Program Office had hired. The examination was "open book"--consumers were allowed to consult relevant materials, such as the instructions for completing payroll tax returns, while taking it.

A few consumers asked to review the skills examination. Five had done so at the time of our visit. However, after reviewing the examination, all five decided not to take it. One consultant reported that a consumer with whom she worked was part of a family that ran a small business. This consumer had planned to take the examination; however, it was still being developed at that time. The consumer decided not to take the examination after it became available, as the family was happy with the services they had been receiving in the interim from the fiscal agent. One fiscal agent executive reported that the "taxes scare

everyone." Apparently, consumers preferred paying the charges for fiscal services to taking personal responsibility for preparing payroll tax returns and other payroll documents.

# C. Major Lessons

We draw three major lessons from the experience of consumers as employers under Personal Preference.

#### 1. Consumers Fulfilled Nonfiscal Responsibilities of an Employer

Most consumers satisfactorily fulfilled the nonfiscal responsibilities of employers-recruiting (which proved the most difficult), hiring, training, supervising, and firing. Consultant training of consumers played a major role in this success.

The amount of help that consumers need from consultants varies. Some consumers have substantial life experience as employers (for example, they might have helped run a family business). Others have little relevant life experience and may need a substantial amount of consultant assistance. When they enroll in a consumer-directed program, they may not understand the basic responsibilities of an employer. They may not understand simple issues, like the consequences of failing to submit a time sheet on schedule, or sensitive ones, like the importance of giving adequate notice of termination. Although such consumers need much more consultant assistance in fulfilling the responsibilities of an employer, they can learn.

#### 2. Recruiting Is a Critical Issue

Under Personal Preference, recruiting was perhaps the most critical issue for consumers who did not have family members or friends they wished to hire as workers. When asked what consulting functions were most critical in enabling consumers to effectively manage their cash allowance, one consultant replied, "Knowing where to go if they have problems, so they can move on. The biggest problem is where to look for workers." The receipt of cash will be delayed for consumers who cannot recruit a worker called for in their cash plan. Program staff reported that consumers who could not recruit a worker tended to become frustrated by this and eventually dropped out of the cash program.

Personal Preference developed written materials to train consultants about recruiting, and the consultants shared these and other materials with consumers. However, consumers received varying amounts of help from consultants with recruiting. Although consultants realized that they were not to serve as case managers, they still had to determine what their new role would be. Both consultants and consumers might have

benefited from more training for consultants on their responsibilities in helping consumers with recruiting.

What else might be done to help consumers recruit workers? The National Program Office helped New Jersey and the other Cash and Counseling states identify approaches to link consumers with workers. One possibility is the development of worker registries. Neither the state Personal Preference office nor the consulting agencies developed worker registries. However, we learned after our visit that the state of New Jersey had applied for, and received a federal grant for, a program to develop worker registries. Part of the impetus for applying for this grant was the state's experience with Personal Preference.

Another possibility is to develop informal lists of Personal Preference workers willing to assist other consumers. The Personal Preference consultants with whom we spoke did not report having such lists. The volume of consulting cases for many Personal Preference consultants may simply be too small to give rise to the development of worker lists.

In addition, consumers who have difficulty recruiting workers could make greater use of agency services, at least until a worker can be identified. Less expensive alternatives to traditional Personal Care Assistance services do exist in New Jersey, and traditional agencies may be willing to negotiate special rates.

Finally, part of the difficulty of recruiting workers under Personal Preference probably was due to a full-employment economy. During our visit, we learned of shortages of personal attendants for other home- and community-based services. During a downturn in the economic cycle, consumers may find it somewhat easier to recruit workers.

#### 3. Consumers Prefer to Share Fiscal Responsibilities of an Employer

Consumers value having the services of a fiscal agent to provide bookkeeping and prepare payroll tax returns and related documents. Under Personal Preference, consumers could assume all fiscal responsibility for the cash allowance if they passed an open-book skills examination. However, no consumers chose to take the test. Rather, all chose to split the fiscal responsibilities of an employer between themselves and the fiscal agent--they retained responsibility for approving and submitting time sheets, and the fiscal agent assumed responsibility for filing state and federal tax returns and related documents. With training from consultants, almost all consumers successfully completed the fiscal responsibilities they chose for themselves.

Is there cause for concern because consumers in Personal Preference unanimously elected to use the services of the fiscal agent? Does their use of this service suggest a lost opportunity for further training to foster independence? We think not. The state of New Jersey was happy to have consumers use the fiscal services and encouraged them to do so by subsidizing the cost of these services. Moreover, the general public often chooses

to pay experts rather than to prepare their own tax forms. Surely we should not expect more of participants in a cash program.

# VIII. MONITORING TO PREVENT ABUSE AND EXPLOITATION

The possibility that consumers could be exploited or that the cash allowance could be abused was a major concern of all involved in the Cash and Counseling Demonstration. Policymakers wanted to ensure that vulnerable consumers were not harmed and that public funds were not squandered. Even so, everyone involved realized that extensive control and oversight are inconsistent with consumer direction. Consumers must have freedom to make choices and manage on their own, even if others view some decisions as misguided.

In this chapter, we describe the methods Personal Preference used to prevent abuse of the allowance and neglect or exploitation of consumers. As of the time of our visit in spring 2001, there had been no abuse of the cash allowance or exploitation of cash recipients under Personal Preference.

#### A. Preventing Abuse of the Cash Allowance

The primary method Personal Preference used to ensure appropriate uses of the cash allowance was state approval of the cash management plan, coupled with fiscal agent review to verify that expenditures were authorized under the cash plan. Personal Preference also provided for consumer review of monthly financial statements from the fiscal agent.

## 1. Fiscal Agent Checks That Expenditures Were Authorized

Perhaps the most important method for preventing abuse of the cash allowance was rigorous fiscal agent review of purchase orders against cash management plans. After the fiscal agent received an approved cash plan and checked it for accuracy, the information in the plan was data entered (using the Peachtree accounting software package) to create a consumer-specific budget. Thereafter, purchase orders were compared to the Peachtree budget as they were received (in this context, a time sheet constitutes a purchase order). Only those purchase orders consistent with the budget (and thus the cash plan) were approved, and an approved purchase order was required before any check was cut. One fiscal agent executive put it this way: "If the purchase order does not mirror the cash management plan, the check is not cut." Before a check was sent to the consumer, the accounting supervisor and the fiscal agent project manager both rechecked the amount against the Peachtree budget.

Others concurred that the fiscal agent's procedures were rigorous. State Personal Preference staff members were pleased with the stringent approach, arguing that New Jersey could afford to be more flexible in approving uses of the cash because it could be confident that actual expenditures would reflect those allowed under the cash management plan. One consultant commented that the fiscal procedures were "so tight ... I can't think how [the procedures] could be abused unless the worker and the consumer were in cahoots."<sup>41</sup>

#### 2. Monthly Financial Statements

New Jersey's contract with the fiscal agent called for the agent to prepare a monthly financial statement for each consumer receiving the cash allowance (with a copy to the state). The main purpose of these statements was to ensure that consumers knew their account balances. However, they also provided a way for consumers to identify error by the fiscal agent. Consumers might also identify error or even abuse by consultants who retained blank, signed copies of cash management plans in order to expedite revisions to cash plans.

At the time of our visit, the requirement that the fiscal agent prepare monthly financial statements for consumers had not been met satisfactorily. The initial financial statements that the fiscal agent had prepared were difficult to understand--even the director of Personal Preference said he did not understand them. In spring 2001, distribution of monthly financial statements to consumers had been suspended until the format of the statement could be revised to make it more user friendly. However, a revised format could not resolve one source of confusion. It was not feasible to list the names of workers who had been paid on the financial statement; the accounting system the fiscal agent was using would support listing them only as "Worker 1," "Worker 2," and so on. (Nonlabor invoices generally could be described in enough detail to avoid confusion.)

Because the financial statements were difficult to understand and irregularly produced, they had limited usefulness in preventing inappropriate use of the cash allowance.<sup>42</sup> Consultants reported that consumers could not use the financial statements to monitor the performance of the fiscal agent, because they did not know the status of their accounts.

<sup>&</sup>lt;sup>41</sup> During our visit, we learned of one case in which a consumer had reported that another consumer and a "worker" were in cahoots to falsify a time sheet and split the earnings between them. Later, we learned that Personal Preference program staff had investigated this case and had judged that nothing untoward had happened. At that time, we also learned of another case in which one consumer reported that a second consumer had been falsifying a worker's time sheet. Again, nothing appeared out of order upon investigation by program staff, who later learned that the two consumers in the second case had a long-standing feud.

<sup>&</sup>lt;sup>42</sup> After our visit, Personal Preference program staff reported that financial statements were being prepared in a timely way.

However, even user-friendly financial statements would not have been an entirely satisfactory way to provide many consumers with information about the status of their accounts. Many consumers simply assumed that all invoices that had been submitted to the fiscal agent had been processed and treated the "bottom line" on the financial statement as accurately reflecting their *current* balance. They did not examine the statement to identify any invoices that were still pending at the time the financial statement was prepared. To avoid this problem, Personal Preference staff recommended to consumers that they telephone the fiscal agent when they needed information on their current account balance.

# **B.** Monitoring to Prevent Exploitation

Under Personal Preference, two methods were used to monitor for neglect or exploitation of the consumer: (1) contact between consultants and consumers, and (2) reassessment visits by independent Medicaid nurses.<sup>43</sup> Personal Preference did not have any explicit mechanisms to prevent exploitation of workers.

#### 1. Contact Between Consultants and Consumers

Consultants and consumers were in contact by telephone and in person. Some contact was routine, while some focused on resolving problems. All contacts provided an opportunity to monitor the condition of the consumer and identify or prevent neglect and exploitation.

Personal Preference required that consultants speak to consumers at least monthly by telephone for the first six months after demonstration enrollment and see them in person quarterly. Consultants worked with consumers to develop the cash management plan. After this was done, consultants told consumers to call them when they had a question. Some consumers, especially those new to the program, called their consultants frequently-more frequently than necessary, according to one consultant. Another said, "I hear from one [particular] consumer every week." Other consumers seldom called. If a consultant had not heard from a newly enrolled consumer in a month, the consultant would call the consumer to fulfill the requirement for monthly telephone contact.

Consultants' monthly calls to consumers were not structured--there was no required interview or checklist (although consultants were required to collect a minimum set of information on consumers). Consultants usually asked how the consumer was getting along and listened for hesitation or other subtle clues that suggested the consumer might be in trouble and that the consultant should investigate further. During the monthly calls, the

Page 95

-

<sup>&</sup>lt;sup>43</sup> In the traditional program, nurses from the home care agency make reassessment visits.

consultants also reminded consumers about the process for revising the cash plan, including the need to start to think about revising the plan when that was pertinent. Consultants also responded to consumers' questions during the monthly telephone calls, which gave the consultants an opportunity to teach consumers about handling specific problems.

Consultants held different views on whether monthly calls were necessary for the first six months. Some endorsed the calls, while others considered them unnecessarily frequent for some consumers. Case Example VIII.1 presents different views on the value of monthly telephone contact between consumers and consultants.

#### Case Example VIII.1: Value of Monthly Telephone Contact

**Consultant A**: It is good to be in touch once a month. Sometimes, the consumer will say something like, "Oh yeah, I was thinking of calling you." Your call may remind them of something that was on their mind.

**Consultant B**: I call them if I do not hear from them for a month. They get tired of my calling. I always ask the same questions. It becomes kind of a joke with some of them. But they understand that I am supposed to call.

Consultants reported seeing consumers in person at least quarterly, as the Personal Preference program required. These visits might be in conjunction with obtaining a signature on a revised cash management plan, although one consultant (who had a small caseload) reported that she made quarterly visits whether or not she had to visit for a special reason. The consultants we talked with did not question the value of quarterly visits.

#### 2. Reassessment Visits

Under contract to the state Personal Preference program, nurses employed by the Medicaid program visited recipients of the cash allowance semiannually to reassess their condition and to develop their care plans for the ensuing six months.<sup>44</sup> During these visits, the Medicaid nurse consultants also had an opportunity to monitor the condition of the consumer and to identify or prevent neglect and exploitation. Through the reassessment visits, the Medicaid nurses had identified instances in which consumers needed more assistance, and care plan hours were increased accordingly.<sup>45</sup> They also identified cases in which the consumers were receiving more hours of personal assistance than were

<sup>&</sup>lt;sup>44</sup> Medicaid beneficiaries receiving traditional Personal Care Assistance (PCA) services (including treatment group members who dropped out of Personal Preference) were reassessed by nurses employed by the PCA agency.

<sup>&</sup>lt;sup>45</sup> The nurses also identified instances in which they judged that care plan hours were excessive; the consumer's hours and cash allowance were reduced accordingly.

justified, and care plan hours were reduced.<sup>46</sup> However, no instances in which consumers were being neglected or exploited had been identified during reassessment at the time of our visit in spring 2001.

#### 3. Exploitation of Workers?

The consumer, as employer, was responsible for employee working conditions. Consultants were trained to respect the consumer's role as employer, and they did not routinely contact workers. However, workers occasionally called consultants or the state Personal Preference office to complain about ill treatment by consumers. It is possible that lack of explicit reporting mechanisms caused some exploitation of workers to be unreported. One possible mechanism might be to give workers the state's toll-free telephone number and invite them to call if serious problems arise in their work environments.

# C. Little or No Exploitation or Abuse

In this section, we first consider the limited evidence of neglect or exploitation of consumers and workers and then the lack of evidence of abuse of the cash allowance. Personal Preference experienced no adverse publicity from what little exploitation occurred.

#### 1. Very Little Exploitation of Consumers and Workers

State Personal Preference staff reported that only a few cases were identified in which a consumer may have been subject to neglect or exploitation. These cases were quickly investigated--before the cash allowance was received--and handled appropriately. Case Example VIII.2 describes the only case of potential neglect or exploitation that the consultants we visited mentioned; it illustrates how a consultant might identify a case of potential neglect or exploitation.<sup>47</sup>

When a consultant reported that something might be amiss, the state Personal Preference office referred the case to a Medicaid nurse, who visited the home to make an assessment. If during the home visit the nurse obtained information that suggested that the consumer had been abused, the state Personal Preference office checked with Adult

<sup>&</sup>lt;sup>46</sup> Unlike some of the nurses employed by personal care agencies, the Medicaid nurses had no personal financial interest in the number of hours in care plans. Some consumers appealed the reduction in their care plan hours, thereby initiating the state's fair hearing process.

<sup>&</sup>lt;sup>47</sup> It is possible that cases of potential neglect or exploitation could be identified prior to enrollment in Personal Preference. However, no instances in which enrollment staff identified a case of potential neglect or exploitation were reported to us.

Protective Services to see whether that office had a record on the case. The Personal Preference staff reviewed the report of the Medicaid nurse and any other evidence in cases of potential neglect or exploitation. If it concluded that neglect or exploitation was likely, the case was referred to Adult Protective Services, and the consumer was disenrolled from Personal Preference and returned to traditional PCA if appropriate.

#### Case Example VIII.2: Handling Cases of Potential Neglect or Exploitation

A consultant indicated that she had felt that something was amiss when she scheduled the initial home visit and that she had, therefore, asked another consultant to accompany her. When they arrived, they found the consumer lying on a couch, comatose. Both consultants felt that this situation was inappropriate, and that Adult Protective Services should be notified. They referred the case to the state Personal Preference office for investigation.

Consumers occasionally treated workers unfairly, but the consultants we visited mentioned only one case in which a consumer seems to have been exploiting a worker. In this instance, a family dispute arose when the son and daughter-in-law, who had been hired as a worker, moved out of the consumer's home, and the consumer withheld the daughter-in-law's paychecks. The worker complained to the consultant, who notified the state Personal Preference office. After a member of the state Personal Preference staff reminded the consumer that, as an employer, she was responsible for paying people who worked for her, she paid the daughter-in-law, thereby resolving the problem.

#### 2. No Abuse of the Cash Allowance

No one we spoke with--representatives of the personal care industry, consultants, fiscal agent staff, or state Personal Preference staff--reported having seen evidence of material abuse of the cash allowance under Personal Preference. A few consumers asked to use the cash allowance for purchases not related to their independence, but such requests were denied.

# **D.** Major Lessons

We draw three major lessons about preventing abuse and exploitation from the experience of Personal Preference.

# 1. Cases of Possible Exploitation of Consumers Can Be Eliminated Before Cash Payment

Unfortunately, elderly and disabled Americans are sometimes neglected and exploited. Therefore, it is not surprising that consultants suspected that a few PCA

recipients entering Personal Preference were being so treated. The state Personal Preference office thoroughly investigated such cases before making a cash payment, and no cases of exploitation were reported after payments began. We conclude that consumers can be protected from the potential neglect and exploitation associated with a cash allowance program if staff are mindful of that possibility and attentive to cases where something seems amiss as they make their initial visits.

#### 2. More Attention to Preventing Exploitation of Workers May Be Warranted

At the time of our visit, no cases of cash recipients being exploited by their workers, friends, or family had been identified, but one case in which a consumer exploited a worker had been. Personal Preference had no formal mechanism for workers to voice grievances about their employment situations. The one case reported suggests that such a mechanism might be useful, especially since consultants were trained not to interfere with the consumer-worker relationship.

#### 3. Ensuring That Expenditures Are Authorized by a Cash Plan Prevents Abuse

From the Personal Preference experience, we draw the lesson that abuse of a cash allowance--even of a substantial amount--is preventable. At the time of our visit, Personal Preference had been operating for more than two years, and there had not been a single incident of cash allowance abuse. This was accomplished by developing a cash plan, taking care that it did not allow inappropriate goods and services, then following rigorous procedures to ensure payment for only those goods and services covered in the cash plan.

Review of receipts was not required and thus was not important in preventing abuse of the cash allowance in New Jersey. It might have been important if consumers had been managing the cash allowance themselves, but at the time of our visit all were using the fiscal agent.

# IX. LESSONS FROM NEW JERSEY

New Jersey's experience with the Cash and Counseling model provides many valuable lessons. In this chapter, we first consider lessons related to the political and administrative feasibility of the Cash and Counseling model in New Jersey and examine whether Personal Preference was implemented as planned. Second, we present lessons about procedures, summarizing those drawn in previous chapters about particular components of Personal Preference, and we draw lessons that cut across components about the basic structure of the Cash and Counseling model as it was developed in New Jersey. Finally, we address the value of the Cash and Counseling model relative to a model of consumer-directed care offering only a cash allowance.

# A. Lessons About Feasibility

New Jersey's primary goal in implementing the Cash and Counseling Demonstration was to test the feasibility of including a cash allowance model of consumer direction as a state plan Medicaid service and to investigate the main issues involved in implementing such a program.

New Jersey achieved this goal. Unopposed by the state's Nursing Practice Board and overcoming initial opposition of traditional agencies, the state implemented Personal Preference. It implemented the design largely as planned, discovered some problems with key parameters, and--learning from its experience--changed its procedures. New Jersey demonstrated that a cash program can work both administratively and politically in that state.

#### 1. Nurse Practice Act Was Not a Barrier

New Jersey's Nurse Practice Act limits the medical tasks that a personal care aide can legally perform and requires that a nurse supervise aides. The state board responsible for nursing practice did not feel that the activities of workers hired by consumers should be similarly limited or that consumers did not meet the requirements for supervision because they were not nurses. The director of Personal Preference presented the case for the Cash and Counseling model to officials of the board. Reportedly, because of the shortage of personal care aides in New Jersey, they viewed Personal Preference as a valuable addition that tapped a different labor supply.

#### 2. Overcoming Opposition of Traditional Agencies

To implement Personal Preference, New Jersey overcame initial opposition from personal care agencies. Their cooperation was important, because without it New Jersey could not identify personal care recipients and the hours of care planned for them. Moreover, agency staff members could actively discourage Personal Care Assistance (PCA) recipients from participating in the demonstration. Although some personal care agency staff members did do this, agencies supplied care plan data as requested, and the industry was generally cooperative.

Several factors explain why New Jersey was able to overcome the initial opposition of personal care providers. First, as Case Example IX.1 indicates, the personal care industry respected the director of Personal Preference and the state program staff. Second, the industry saw the adoption of consumer-directed care as inevitable and, indeed, beneficial for some people with disabilities. Third, Personal Preference responded to industry concerns, especially by discouraging consumers from hiring agency employees as workers. The industry lost few staff members to Personal Preference consumers. Fourth, because a smaller percentage of PCA recipients participated in Personal Preference than initially anticipated, agencies did not lose a substantial amount of business.

#### Case Example IX.1: Industry Respect for Cash Program Director and Staff

The state has good people trying to make the program work fairly for everybody. The person chosen to head this up made a positive difference. He has experience with disability, is open, fair, has a history in personal care and community-based services, and knows what clients need. We knew we could trust him to be honest. He is the least bureaucratic of state people. He sincerely cares. If not for him, we would have hated the program more.

#### 3. Implemented as Planned and Learning from Experience

Initially, Personal Preference was implemented as planned. As New Jersey learned from its experiences, it shifted the responsibility for certain tasks and revised detailed procedures. However, few of these changes involved the program's fundamental design.

**Outreach and Enrollment.** The design for Personal Preference called for New Jersey to enroll only PCA clients, and the state implemented and maintained this approach. Although Medicaid beneficiaries who were eligible for PCA could enter that service with the goal of subsequently enrolling in Personal Preference, New Jersey did not solicit their participation even when enrollment in Personal Preference was lagging. Nor did New Jersey solicit Personal Preference participation from beneficiaries participating in one or more of the state's home- and community-based waivers.

New Jersey did revise procedures for outreach to PCA clients. Initially, the state planned to focus on those who had recently been assessed (or reassessed). Mailings were to go out as consumer data forms were received following assessment. New Jersey departed from this design in that the timing of mailings was divorced from assessment. The state began sending mailings to all PCA clients who had not previously expressed any interest in Personal Preference. At the time of our visit, the state was considering sending them a letter from the governor, with the hope of giving enrollment a boost.

As the program learned from experience, the state also revised enrollment procedures. Attempting to increase the pace of enrollment, but with costs mounting, New Jersey and its first enrollment contractor repeatedly revised the detailed procedures for contacting those who had expressed interest in the cash program. Major changes included shifting from in-person to telephone presentations in selected cases and reorganizing work so that a given staff member handled a case through the entire enrollment process.

Despite these changes, New Jersey was dissatisfied with the work of its independent enrollment contractor and eventually hired state staff members as enrollment specialists. Because the number of in-house enrollment specialists was smaller than the number of contractor enrollment specialists, Personal Preference focused enrollment on the more populous areas of the state. However, the state continued to use many enrollment procedures that it and the enrollment contractor had developed at the start of the cash program.

Cash Allowance, Consulting, and Fiscal Services. The design for Personal Preference called for many consulting agencies, each serving a different area of the state. New Jersey kept this basic structure, but the number of consulting agencies was reduced dramatically (from 34 to 12 at the time of our visit) as some dropped out of the cash program and as New Jersey stopped referring consumers to agencies not performing satisfactorily. Moreover, New Jersey had planned to give consumers a choice of consulting agencies, but as consumers did not seem to value this choice, the state began assigning them to agencies. Moving from choice to assignment of consulting agencies was a fundamental change in the design of Personal Preference.

A legal dispute prolonged the contracting process for fiscal services. However, New Jersey selected a single fiscal agent to serve cash recipients throughout the state, as planned.

New Jersey did not change the fundamental payment structure for consulting and fiscal services. As planned, 10 percent of the cashed-out value of care plans was reserved to cover the costs of consulting and the bulk of the costs of fiscal services. Schedules of fees were developed for fiscal services. The state paid for some fiscal services, and consumers were charged modest fees for fiscal services they used. The

only material change in payment was increases in the rates at which consultants were paid. These increases were partly in response to agency concerns that the payment did not cover the costs agencies incurred in providing consulting services.

New Jersey followed the procedures it had designed for the development and revision of cash management plans. Consultants worked with consumers to develop cash management plans basically as planned. Minor revisions in detailed procedures were adopted over time, as the program learned from experience. Techniques were developed to reduce the need for formal revisions of care plans (for example, by writing more flexible plans) and to make formal revisions more efficient (for example, by putting cash plans onto electronic spreadsheets). As planned, the state Personal Preference staff reviewed all cash management plans.

The design for Personal Preference called for expenditures only in accordance with approved cash management plans. The fiscal agent successfully implemented this aspect of the design as planned, and this principle was never violated.

#### 4. Avoiding Abuse and Exploitation

Critics of Personal Preference were concerned that vulnerable populations would be neglected or exploited or that the cash allowance would be abused. Proponents feared that even a single, widely publicized instance of abuse of the cash allowance or of neglect or exploitation of the consumer could make it politically impossible to adopt an ongoing cash program. As of the time of our visit, however, there had been no abuse of the allowance in New Jersey. State review of all cash management plans ensured that all included goods and services were allowable, and the fiscal agent employed rigorous checks to ensure that it only paid invoices that conformed to the cash plan. During their initial visits, consultants identified a few cases of possible exploitation of consumers, which were referred to the state Personal Preference office for investigation and resolution. No neglect or exploitation of *cash recipients* was detected. Finally, one incident was reported in which a consumer appears to have exploited a worker (by withholding pay to a relative following a family dispute). None of the small number of incidents that occurred resulted in adverse publicity.

#### 5. Achieving Budget Neutrality

Generating savings was *not* one of New Jersey's goals in implementing Personal Preference. However, a cash program that is appreciably more costly than traditional PCA would not be politically viable in that state. Moreover, the terms and conditions of the demonstration waivers required that Personal Preference be budget neutral (that is, cost no more per person per month than the traditional program). A definitive answer as to whether Personal Preference was no more costly to operate than the traditional program

must await the impact analyses based on Medicaid claims. Here, we provide some information relevant to those analyses.

#### a. Cost of Personal Preference

As explained in Chapter II, New Jersey set aside 10 percent of the cashed-out value of every care plan (which averaged about \$1,300 a month at the time of our visit) to cover the cost of consulting and the bulk of the cost of fiscal services. Were these funds (together with those generated by modest user fees charged consumers) adequate to cover the costs incurred for consulting and fiscal services? In spring 2001, it seemed that fees that the fiscal agent earned from consumers and the state were adequate to cover the cost of fiscal services and that the 10 percent set-aside was adequate to fund consulting and fiscal services.

Early in the demonstration, the fiscal agent had serious cash flow problems. Although New Jersey had provided the fiscal agent with start-up funds, it used them for start-up capital costs, primarily the purchase of equipment. The fiscal agent had projected that it could not cover its operational costs until the caseload reached 250. Caseload buildup was slow, and it reached 250 only about a year after the first cash allowance was paid. Moreover, some of the assumptions underlying the projection of 250 as an adequate caseload were overly optimistic. The average monthly cash allowance was slightly lower than assumed (the assumption was \$1,500 monthly), and the average number of workers hired per consumer was also slightly lower than assumed (1.3 instead of 1.5). To cover the initial cash flow problems, the host organization for the fiscal agent provided an internal loan.

Three important factors contributed to holding down the cost of Personal Preference in the demonstration. First, for assessing budget neutrality, two types of costs that would not be a part of an ongoing cash program were excluded. As New Jersey does not plan to have a separate enrollment effort in an ongoing program, the cost of enrollment in the demonstration was covered separately. (A grant from RWJF covered the salaries of the state enrollment staff.) Moreover, in an ongoing cash program, as in the demonstration, New Jersey plans to separate assessment from consulting. Assigning assessment to consultants could create a conflict of interest, because consultants might act as advocates for consumers and increase their care plan hours to raise the amount of their cash allowances. Under Personal Preference, New Jersey's Medicaid case management allowance covered the costs of reassessment visits for cash recipients. PCA agency nurses conducted reassessments for PCA recipients (including control group members), and those costs were included in the hourly rate the state paid for traditional PCA services.

The second factor holding down the cost was that New Jersey's payment structure limited the cost of consulting under Personal Preference. It paid a lump-sum fee to consultants to develop a cash management plan (initially \$53, raised to \$65 at the time of

our visit) followed by an hourly rate for consulting (initially \$18, raised to \$20). Even after this increase, the director of Personal Preference viewed these rates, especially the lump-sum fee for the cash management plan, as a bit low, and he was considering an additional increase when we visited.<sup>48</sup> New Jersey also held down the cost of consulting by setting a maximum number of hours (19) for which a consultant could be paid for a given consumer. At the time of our visit, New Jersey had no plans to increase the maximum number of hours.<sup>49</sup>

#### b. Need for a Discount Rate?

Since the amount of the cash allowance under Personal Preference is based on cashing out care plan hours, the cost of the cash allowance *relative* to the cost of traditional PCA care depends on the extent to which PCA recipients actually receive the care called for in their care plans. As indicated in Chapter II, while planning the demonstration, New Jersey found no appreciable difference between cost of hours of care planned and received for a sample of PCA recipients--that is, the ratio of the two costs was approximately one. Consequently, the state decided not to discount care plan hours before cashing them out.

From the site visit information, we cannot determine whether the *ratio* of cost of hours of care received to hours of care planned changed during the demonstration in such a way that discounting was needed to ensure budget neutrality. However, it is clear that, during the demonstration, hours of care planned and hours of care received were changing for the PCA population. Personal care industry representatives reported that the state had begun (1) to scrutinize beneficiary appropriateness for personal care more intensively, which resulted in denial of services to some long-time recipients; and (2) to reduce hours of care planned. These industry representatives reported that hours of care received had fallen by almost 30 percent in the months before our visit, from 22 to 16 hours a week, on average. In addition, during the demonstration, changes in the labor supply may have affected hours of care received if agencies found it more difficult than before to staff all the hours of care that had been planned for their clients. With care planned and care received in flux, it is possible that the ratio of care planned and care received also changed so as to make a discount rate necessary.

<sup>&</sup>lt;sup>48</sup> Later, the lump-sum payment was raised to \$75 and the hourly rate to \$26.

<sup>&</sup>lt;sup>49</sup> The maximum number of hours was later increased to 20.

#### **B.** Lessons About Procedures and Structure

Some of the lessons learned from Personal Preference pertain to procedures for particular components of the program. Other lessons cut across components and pertain to its basic structure.

#### 1. Component-Specific Lessons

#### a. Outreach and Enrollment

One lesson of Personal Preference is that "taking all comers" is workable. New Jersey's experience suggests that consumers and representatives can screen themselves for appropriateness for a cash program, but it would be difficult (if not impossible) to develop a formal screening process to identify in advance consumers who are unable to manage the cash allowance, either by themselves or with the help of representatives.

Providing for language diversity is important in a cash program, because non-English-speaking consumers may be able to benefit greatly from hiring workers who know their language and are familiar with their culture. Yet mounting a successful outreach and enrollment effort in a diverse, multilingual state like New Jersey is clearly a challenge.

Personal Preference offers some lessons about techniques to meet that challenge. Enrollment specialists for Personal Preference spoke the languages most common in the state. Many program materials were translated into the common languages, and a notice in 14 languages asked consumers to have a family member or friend translate materials in the accompanying package. Family members or friends also translated when enrollment specialists visited or telephoned. When no other translator was available, the specialists used the AT&T language line for telephone calls. Finally, information should also be available in multiple media, as well as in multiple languages, for those with vision and hearing impairments and those who are not competent readers in any language.

Another lesson of Personal Preference is the importance of including family members in enrollment home visits and of maintaining flexibility in scheduling these visits so that they can attend. In general, PCA recipients are more comfortable if family members are present and participate in the decision. Often, family members will be directly involved as workers or representatives. To accommodate the schedules of family members, home enrollment visits should be scheduled outside business hours upon request.

New Jersey's use of two approaches to organizing outreach and enrollment--external contracting and the hiring of internal state staff members--provides an opportunity to assess their advantages and disadvantages in a constant environment. A key advantage of external contracting was that services could be put in place quickly by amending a current state contract. In contrast, state employees were not on the job until many months

after approval to hire them was sought (as the director of Personal Preference had expected). The key advantage of having state employees conduct outreach and enrollment was that the state Personal Preference staff had more control of enrollment procedures. Neither the external contractor nor state employees, however, could sustain a pace that met the evaluation target for enrollment. The basic organizational structure of outreach and enrollment in New Jersey does not seem to explain the failure to meet sample-size targets for the evaluation.

Revisions to procedures might have increased enrollment in Personal Preference (although we cannot be sure either of this or of the magnitude of any increase). New Jersey did not mail letters from the governor or other high-ranking officials to PCA recipients as part of its marketing campaign--a technique that Cash and Counseling programs in Arkansas (and later Florida) used with considerable success. In addition, for part of the demonstration intake period, outreach and enrollment was not provided across the entire state of New Jersey. However, the areas with the greatest concentrations of newly assessed PCA recipients were always covered, and the number of enrollees forgone was probably small.

New Jersey's experience in Personal Preference confirmed that the cash program was attractive to a sizable minority of PCA recipients. About 1,750 people had enrolled in the demonstration by the time enrollment was closed in New Jersey in June 2002. They represented about 14 percent of the number of PCA recipients in the year before the demonstration began (about 12,000). Just over half the 1,750 enrollees (53 percent) were elderly, which proves that a cash program is attractive to elderly consumers in New Jersey. However, relative to the size of the populations of elderly and nonelderly PCA recipients, elderly beneficiaries were less likely than nonelderly ones to enroll. The elderly enrollees represented just over 10 percent of the estimated number of elderly PCA recipients the year the demonstration began, while the nonelderly enrollees represented over 20 percent of the estimated number of nonelderly PCA recipients that year.

#### b. Consulting and Cash Management Planning

Planning is essential to making the initial consulting home visit as useful as possible. Consumers must begin to think about how they would like to spend the cash allowance, whom they would like to hire, and whether they want to name a representative to help them. It is important that potential representatives attend the initial consulting home visit and that potential workers attend if they have been identified. Moreover, it is helpful for all family members involved in the care of the consumer to be present. With good planning, the need for multiple home visits can be minimized, thereby reducing costs and eliminating a source of delay in the receipt of the first cash payment.

Revisions to a cash management plan are often necessary, and they consume substantial staff resources. To reduce the number of circumstances in which a formal cash

plan revision was required, Personal Preference wrote flexible cash plans that did not name workers or vendors. The program also developed useful techniques to streamline the revision of cash plans. One was entry of the original cash plan into an electronic spreadsheet. Another was using tickler files to remind the consultant to begin to plan for expected changes, such as a revision of the cash management plan when most of the funds needed for a planned purchase had been saved.

An important lesson learned from Personal Preference is that the time required for each step in cash planning adds up and can appreciably delay the receipt of the cash allowance. Personal Preference eliminated one step and reduced the time required for another, thereby reducing delay. It eliminated consumer choice of consulting agencies and reduced notice time to personal care agencies serving consumers who were about to begin to receive the cash allowance.

Communication problems can also extend the time to receipt of cash. (In Section 2.b below, we discuss the factors leading to communication problems in Personal Preference.) Lessons from Personal Preference about ways to improve communication include training of consumers about whom to call regarding what issues, exception reports from the fiscal agent indicating the nature of the error requiring resolution, and multiple-party telephone calls (for example, between the consultant, fiscal agent, and consumer) to reduce the need for several one-on-one calls.

#### c. Representatives

The representative selection process is an extension of the relationships that consumers already have in place and the assistance that they already receive. Thus, almost all representatives are family members or friends. Consumers who are already receiving assistance with their affairs generally were pleased to continue to have assistance. Representatives often were pleased to serve in that capacity to make it possible for the consumer to participate in the cash program.

The need for a representative depends on the familiarity of the consumer (and of the consumer's family) with the cash program, the availability of consumer training from consultants, and (rarely) the consumer's environment (for example, the availability of a talking computer and a scanner for a blind consumer)--as well as the consumer's level of disability.

The role of the representative varies. For those people who are completely unable to communicate their preferences, the representative obviously must be the sole decision maker. Typically, however, the consumer and representative both contribute to the management of the cash allowance, with the nature and extent of their contributions varying. Representatives may solicit consumer preferences, or consumers and representatives may function as a team. The consumer may even be the primary decision

maker for the team, while the representative serves as the consumer's liaison with the outside world.

Perhaps the major lesson of Personal Preference with respect to representatives is that consultants judged that representatives were obtaining input from the consumers when possible and were faithful to the best interests of the consumers.

#### d. Consumers as Employers

A key lesson of Personal Preference is that consumers unanimously chose to use the fiscal agent. In doing so, they split the fiscal responsibilities of an employer between themselves and the fiscal agent. The consumer was responsible for timesheets, and the fiscal agent was responsible for the employer's other major fiscal tasks (including filing state and federal payroll tax returns and cutting paychecks).

Personal Preference teaches that consumers can satisfactorily fulfill the nonfiscal responsibilities of employers--recruiting (which proved the most difficult), hiring, training, supervising, and firing. Another lesson is that consultant training played a major role in this success.

The amount of consultant assistance that consumers needed varied considerably. The experience of Personal Preference is that even consumers who need a substantial amount of consultant assistance can learn in time to fulfill successfully the responsibilities of an employer, provided they can identify a worker to hire.

Recruiting was critical for those consumers who did not have family or friends whom they wished to hire as workers. The receipt of cash was inevitably delayed for consumers who had difficulty recruiting a worker. Program staff reported that consumers who could not recruit a worker tended to drop out of the cash program.

Personal Preference developed written materials to train consultants about recruiting, and the consultants shared materials with consumers. However, some consumers, particularly those without family or friends available to be workers, required more help with recruiting. Moreover, consultants were uncertain about how much assistance they were to provide consumers with recruiting, which suggests that both consultants and consumers would benefit from more consultant training on recruiting. In addition, New Jersey is developing a worker registry (outside Personal Preference) to help such consumers. Finally, experience in New Jersey indicates that consumers who have difficulty recruiting workers might use agency services, at least until a worker can be identified, and that less costly agency services may be available. For example, one agency was willing to negotiate a special rate for a Personal Preference consumer.

Nearly all consumers submitted timesheets on schedule. Consultants trained consumers on how to fill out timesheets, and the fiscal agent reminded consumers to submit them on time.

#### e. Preventing Abuse of the Cash Allowance and Exploitation of Consumers

From the experience of Personal Preference, we draw the lesson that abuse of a cash allowance--even an allowance of a substantial amount--can be prevented. This can be done by careful development of a cash plan, followed by rigorous procedures to ensure payment for only those goods and services allowed under the plan. Review of receipts was not required under Personal Preference.

Another lesson is that cases of neglect or exploitation of consumers can be eliminated before cash payment is made. This can be done if program staff members keep the possibility of such cases in mind and look into any in which something seems amiss. No cases of possible neglect or exploitation were identified *after* consumers began to receive the cash payments.

More attention to preventing exploitation of workers may be warranted. One such case was reported to a consultant. Consultants were trained not to become involved in employer-worker relationships, and Personal Preference had no mechanism for workers to voice grievances about their employment situations.

#### 2. How Should a Cash Program Be Structured?

The New Jersey experience provides two important lessons about structuring a Cash and Counseling program. These lessons pertain to (1) the number of consulting agencies, and (2) the effect of program structure on communication and on delay of receipt of the cash allowance.

#### a. Number of Consulting Agencies

New Jersey, mindful of the cultural diversity of its Medicaid PAS caseload, initially recruited 34 consulting agencies, to give consumers a choice. However, having a program with many agencies proved to be a problem. In some areas of the state, multiple agencies were never recruited. Other consulting agencies performed poorly, and Personal Preference dropped them from its roster. The program did not attempt to replace them, in part because it had learned that consumers in New Jersey did not consider having a choice of consulting agency important. Moreover, consumer choice of agency contributed to delays in the receipt of the first cash payment. By the time of our visit, 12 agencies were providing Personal Preference consulting, and the state was pleased with their performance in general.

We draw the lesson that having a large number of consultant agencies and allowing consumers their choice entails costs and may yield little or no gain.

As the number of consulting agencies increases, the average caseload size per agency falls (for a given statewide caseload). Some agencies will likely have caseloads that are too small to support efficient provision of consulting services. With small caseloads, consultants will learn from experience slowly, as they will seldom be assigned new cases. Moreover, they will grow "rusty," since they will not constantly be performing consultant duties. When consultants leave agencies, their replacements will start at the bottom of the learning curve. Furthermore, the caseloads at some agencies will support only one consultant, and backup is necessary when the consultant is not in the office. Moreover, providing peer support and supervision is more difficult with a lone consultant.

Having a large number of consulting agencies also consumes state staff resources. State program staff must recruit the consulting agencies and perhaps replace dropout agencies. As the number of agencies increases and the average agency caseload falls, the likelihood of an agency dropping out increases because it is more likely to view the cash program as a sideline. If a large number of consulting agencies is to be maintained, state staff must devote effort to recruiting replacement agencies. In addition, state staff resources are required to ensure the quality of consulting services. The more agencies, the more state effort required for quality assurance.

# b. Effect of Program Structure on Communication and Delay in Receipt of Cash Allowance

Both state Personal Preference staff and consultants reported frequent communication problems among consumers (or representatives), consultants, state program staff, and fiscal agent staff. A consumer, consultant, member of the state program staff, or member of the fiscal agent staff might have a question or comment and direct it to any of the other parties. That is, there were 12 one-way communication pathways in the New Jersey cash program: (1) consumer to consultant, (2) consumer to state program staff, (3) consumer to fiscal staff, (4) consultant to state program staff, (5) consultant to fiscal staff, (6) state program staff to fiscal staff, (7) fiscal staff to state program staff to consultant, (9) fiscal staff to consumer, (10) state program staff to consultant, (11) state program staff to consumer, and (12) consultant to consumer. When the consumer had a representative, the number of one-way communication pathways increased to 20. With this number of communication pathways, it is not surprising that communication problems arose in Personal Preference and that these problems were sometimes serious enough to delay the receipt of the cash allowance.

An example is the completion of the cash management plan. Several people participated in the development and review of every cash management plan, including every revision. The consumer (perhaps with the help of a representative) and the

consultant developed the plan. When the plan had been completed, it was sent to a member of the state Personal Preference staff to review the included goods and services. If the cash plan was approved, the Personal Preference office sent it to the fiscal agent, where the calculations were double-checked for accuracy before the consumer's cash account was established (or revised). If problems were identified in either state program office or fiscal agent review, the paperwork was returned to the consumer. If the consumer did not understand the problem and how to resolve it, he or she would have to contact the consultant or fiscal agent.

The best way to resolve communication problems and speed the receipt of the cash allowance may be to reduce the number of people involved day to day. Arkansas's experience suggests two approaches to reducing the number of people (see Phillips and Schneider 2002). The first is to combine consulting and many fiscal services in the same person. Under the Arkansas cash program, the same organization (in a geographic area) provides consulting and fiscal services, and the same *individuals* provide consulting and perform key fiscal tasks. The second approach is to limit the role of state staff in reviewing cash management plans. Arkansas state staff reviewed only those cash plans that called for items not on a list of allowable goods and services. Under the demonstration, New Jersey gained experience on which to base a list of allowable goods and services.

## C. Value of Counseling and Fiscal Services

The Cash and Counseling model has a fundamental tenet that distinguishes it from other models of consumer direction: the provision of services to help consumers manage their cash allowances. Some critics of the Cash and Counseling model argue that an unfettered cash allowance would be preferable, because such an allowance is more consistent with the philosophy of consumer direction than a program that imposes restrictions on, and monitors, the uses of the cash allowance. States, however, must balance this argument with the concern that Medicaid funds might be misused, which could jeopardize political support for the program.

What can we learn from the New Jersey experience about the burden imposed by restrictions on the uses of cash and about the value of counseling and fiscal services to consumers?

**Burden of Restrictions on Uses of Cash.** The New Jersey experience suggests that most consumers who participated did not find the restrictions on the use of cash

<sup>&</sup>lt;sup>50</sup> Arkansas uses the term "counselor" instead of "consultant." For clarity, we retain the New Jersey terminology here.

troublesome in their everyday lives. Personal Preference insisted that the allowance be used for purchases related to the consumer's need for personal assistance, and most consumers wanted to purchase "standard items." Nonetheless, New Jersey tried to foster creative uses of the allowance and interpreted the allowance broadly.

Value of Fiscal and Consulting Services. Clearly, consumers in Personal Preference valued the services of the fiscal agent. Although charged modest fees from the cash allowance for the use of fiscal services, all consumers chose to use the fiscal agent.

The value of consulting services is more difficult to assess than the value of fiscal services. Consumers varied greatly in the amount of advice and assistance they needed from consultants. A few consumers needed little or none. Many needed oral explanations of the program and its procedures, in addition to written materials. Still others needed a great deal of assistance as they developed a cash management plan and recruited workers--sometimes more than the 19 hours annually for which consultants could be paid under Personal Preference. For nearly all consumers, consultant advice seems to have been a valuable service. Overall, consulting seems to have been valuable--perhaps essential--to the success of the cash program in New Jersey.

# X. TOWARD AN ONGOING CASH PROGRAM IN NEW JERSEY

Based on its experience in the Cash and Counseling Demonstration, New Jersey was working at the time of our visit in spring 2001 to make many of the features of Personal Preference available under its Medicaid state plan. The state had already expanded the consumer-directed programs offered to Medicaid beneficiaries and was reviewing how cash allowance programs might respond to the need to provide care in the least restrictive setting possible. Later, New Jersey decided to seek another demonstration waiver under Section 1115.

## A. Broader Personal Care Allowance

In spring 2001, New Jersey was seeking to expand its Medicaid state plan to add a consumer-directed form of Personal Care Assistance (PCA) as a permanent program. Under a broader personal assistance benefit, consumers could choose their own workers, but the program could not have all the features implemented under the demonstration waiver. Given its experience in the demonstration, New Jersey has no difficulty in giving up two features not currently possible without waivers: (1) having legally liable relatives hired as workers, and (2) direct payment of the cash to the consumer. Under federal Medicaid regulations, spouses of adults are legally liable for providing personal assistance and cannot be paid to provide that assistance (unless that regulation is waived). New Jersey is willing to forgo hiring spouses, because most of the consumers participating in the demonstration were not married. According to federal regulations, cash accumulated by consumers must be included in means tests for Medicaid eligibility and for other public programs (unless the relevant provisions are waived). New Jersey is willing to forgo direct payments in an ongoing program, because all consumers participating in the demonstration elected to use the fiscal agent. However, the state believes it is important to allow consumers to have a small proportion of the allowance available for incidental expenses (such as taxi fare) and is interested in ways to accommodate that need.

The current Medicaid PCA regulations refer to "human assistance."<sup>51</sup> New Jersey would like to see the definition of assistance expanded to include goods and services, authorized in a plan of care that would meet a consumer's needs for personal assistance services. The state is particularly interested in being able to offer equipment, and to a

<sup>&</sup>lt;sup>51</sup> Shortly after our visit to New Jersey, CMS indicated that a change in federal statutes probably was not required to broaden the definition for personal assistance. Rather, changes in regulations, including changes in the Medicaid Manual, would be sufficient.

lesser extent, home modifications, in a consumer-directed form of PCA under the state plan.

After our visit, CMS responded to New Jersey's request for consideration of an expansion of the personal care benefit, indicating a need for study of the cost of covering equipment and, possibly, modifications. CMS was not prepared to expand the personal care benefit pending the results of such a study.

#### **B.** Continuation of Personal Preference

In 2002, CMS made available the IndependencePlus templates for waiver applications from states interested in consumer-directed care. Using the IndependencePlus templates, New Jersey began to prepare to submit an application for a waiver under Section 1115 for an ongoing Personal Preference program.

As of early 2003, New Jersey was planning two major changes in the design of Personal Preference as part of its proposal for a new waiver program. First, the consultants would be allowed to approve cash management plans that contained only items on a prespecified list developed by the state based on its experience in the demonstration. Some of the items on the list would be conditional on the consumer's disability. For example, blenders and food processors would be listed only for consumers who had difficulty with eating. Thus, state-level approval of cash management plans would be required only for unusual requests. The second major change that New Jersey is planning is to combine all consulting and fiscal services in a single organization, gradually phasing out the work of the agencies that offer consulting alone. One of the state's principal reasons for combining consulting and fiscal services is to reduce the need for coordination and communication, thereby reducing time to the receipt of cash. Another is to increase consultant caseloads and thereby increase consultant efficiency.

# C. Extending Consumer Direction to Other Populations

New Jersey has long valued public programs that foster personal responsibility. Consumer-directed programs are consistent with that value. Before the Cash and Counseling Demonstration, New Jersey offered the Personal Assistance Services Program (PASP). As discussed in Chapter II, PASP is a state-funded program for adults with physical disabilities who are able to direct their own care and who are working, looking for work, or in school. PASP participants recruit, hire, supervise, and fire their own personal assistance workers.

The director of the Division of Medical Assistance and Health Services--the division of the New Jersey Department of Human Services that housed Personal Preference--felt

that a cash allowance had wide applicability. Throughout the Cash and Counseling Demonstration, she promoted the cash model and advocated its adoption within other divisions of the department. For example, she asked the director of Personal Preference to make presentations to executives heading other divisions of the Department of Human Services.

The successful implementation of Personal Preference--particularly the high levels of consumer satisfaction with the program and the absence of abuse or exploitation--reinforced the validity of consumer-directed programs and contributed to the adoption of other cash allowance programs in New Jersey. In spring 2001, two other divisions had already implemented cash programs. The Department of Health and Senior Services had initiated a consumer-directed waiver program for elderly beneficiaries--called Enhanced Community and Homecare Options (ECHO)--and the Division of Developmental Disabilities had initiated a self-determination project for beneficiaries it serves.

The director of Personal Preference had been working with the executives heading these other programs to identify state statutory and regulatory impediments to expansion of cash programs in New Jersey. For example, they determined that state legislation was required to permit New Jersey to adopt federal fair labor standards, including exemptions for live-in and companion services--thereby giving New Jersey consumers greater flexibility in setting hours and wages for employees hired under cash programs.

Finally, New Jersey's interest in cash programs has been heightened as the state seeks to meet the mandate of the U.S. Supreme Court decision in *Olmstead v. L.C.*, which requires that states provide care for Medicaid beneficiaries in the least-restrictive setting reasonable. The state's Olmstead Working Group examined how the cash model might be used in providing long-term care in the less-restrictive settings. After our visit, New Jersey applied for and received a "Systems Change" grant from CMS. A key aspect of this grant is to develop worker registries to assist consumers in recruiting workers.

## REFERENCES

- Cameron, Kathleen. "Cash and Counseling: A Model for Empowerment and Choice for Long-Term Care." *Perspectives on Aging*, July-September 1995, pp. 22-25.
- Cameron, Kathleen, and Robert Lagoyda. "Cash and Counseling Technical Analysis #1: The Counseling Component, Draft." Washington, DC: Center on Consumer-Directed Services at the National Council on Aging and University of Maryland, Center on Aging, February 1997.
- Doty, Pamela, J. Kasper, and S. Litvak. "Consumer-Directed Models of Personal Care: Lessons from Medicaid." *Milbank Quarterly*, vol. 74, no. 3, pp. 377-409, 1996.
- Eustis, Nancy. "Consumer-Directed Long-Term Care Services: Evolving Perspectives and Alliances." *Generations*, vol. 20, no. 3, fall 2000, pp. 10-15.
- Flanagan, Susan. "Payment, Employment-Related Tax, Legal Liability, and Quality Assurance Issues Related to In-Home Care Programs That Use Consumer-Directed Care Attendants." Cambridge, MA: SysteMetrics, March 1994.
- Jackson, Mary. "Rationing Case Management: Six Case Studies." Cambridge, MA: MEDSTAT Group, November 1994.
- Kane, Rosalie. "Let's Just Send Cash for Long-Term Care: Conceptual Underpinnings for a Cash and Counseling Demonstration." Paper presented at the Annual Meeting of the Cash and Counseling Demonstration, Washington, DC, October 1996.
- Kitchener, Martin, and Charlene Harrington. "Medicaid 1915(c) Home and Community Based Waivers: Program Data 1991-1999." San Francisco, CA: University of California, San Francisco, August 2001.
- LeBlanc, Allen, Christine Tonner, and Charlene Harrington. "State Medicaid Programs Offering Personal Care Services." *Health Care Financing Review,* vol. 22, no. 4, summer 2001, pp. 155-173.
- Mahoney, Kevin J., Kristin Simone, and Lori Simon-Rusinowitz. "Early Lessons from the Cash and Counseling Demonstration and Evaluation." *Generations*, vol. 24, no. 3, fall 2000, pp. 41-46.

- Phillips, Barbara, and Barbara Schneider. "Moving to IndependentChoices: The Implementation of the Cash and Counseling Demonstration in Arkansas." Princeton, NJ: Mathematica Policy Research, Inc., May 2002.
- Simon-Rusinowitz, L., and B. Holland. "Adopting a Disability Approach to Home Care Services for Older Adults." *Gerontologist*, vol. 33, no. 2, 1993, pp. 159-167.
- Simon-Rusinowitz, Lori, Kevin J. Mahoney, Sharon M. Desmond, Dawn M. Shoop, Marie R. Squillace, and Robert A. Fay. "Consumer Preferences for a Cash Option Versus Traditional Services in New Jersey; Telephone Survey Technical Report." Baltimore, MD: Center on Aging, University of Maryland, 1998.
- Velgouse, Linda, and Virginia Dize. "A Review of State Initiatives in Consumer-Directed Long-Term Care." *Generations*, vol. 24, no.3, fall 2000, pp. 28-33.

# APPENDIX A. PERSONAL PREFERENCE MATERIALS



#### NEW JERSEY DEPARTMENT OF HUMAN SERVICES State Office on Disability Services Personal Preference Program

#### PARTICIPANT STATEMENT OF RIGHTS AND RESPONSIBILITIES

#### **RIGHTS**

I have the right to create a Cash Management Plan to meet my needs within the Personal Preference Program guidelines for use of the cash grant.

I have the right to change my Cash Management Plan to meet my needs within the program guidelines for use of the cash grant.

I have the right to be treated with dignity and respect.

I have the right to privacy and confidentiality.

I have the right to live as I choose, in my own home, free from judgment or interference.

I have the right to decide about how to spend my cash grant or to have someone I choose help me with decisions about the Personal Preference Program.

I have the right to bring whomever I wish to all meetings pertaining to the Personal Preference Program.

I have the right to file a complaint with the Personal Preference Program State staff at 1-888-286-3035 (**Toll Free**) for any reason, including being advised to disenroll.

#### RESPONSIBILITIES

I must notify my consultant within five days of admission to a hospital, nursing facility, rehabilitation facility, or any other institution. I understand that I am not entitled to my cash grant during the time I spend in a facility.

I must keep scheduled appointments.

I am responsible for hiring, supervising and firing my employees and all the responsibilities that go with hiring employees.

I must treat my employees, the consultant, and others who work with the Personal Preference Program the same way I expect to be treated.

I am responsible for what is included in my Cash Management Plan and for managing my cash grant accordingly.

I am responsible for all required paperwork and adhering to all tax and labor laws.

I am responsible for answering interview questions from Mathematica Policy Research.

I have read and /or understand these rights and responsibilities.

Participant/Representative signature	Date	
Consultant signature PPR&R(12/98:6/99:7/99)	Date	

# RESPONSIBILITIES OF REPRESENTATIVE

# New Jersey Department of Human Services Division of Disability Services Personal Preference Program

(New Jersey Cash & Counseling Demonstration)

# Representative Description

A representative may be a participant's legal guardian, a family member, or any other

individual identified who willingly accepts responsibility for performing cash management tasks that the participant is unable to perform. A representative must evidence a personal commitment to the participant and must be willing to follow their wishes and respect their preferences while using sound judgment to act on their behalf. Representatives receive no monetary compensation for this service, and may not serve as an employee of the participant.

# Specifically, the representative must be willing to:

- Work with the Cash & Counseling consultant to provide information to develop the cash management plan on the participant's behalf.
- Use the cash grant for the items outlined in the Cash Management Plan as the participant wishes.
- Maintain records, as required by the State, regarding expenditures and activity with the fiscal intermediary.

# Representatives may be necessary for participants under certain conditions as defined below:

# Voluntary Representative

The participant requests that a representative serve on their behalf, or a consultant recommends that the participant choose a representative and the participant agrees.

# **Predetermined Representative**

The participant has a legal guardian or other court appointed representative in place at the time of enrollment and that individual will serve as the designated representative on the client's behalf.

# Mandated Representative

The client is enrolled in Personal Preferences and has misspent funds from the cash allowance, or their functioning has deteriorated in such a way that they are no longer able to manage their cash benefit.

repres.doc 2/98:12/98:8/24/99:8/25/99:7/1/01

# COLORED FLYER MAILED TO PCA RECIPIENTS



# State of New Jersey

DEPARTMENT OF HUMAN SERVICES STATE OFFICE ON DISABILITY SERVICES PO BOX 700 TRENTON NI 08625-0700

MICHELE K. GUHL

Commissioner

WILLIAM A. B. DITTI Executive Director TEL (609) 292-7800

CHRISTINE TODD WHITMAN

Do You Want Greater Independence,
Personal Choice and Flexibility
When Using Personal Care Services?
Then Don't Miss This Exciting Opportunity!!!

September 2000

Dear Medicaid Personal Care Consumer:

New Jersey Medicaid is offering an exciting new program for people who need personal care assistance, called Personal Preference: New Jersey's Cash and Counseling Demonstration Program. This is a research program, Personal Preference will study people who buy their own personal care services and those who get services from an agency. One half of the people who agree to participate will receive a cash grant, while the other half will continue to get services the way they do now.

# Personal Preference Allows You To Have More Control Over Your Care.

- ξ You Make The Decisions!
- E You Decide What Services You Want & Need!
- $\xi$  You Can Hire Anyone You Want; Relatives, Friends, Neighbors!
- $\xi$  You Can Purchase Equipment, Devices and Make Home Modifications!
- ξ You Receive Money Every Month To Buy Your Own Personal Care Services!

IF YOU ARE INTERESTED, PLEASE CALL TODAY (TOLL FREE)

1-888-285-3036. THE CALL IS FREE, AND PERSONAL

PREFERENCE STAFF WILL BE HAPPY TO SPEAK WITH YOU

FURTHER.

Estado de Rew Tersep

DEPARTAMENTO DE SERVICIOS HUMANOS

OFICINA ESTATAL DE SERVICIOS A DISCAPACITADOS

P.O. Box 700

Trenton, NJ 08685-0700

CHRISTINE TODD WHITMAN

Gobernadora

MICHELE K. GUHL Comisionada

¿Desea usted mayor independencia, Director Ejecutivo TEL. (609) 292-7800 Selecciones personalizadas y flexibilidad al usar los servicios de cuidado personal? ¡Entonces no se pierda esta gran oportunidad!

Septiembre de 2000

Estimado cliente de servicios de cuidado personal de Medicaid:

Medicaid de New Jersey está ofreciendo un nuevo programa para personas que necesitan servicios de asistencia para el cuidado personal llamado Preferencia Personal: Programa de Demostración de Asesoría y dinero en efectivo de New Jersey. Éste es un programa de investigación que hará un estudio sobre personas que

medio de una agencia. Una mitad de las personas que acepte participar recibirá una subvención en efectivo, mientras que la otra mitad continuará obteniendo sus servicios de la misma manera en que lo hacen en la actualidad.

adquieren sus propios servicios de cuidado personal y aquéllos que los adquieren por

El programa Preferencia Personal le permite tener mayor control sobre su cuidado personal.

- IUsted toma las decisiones!
- iUsted decide cuáles son los servicios que necesita y deseal
- · iUsted puede emplear a quien quiera: familiares, vecinos, amigos!
- iUsted puede adquirir equipo y dispositivos y hacerle modificaciones a su casa!
- iUsted recibe dinero cada mes para adquirir sus propios servicios de cuidado personal!

SI ESTÁ INTERESADO EN EL PROGRAMA, POR FAVOR LLAME HOY MISMO AL 1-888-285-3036. LA LLAMADA ES GRATUITA Y LOS REPRESENTANTES DE PREFERENCIA PERSONAL CON GUSTO HABLARÁN CON USTED.

# NOTICE IN MULTIPLE LANGUAGES

State of New Jersey **Department of Human Services** State Office on Disability Services Personal Preference Program

# Important Notice

Please have this information translated immediately.

Importante! Fate tradurre quest'informazione immediatamente.

[Haitian Creole] Trè enpôtan! Fè tradui sa a kounye an.

[Italian]

Important! Faites traduire cette [French] information immédiatement.

Importante! Mande traduzir esta [Portuguese]

informação imediatamente.

Pilne! Prosze o niezwłoczne [Polish] przetłumaczenie tej informacji.

重要!請立刻將此信息翻譯成中文。 [Chinese]

Προσοχή! Δώστε να σας μεταφράσουν [Greek] αμέσως αυτές τις πληροφορίες.

QUAN TRONG! CÂN DỊCH [Vietnamese] TIN TỰC NÀY NGAY.

ពិតមានបន្ថែម! [Cambodian] សូមបកប្រែនៅការប្រកាសនេះជាបន្ទាន់

[Laotian] י כר ברד מב

ຈຳຮັບແປການປະກາດນີ້ໂດຽດວັນ

Կը խնդրուի այս տեղեկութիւնը

[Armenian]

անմիջապէս թարգմանել տալ։

Внимание! Немедленно обеспечьте перевод настоящей

[Russian] информации!

:Importante! Por favor de traducir esta [Spanish] información inmediatamente.

# CASH MANAGEMENT PLAN FORMS

# AND RELATED MATERIALS

# STATE OFFICE ON DISABILITY SERVICES PERSONAL PREFERENCE PROGRAM Cash Management Plan

Consumer Name:				Cash Grant Amount:	¥.	
Representative Name:				Medicaid #:		
Type of Plan: (check one)	Initial Revision	Reassessment		Start Date:		
. Direct Employment Service Type/Description	Worker	Hourly Wage	Total Taxes Per Hour	Sum of Hourly Wages & Taxes	# of Hours Per Month	Total Monthly Cost
			Total Monthi	Total Monthly Employment Costs	S	S
I. Purchase of Agency Services				Number of		Total
Service Type/Description	Agency Name	Frequency	Unit Cost	Units Per Month	•	Monthly Cost
			Total Monthi	Total Monthly Agency Services Costs	Costs	\$
II. Miscellaneous Expenses						Total
Expense Type/Description	Provider Description		Frequency	Unit Cost	_	Monthly Cost
Cash						
Home / Workers Comp Insurance						
			Total Month	Total Monthly Miscellaneous Costs	sts	S

V. Special Purchases/Modifications	øι	Proposed	Estimated	Estimated # of	Total
Description of Work/Purchase	Contractor/Provider Name	Purchase Date	Cost	Monthly Payments	Monthly Cost
		Total Monthly	y Special Purch	Total Monthly Special Purchase/Modification Costs	8
V. Fiscal Intermediary Services & Fees	ees				Total
Description of Services and Fees	# of Units	Unit Cost			Monthly Cost
Transaction Fees					
Criminal Background Investigation Fee	Fee				
Employee Health Insurance					

Medicaid #:

Consumer Name:

Consumer Name:			Medicaid #:
Representative Name:			
RECONCILIATION OF MONTHLY CASH BENEFIT	П	Monthly Amount	
A. Total Monthly Cash Benefit			
B. LESS Cost of Direct Employment	(Section I)	\$	
C. LESS Cost of Agency Services	(Section II)	\$	
D. LESS Cost of Other Expenses	(Section III, IV)	\$	
E. LESS Cost of Fiscal Intermediary Services	(Section V)	\$	
(A minus the sum of B, C, D & E) MON1	MONTHLY BALANCE		
Decision Tree Completed: Yes No			
CMP Designed By:			
Consumer Signature:			Date:
Representative Signature:			Date:
Consultant Review:			Date:
(Signature and Title)			
Agency Name:			Phone#
State Drogram Office Approvel:			Doto.
PPcmp 7/99:8/99 (Signature and Title)			Date.
	Page 3 of 3		

### STATE OFFICE ON DISABILITY SERVICES

### Personal Preference Program

## Instructions for Completing the Cash Management Plan

Please read through the directions carefully before entering information. If you have questions concerning the completion of the Plan please contact your Consultant. If your Consultant is not available, call the State Office On Disability Services at 1-888-285-3036 (toll free) for assistance.

Please complete the identifying information at the top of the first page of the Cash Management Plan as well as on pages two and three. Please check whether the plan is the Initial Plan, Revision, or was triggered by a Nursing Reassessment that resulted in a change.

SECTION I: DIRECT EMPLOYMENT

<u>Service Type/Description:</u> List all services you would like to receive through individual workers who will be directly hired and supervised by <u>you</u>, as the consumer.

<u>Worker</u>: List the name of the individual worker or workers you would like to employ to provide the services you need in this section. <u>You should also list any worker that you would like to use as a back up so that they may be able to receive a pay check for future services provided. Indicate in this section whether the worker(s) listed will be a regular employee or a back up. *Please note*: the back-up worker wages and taxes are not to be counted in towards your total monthly costs.</u>

Hourly Wage: List the hourly wage that you intend to pay each worker listed above.

<u>Total Taxes Per Hour</u>: Indicate the total hourly tax expense for each wage listed above. You may obtain this information from Community Access Unlimited (CAU), the Fiscal Intermediary by calling 1-877-354-9944.

<u>Sum of Hourly Wages/Taxes</u>: List the sum of the hourly wage and hourly tax expense for each worker listed.

Number of Hours Per Month: Multiply the number of hours per month you intend to hire each worker by 4.33.

<u>Total Monthly Cost</u>: Multiply the hourly wage and taxes by the number of hours per month to obtain the total cost per month for each worker you hire.

<u>Total Monthly Employment Costs</u>: Indicate the sum of the total cost per month for all workers hired.

Service Type / Description: List all services you would like to receive through an agency. You should also indicate any agency that you would like to consider using as a back up. Please note: the back-up agency costs are not to be counted in towards your total monthly costs.

Agency Name: List the name of the agency you would like to use to provide the service listed.

PURCHASE OF AGENCY SERVICES

SECTION II:

stated.

expenses listed above.

Frequency: Indicate the frequency in which each service is purchased every month.

Unit Cost: List the unit cost you intend to pay for each service listed.

Number of Units Per Month: List the number of units of service per month you intend to purchase. For example, housekeeping service twice per month would be considered 2 units per month.

Total Monthly Cost: Multiply the number of units per month by the unit cost for each agency you plan to use.

Total Monthly Agency Services Costs: Indicate the sum of the total cost per month for all agencies listed, with the exception of backup agencies as previously

SECTION III: MISCELLANEOUS EXPENSES

Expense Type / Description: List all other expenses you plan to have each month. Please note: you will need to include homeowner or renter's insurance or a rider to your insurance as regular expense to guarantee that your workers are covered

for worker's compensation.

On the line that says "Cash" write the amount you will need in cash each month to buy things, such as supplies not covered under Medicaid or taxi fare.

<u>Provider Description</u>: Indicate the intended provider for each expense listed above. Describe any item or service you will purchase next to the cash line but under the provider description category.

Frequency: Indicate the frequency each expense item will occur every month.

Unit Cost: List the unit cost you intend to pay for each miscellaneous expense.

<u>Unit Cost</u>: List the unit cost you intend to pay for each miscellaneous expense.

<u>Total Monthly Cost</u>: Indicate the total monthly cost for each individual expense.

Total Monthly Miscellaneous Costs: Indicate the sum of the total cost for all

<u>Description of Work / Purchase</u>: List all one-time purchases, modifications or services you intend to buy through your Cash Management Plan. This includes all contractual work (modifications) you plan for your home.

<u>Contractor / Provider Name</u>: Indicate the name of the contractor or provider you intend to use for the work/purchase indicated.

SPECIAL PURCHASES / MODIFICATIONS

SECTION IV:

Intermediary Expenses.

Proposed Purchase Date: Indicate the month and year you expect to make each purchase listed.

Estimated Cost: Indicate the approximate total cost of each purchase listed

Estimated # of Monthly Payments: Indicate how many months it will require to pay for each purchase or modification listed.

Total Monthly Cost: Indicate the total monthly cost for each purchase or

modification listed. (This figure should represent the estimated cost divided by the estimated # of monthly payments). This number indicates the amount of the monthly payment(s).

Total Monthly Special Purchase/Modification Costs: Indicate the sum of the total

monthly cost for all expenses listed above under this section.

SECTION V: FISCAL INTERMEDIARY SERVICES & FEES

Description of Services and Fees: Indicate the type(s) of service(s) provided by

the Fiscal Intermediary Service Organization (for instance, background

# of Units: Indicate the number of units to be charged to your grant for each type of fee that is used.

<u>Unit Cost</u>: Indicate the unit cost for each fee type that is used. This information can be obtained by contacting the Fiscal Intermediary Organization.

Total Monthly Cost: Indicate the total monthly expense for each fee type by multiplying the number of units times the unit cost.

multiplying the number of units times the unit cost.

Total Monthly Fiscal Intermediary Costs: Indicate the sum of the total cost for all expenses listed above by adding them together to obtain the Total Monthly Fiscal

# RECONCILIATION OF MONTHLY CASH BENEFIT A. Total Monthly Cash Benefit: Indicate the amount of your total monthly

B.

D.

the workers that you are employing directly.
 C. Agency Purchase of Service Expenses: Indicate the total amount of expenses for services that you are purchasing through agencies.

Direct Employment Expenses: Indicate the total amount of expenses for

Miscellaneous Expenses, Special Purchases/Modifications: Indicate the

budget as identified on page one of your Cash Management Plan.

- budgeted amount of the cash grant that is set aside for future one-time purchases, modifications, or for cash.
   E. <u>Fiscal Intermediary Expenses</u>: Indicate the total amount of expenses for bookkeeping and other business related costs under the Fiscal Intermediary.
  - MONTHLY BALANCE: Subtract the expenses listed in B, C, D, and E, from the Total Monthly Cash Benefit (A) amount indicated above. In completing this calculation, your Monthly Balance should be as close to "0" as possible without having a negative balance. Any extra funds will be kept in your account.
- your account.

  Please sign, or ask your representative to sign and date your Cash Management Plan.
- Carefully review the Cash Management Plan for accuracy and return it to your Consultant for review. Your Consultant will then complete and sign that the Cash Management Plan was reviewed, and will send it to the State Program Office for signature and approval.
- The State Program Office will call you if they have any questions.

Instcmp1: 6/99:7/99:8/99:8/6/99:8/10/99:8/16/99:11/12/99

### CRITERIA FOR USES OF CASH BENEFIT

The following items are allowable under the Personal Preferences Program:

Purchase of services from individuals, including family members.

Background checks and benefits for employees.

Purchase or increase in rental or homeowner's insurance, or other liability insurance as it relates to participant's role as employer.

Adult Day Care

Caregiver training and education which enables the caregiver to deliver home care with high levels of quality. May be purchased from a variety of sources, including a home care agency or a vocational or technical school.

Chore Services. This includes outside chores that provide for a safe environment and access in and out of the home.

Cleaning Service from firms or individuals.

Food preparation and delivery of prepared foods.

Transportation services not currently paid for under Medicaid.

Laundry service from a Laundromat or other provider.

Errand service to assist with banking, shopping and other types of routine tasks.

Home modifications such as ramps and grab bars, installation of visual or tactile alarms as well as wander alarms and other modifications not currently paid for by Medicaid.

Respite services to relieve unpaid caregivers.

Supplies and equipment that promote or enhance independence that are not currently paid for by Medicaid.

Services Not Authorized For Payment With Cash Management Plan without State review.

Food

Entertainment equipment or supplies such as videos, VCR's, televisions, stereos CD's.

Illegal drugs or alcohol.

Travel for vacations or entertainment.

Istapsrv 2/98

# REPRESENTATIVE SELF-SCREENING QUESTIONNAIRE

# **New Jersey Department of Human Services** State Office on Disability Services Personal Preference Program (New Jersey Cash & Counseling Demonstration)

Representative Screening Questionnaire				
Name of Participan	t			
Medicaid #:		Phone #:	(	)
Name of Potential F	Representative	S		
Address:				
Phone #: (	)	Relationship:		
Are you a: family	member	friend	le	egal guardian of the erence Program?
have contact with the	ne participant:_			tionship and how often you
	ney from the pa	articipant or anyo	one else to	o care for the participant?
	nd for what pu	rpose?		
you understand you participant's repres	our functions entative?	and are you v Yes:	willing to	es of the representative, do volunteer to serve as the No:
Are you willing to si		on form stating t Yes:		No:
Do you understand		ot pay yourself in Yes:		? No:
If you have any qu	estions, please	e ask the consul	Itant or ca	II (toll free), 1-888-285-3036

before signing the designation form.